

What's cooking? an assessment of the potential impacts of selected novel alternatives to conventional animal products Rueda, O.; Scherer, L.A.; et al.

Citation Rueda, O., & Scherer, L. A. (2023). *What's cooking?: an assessment of the potential impacts of*

selected novel alternatives to conventional animal products. Nairobi: UNEP. doi:10.59117/20.500.11822/44236

Version: Publisher's Version

License: Leiden University Non-exclusive license Downloaded from: https://hdl.handle.net/1887/3718770

Note: To cite this publication please use the final published version (if applicable).

What's cooking?

An assessment of the potential impacts of selected novel alternatives to conventional animal products





© 2023 United Nations Environment Programme **ISBN:** 978-92-807-4103-2 Job number: DEW/2594/NA **DOI:** https://doi.org/10.59117/20.500.11822/44236

This publication may be reproduced in whole or in part and in any form for educational or non-profit services without special permission from the copyright holder, provided acknowledgement of the source is made. The United Nations Environment Programme would appreciate receiving a copy of any publication that uses this publication as a source. No use of this publication may be made for resale or any other commercial purpose whatsoever without prior permission in writing from the United Nations Environment Programme. Applications for such permission, with a statement of the purpose and extent of the reproduction, should be addressed to unep-communication-director@un.org.

Disclaimers

The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations concerning the legal status of any country, territory or city or area or its authorities, or concerning the delimitation of its frontiers or boundaries.

Mention of a commercial company or product in this document does not imply endorsement by the United Nations Environment Programme or the authors. The use of information from this document for publicity or advertising is not permitted. Trademark names and symbols are used in an editorial fashion with no intention on infringement of trademark or copyright laws.

The views expressed in this publication are those of the authors and do not necessarily reflect the views of the United Nations Environment Programme. We regret any errors or omissions that may have been unwittingly made.

© Maps, photos and illustrations as specified

Suggested citation

United Nations Environment Programme (2023). Frontiers 2023. What's Cooking? An assessment of the potential impacts of selected novel alternatives to conventional animal products. Nairobi. https://doi.org/10.59117/20.500.11822/44236.

Production: Nairobi.

URL: https://www.unep.org/resources/whats-cooking



Acknowledgements

Lead Authors

Cleo Verkuijl, Stockholm Environment Institute (SEI), Harvard Law School; Rajeswari Raina, Shiv Nadar University; Robert Watson, University of East Anglia

Substantive Support:

UNEP support team, Early Warning and Scientific Assessments Division, Scientific Assessments Branch. Rachel Kosse, Allan Lelei, Brigitte Ohanga, Adele Roccato, Pinya Sarasas, Edoardo Zandri

Authors

Ploy Achakulwisut, SEI Asia; Michael Carter, The Good Food Institute (GFI); Varun Deshpande, GFI India; Dolapo Enahoro, International Livestock Research Institute (ILRI); Shayna Fertig, GFI; Carsten Gerhardt, Kearney; Jonathan Green, SEI York; Ndaindila Haindongo, Food and Agriculture Organisation of the United Nations (FAO); Mario Herrero, Cornell University; Ana Maria Loboguerrero, Bioversity International -CGIAR; Daniel Mason-D'Croz, Cornell University; Tsakani Ngomane, University of the District of Columbia; Katie Noble, Leverhulme Centre for Anthropocene Biodiversity, University of York University of York; Oscar Rueda, Leiden University; Laura Scherer, Leiden University

Reviewers

Genaro Aguilar, Instituto Politécnico Nacional; Isabel Celia Homyak, University of California Berkeley; Philip Albinelli, FAO; Pablo Arena, Cliope Group, UTN/CONICET; Howard, Michigan State University; Nurul Huda, Faculty of Food Science and Nutrition, University Malaysia Sabah; Mirey Atallah, UNEP; Charles Bartlett, UNEP; Bockline Jason Jabbour, UNEP; Sylvia Karpagam, Freelance; Omedo Bebe, Egerton University; Claire Bomkamp, GFI; Jinfeng Chang, Zhejiang University / International James Lomax, UNEP; Ingrid Ym Ruth Odegard, CE Delft; Suzan Oelofse, Council for Scientific and Industrial Institute for Applied Systems Analysis (IIASA); Michelle Research; Cecilia Moraa Onyango, University of Nairobi; Colgrave, CSIRO; Adam Dorr, RethinkX; Jan Dutkiewicz, Pratt Institute; Andrea Hagyo, European Environment Martina Otto, UNEP; Jacob Robert Peacock, Rethink Agency; Anju Chadha, IIT Jammu; Ayako Ebata, Institute Priorities; Jeff Sebo, New York University; Sergiy Smetana, DIL German Institute of Food Technologies (DIL of Development Studies, University of Sussex; Kun Long e.V.); Madhura Swaminathan, Indian Statistical Institute; John Eng, A*STAR- Agency for Science, Technology and Research; Michele Fontefrancesco, University of Elliot Swartz, GFI; Oliver Taherzadeh, Leiden University; Hanna Tuomisto, University of Helsinki; Mariana Hase Gastronomic Sciences; Bruce Friedrich, GFI; Carlos Maria Gonzalez Fischer, Cornell University; Dominic B A Glover, Ueta, Technische Universität Dresden; Marina Venancio, Institute of Development Studies, University of Sussex; UNEP; Helena Wright, FAIRR Initiative; Alvaro Gabriel Zopatti, UNEP Hugh Charles Jonathan Godfray, Oxford Martin School, University of Oxford; Andres Guhl, Universidad de los Andes; Andrea Hinwood, UNEP; Simone Højte, CONCITO;

Supporting UNEP Divisions and Offices

Communications, Economy, Ecosystems, Law, Policy and Programme, Regional Office for Latin America and the Caribbean, Office of the Chief Scientist

Editorial, design and translation support

Stephen Graham, Science Editor; Amanda Lawrence-Brown, Science Editor (UNEP); Carlos Reyes, Graphic Design and Cover Illustration

Thanks also to

Angeline Djampou, Nicolien Delange, Niklas Hagelberg, Andrea Hinwood, Jason Jabbour, Jian Liu, Maarten Kappelle, Jane Muriithi, Kelly West, Yahan You

Partners and funders

Financial support from The Belgium Foreign Affairs, Foreign Trade and Development Cooperation to produce this publication is gratefully acknowledged.



List of abbreviations

ASF	Animal source foods	IPBES	Intergovernmental Science-Policy Platform on Biodiversity
CCAC	Climate and Clean Air Coalition		and Ecosystem Services
CIFOR	Center for International Forestry Research	IPCC	Intergovernmental Panel on Climate Change
COVID-19	Coronavirus disease	IPES-Food	International Panel of Experts on Sustainable Food System
EFSA	European Food Safety Authority	LCA	Life cycle assessment
FAO	Food and Agriculture Organization of the United Nations	NCD	Non-communicable disease
FBS	Fetal bovine serum	OECD	Organization for Economic Cooperation and Development
FCR	Feed Conversion Ratio	RCT	Randomized controlled trial
FDA	Food and Drug Administration (USA)	UNDESA	United Nations Department of Economic and Social Affairs
FSANZ	Food Standards Australia New Zealand	UNEP	United Nations Environment Programme
GFI	The Good Food Institute	UNICEF	United Nations Children's Fund
GHG	Greenhouse gas	UPF	Ultra-processed food
IFAD	International Fund for Agricultural Development	USDA	United States Department of Agriculture
ILRI	International Livestock Research Institute	WHO	World Health Organization

ns

S

Glossary

Animal sentience

Sometimes refers to the animal's capacity for any type of subjective experience, and sometimes to the capacity to have subjective experiences with a positive or negative valence, such as pain or pleasure (Browning and Birch 2022).

Animal source foods

Products sourced from animals. In this report, the term is used to refer to foods derived from animals. such as beef, pork, mutton, poultry, and dairy.

Antimicrobial resistance

A global public health issue characterized by the ineffectiveness of antibiotic, antiviral, antiparasitic and antifungal treatments stemming from inappropriate use of antimicrobials, often as a result of chemical and biological pollution from the pharmaceuticals, agriculture and healthcare sectors and municipal waste (UNEP 2023).

Macronutrient

Nutrients that provide calories or energy and are required in large amounts to maintain body functions and carry out the activities of daily life. There are three broad classes of macronutrient: proteins, carbohydrates and fats (WHO 2023a).

Micronutrient

Vitamins and minerals needed by the body in very small amounts. However, their impact on a body's health is critical, and deficiency in any of them can cause severe and even life-threatening conditions. They perform a range of functions, including enabling the body to produce enzymes, hormones and other substances needed for normal growth and development (WHO 2023b).

Biomass fermentation

The process of using microorganisms to make protein-rich food, where the microorganisms produced are themselves the primary ingredient.

Biopsy

Technique to collect tissue samples from living donor animals (Melzener et al. 2020).

Cell lines

Populations of cells that can be maintained for an extended period.

Cultivated meat

Meat produced directly from animal cells. This is done by extracting cells from a living animal and growing them in bioreactors. Cells can be differentiated into muscle, fat and other cell types to create products that have the same or similar three-dimensional structure, nutrition profile and organoleptic properties as conventional meat.

Mycoprotein

A fungal-derived protein source with a fibrous structure (Ahmad et al. 2022).

Non-communicable disease

Noncommunicable diseases (NCDs), also known as chronic diseases, are not transmissible directly from one person to another. NCDs tend to be of long duration and are the result of a combination of genetic, physiological, environmental and behavioural factors (World Health Organization [WHO] 2023).

Novel animal source food alternatives

Products with an appearance, taste, smell and texture similar to or even indistinguishable from conventional ASF, produced through new scientific approaches.

Novel plant-based foods

These products aim to replicate the sensory experience of animal products by combining plant protein (typically from soy or pea) with fats, vitamins, minerals and water (Figure 3.1).

Organoleptic experience

Culture media

Contains the nutrients and growth factors needed to cultivate cells outside an animal's body and culture the muscle, fat and connective tissue cells (O'Neill et al. 2020).

Enteric fermentation

A natural part of the digestive process in ruminant animals such as cattle, sheep, goats and buffalo. Microbes in the digestive tract, or rumen, decompose and ferment food, producing methane as a by-product (CCAC 2023).

Feed conversion ratio

A key characteristic describing requirements of crops per unit of end product; describes the efficiency of turning feed crops into animal meat products (Sinke et al. 2023).

Fermentation-derived products

Foods produced using biomass or precision fermentation. Biomass fermentation is the process of using microorganisms to make protein-rich food, where the microorganisms produced are themselves the primary ingredient. Precision fermentation uses microorganisms to produce specific functional ingredients, including proteins, vitamins and flavour molecules. These can be used in novel plant-based food to improve taste or texture, and in cultivated meat to enable more efficient growth (Figure 3.3).

Sensory properties like flavour, aroma, texture, bite, moisture, mouthfeel, appearance and colour.

Foodways

The eating habits and culinary practices of a people, region, or historical period.

Functional diversity

An important component of biodiversity that characterizes the variability of functional traits within a community, landscape or even large spatial scales. It can influence ecosystem processes and stability (Ma et al. 2019).

Species richness

Represents a measure of the variety of species based simply on a count of the number of species in a particular sample (Fedor and Zvaríková 2019).

Precision fermentation

Uses microorganisms to produce ingredients, including particular proteins, flavours, vitamins and fats, to be added to a final food product. These can be used in novel plant-based food to improve taste or texture, and in cultivated meat to enable more efficient growth.

Heme protein

Iron-containing proteins, such as leghemoglobin and myoglobin, which give plant-based meats a taste and aroma similar to that of conventional meat.

Just transition

Emphasizes that large socio-economic shifts including in response to climate change should be planned and implemented in a way that is socially fair. Its principles encourage governments to work with stakeholders to design policies that will help to minimize disruptions and maximize benefits for stakeholders affected by transition.

Protein fractionation

The extraction of protein from the rest of the plant.

Scaffolding

Materials used to support and guide tissue formation for tissue-engineered constructs include synthetic polymers, self-assembling peptides, extracellular matrix (ECM) molecules and plant- or fungus-derived materials (Bomkamp et al. 2021).

Sensory profile

Appearance, taste, smell and texture.

Zoonoses

Diseases that can spread between animals and people, moving from wild and domesticated animals to humans and from humans to animals (UNEP 2021b).

Contents

This report is designed to be read on screens. Some pages may not print with a legible font size on a standard A4.

1. Introduction

3.

New technological solutions are being developed to provide an alternative to animal source foods

Acknowledgemen	ts III
List of abbreviatio	ons IV
Glossary	V
Foreword	VII
Key findings	VIII

About this report VIII

Executive summary IX



2.

Animal source foods significantly impact the environment, human health, socioeconomic dynamics and animal welfare





- 2.1 Global demand for animal source foods continues to increase 5
- 2.2 The current animal agriculture system contributes to climate change, pollution and biodiversity loss 7
 - 2.2.1 Greenhouse gas emissions from animal agriculture significantly contribute to climate change7
 - 2.2.2 Some animal agriculture systems con-tribute to air and water pollution and soildegradation 8
 - 2.2.3 Land use, land-use change, pollution and climate change associated with animal agriculture systems are important drivers of biodiversity loss
- 2.3 Animal agriculture is associated with bothbenefits and harms for public health11
- 2.4 Animal agriculture operates within
 broader food system inequalities and meat
 can carry important cultural and social
 significance
 13
- 2.5 Animal agriculture raises and kills tensof billions of sentient animals annually 14
- 2.6 Interventions have been proposed to reduce the adverse environmental effects of the current animal agriculture system
 15

3.1	Nove and t	el meat and dairy alternatives aim to mimic the taste exture of animal source food	17
	3.1.1 3.1.2	Novel plant-based meat uses new technological approaches to closely imitate the sensory profile of meat Cultivated meat is made from animal cells grown	17
	02	in a culture medium	19
	3.1.3	Fermentation-derived products produce protein-rich food	
		using microorganisms	22
3.2	Ther	e is a wide range of estimates for market share and	
	cons	umer uptake of novel alternatives	24
3.3	Nove	I meat and dairy alternatives have potential environmental	
	bene	fits compared to conventional counterparts	27
	3.3.1	Alternatives could reduce greenhouse gas emissions	
		and use less land and water	27
	3.3.2	Novel plant-based meat	28
	3.3.3	Cultivated meat	28
	3.3.4	Fermentation-derived products	31
	3.3.5	Transparency is important for realizing	
		all potential environmental benefits	32
3.4	Alternatives could reduce the risk of zoonotic diseases, antimicrobial resistance and foodborne illnesses; further research on nutritional		I
	impli	cations is needed	32
	3.4.1	Reduced risk of zoonotic diseases, antimicrobial	
		resistance and foodborne illnesses	32
	3.4.2	Nutritional implications	32
	3.4.3	Impacts on workers and communities	35

3.5 Socioeconomic impacts will depend on the degree of uptake	9
of alternatives; more research is needed to fully understand	
these implications	35

3.6 Alternatives promise significantly reduced animal welfare issues 37

4.

Policy and regulatory environments can significantly influence the future of alternatives



4.1	1 Current policies impacting novel mea		
	and dairy alternatives		

- 4.2 Potential policies to support novel alternatives to conventional animal source foods
 43
 - 4.2.1 Policies targeting producers
 - 4.2.2 Policies targeting consumers4.2.3 Policies to support a sustainable and just transition
 - 4.2.4 Policies to support an enabling environment
 4.2.5 Multilateral cooperation

5.

Conclusion



List of tables

Table 2.1 Meat consumption per capita (kg per year)	5	F
Table 2.2 Greenhouse gas emissions from different sources	7	F
Table 2.3 Selected proposed interventions to reduce adverse impacts from animal agriculture	15	F
Table 3.1 Summary of different life cycle assessments for cultivated meat	30	F
Table 3.2 Summary of different life cycle assessments for fermentation-derived products	31	F
Table 4.1 Policy instruments targeting alternatives to conventional animal products	41	F
Table 4.2 Regulation of novel foods in selected countries	42	∎ f¢
Table 5.1 Overview of the environmental, health, social and animal welfare implications of novel plant-based, cultivated and fermentation-derived alternatives, compared to conventional meat and dairy products	53	F te F

List of figures

5	Figure 1 Scope of the report	viii
7	Figure 2.1 Global meat supply for human consumption	6
_	Figure 2.2 Global land use for food production	9
5	Figure 2.3 Environmental impacts of current meat production	10
)	Figure 2.4 Health and environmental impacts of different foods	12
	Figure 3.1 Novel plant-based meat	18
	Figure 3.2 Cultivated meat	20
	Figure 3.3 Types of fermentation	23
 	Figure 3.4 Global alternatives to conventional meat industry forecasts by year	25
	Figure 3.5 Breakdown of venture investments by company's country, technology type and type of product	26
3	Figure 3.6 Number of people a cow can feed for a year	38

40

43

45

46

48

49

Foreword

We are what we eat, and that makes us: unsustainable.

How we produce and consume food is contributing to Earth's triple environmental crisis: the climate emergency, nature and biodiversity loss, and pollution and waste, with livestock production and consumption playing a key role in all three.

Animal agriculture holds critical economic, social and cultural value. It is vital to the livelihoods of rural households especially in developing countries, and the global animal agriculture industry employs and provides healthy and protein-rich food for millions of people. Yet the animal agriculture industry is also a significant contributor to emissions of greenhouse gases – both direct animal emissions, and those associated with land clearing and grow-



ing animal food. Making room for more and more livestock and fodder crops is driving the loss of tropical forests, while excess animal manure and chemical fertilizers are polluting our groundwater, rivers and seas.

As global demand for meat and dairy products continues to rise, their production and consumption pose significant challenges for public health and animal welfare. Eating too much red and processed meat contributes to high rates of obesity and diabetes. Animal agriculture raises the risk of new zoonotic diseases and antimicrobial resistance. Many animals are raised and slaughtered in conditions that undermine their welfare.

It is clear that food systems, including the meat and dairy sector, must be part of the social and economic transformations required to halt and reverse the damage we are inflicting on Earth's natural systems. Achieving the Sustainable Development Goals, limiting global warming under the Paris Agreement and fulfilling the Kunming-Montreal Global Biodiversity Framework all depend on it.

The United Nations Environment Programme (UNEP) is committed to searching for science-based solutions and using its convening power. This work includes identifying and exploring emerging issues of environmental concern, including through the UNEP Frontiers reports.

This special edition of UNEP's Frontiers report explores the available evidence on novel plant-based foods, cultivated meat and fermentation-derived products as alternatives or complements to conventional meat and dairy



Inger Andersen Executive Director United Nations Environment Programme

that could potentially leave a much smaller environmental footprint.

The report provides an overview of scientific knowledge of these novel alternatives as elements of a wider reform of food systems. It examines the implications of their uptake for the environment and human health, as well as for societal dynamics and animal welfare. However, the full extent of their environmental, social and human health impacts is not yet fully understood. Drawing on examples from around the world, it also looks at the state of policies and regulations and identifies the tools that governments can use to steer the development of the sector.

The report also identifies where further research is urgently needed to plug knowledge gaps and inform growing public debate about the pros and cons of novel meat and dairy alternatives.

Novel alternatives to meat and dairy could play a positive role in a transformed global food sector, generating employment and technologies that power the sustainable economies of the future. But such a shift could also threaten a range of existing jobs as well as raise important questions about how it will affect disparities between the Global North and Global South and rural and urban communities, and further concentrate the market power of big companies.

Policymakers have a particular responsibility for ensuring that any such transitions are socially fair and well managed, and do not undermine food security or result in a more inequitable world for social minorities, including

women and Indigenous Peoples.

More and more people understand that we need to change our unsustainable ways. Mapping the frontiers of our knowledge, including of the impacts of what we eat and the potential of new technologies, can help us find the best path toward a better future.

Key findings

- Globally, food systems are responsible for about 30 per cent of the current anthropogenic greenhouse gas emissions driving climate change. Animal products—including animal emissions, feed, changes in land use and energy-intensive global supply chains—account for almost 60 per cent of food-related emissions, for a total of 14.5–20 per cent of global emissions.
- Impacts of the growing demand for animal source foods (ASF) take place in a context of unsustainable farming methods and overconsumption, especially in middle and high-income countries. Overall, production and consumption significantly contribute to climate change, air and water pollution, biodiversity loss, and soil degradation.
- While ASF are an important source of nutrition, high intake of red and processed meat is associated with increased risk of non-communicable diseases. ASF production has also been associated with public health risks such as zoonotic disease and antimicrobial resistance, and animal welfare concerns.
- Novel plant-based meat, cultivated meat and fermentation-derived foods show potential for reduced environmental impacts compared to many conventional ASF. They also show promise for reduced risk of zoonoses and antimicrobial resistance, and can significantly reduce onimal welfare concerns accepted with conventional animal agriculture.



reduce animal welfare concerns associated with conventional animal agriculture.

- Further research is needed to understand the potential socioeconomic and nutritional implications of novel ASF alternatives. Policymakers could also help maximize beneficial outcomes by taking steps to safeguard food security, jobs, livelihoods, social and gender equity and culture.
- The degree of uptake of these novel alternatives will likely depend on their cost, taste and social and cultural acceptability and on how they are regulated.
- Governments have numerous policy options to explore and support the potential of novel alternatives, including support for (open-access) research and commercialization and just transition policies.
- If supported by appropriate regulatory regimes and governance instruments, novel ASF alternatives can play an important role, likely with regional differences, in a shift towards food systems that are more sustainable, healthier and less harmful to animals.

Figure 1. Scope of the report

Introduction

2.

Animal source foods significantly impact the environment, human health, socioeconomic dynamics and animal welfare

3.

New technological solutions are being developed to provide an alternative to animal source foods

4.

Policy and regulatory environments can significantly influence the future of alternatives

5. Conclusion

About this report

This report focuses on the potential environmental, health, social and animal welfare implications of the uptake of novel meat and dairy alternatives, in particular novel plant-based, fermentation-derived and cultivated products. A team of interdisciplinary experts has assessed the available evidence on the impacts of these alternatives in comparison with their conventional counterparts, identifying pertinent considerations for policymakers involved in regulating, investing in or providing other support for novel meat and dairy products and highlighting research gaps.

The report does not explore in depth how developing and shifting to novel alternatives compares with other strategies for reducing the negative impacts of the current meat and dairy industry, such as substituting meat and dairy with more traditional vegetarian and vegan products (e.g. tofu, tempeh), developing insects as a source of human and animal feed, promoting extensive and regenerative animal agriculture, reducing meat and dairy demand through pricing (e.g. meat taxes) or direct interventions to reduce animal emissions (e.g. feed additives). Neither does the report address fish and other aquatic animals, or meat from (other) wild animals.

Executive summary

Globally, animal source food (ASF) contribute substantially to many countries' economies and are a major source of employment and income. They are also an important source of protein, vitamins, minerals and other nutrients, especially in food-insecure settings, and carry special significance for many demographic groups and cultures. At the same time, studies have generally found that high intake of red and processed meat is associated with increased risks of obesity and non-communicable diseases. Global production and consumption of ASF, including beef, pork, mutton, poultry and dairy have increased substantially in the last decades, with significant regional variations, as a result of population growth, rising incomes and generally supportive government policies, among other factors. Based on projected increases in population and per capita meat consumption, current global meat consumption is projected to increase by 50 per cent or more by 2050 (notably with major regional differences).



Animal agriculture, including animal feed production, is estimated to contribute 14.5–20 per cent of global human-caused GHG emissions, thus contributing significantly to human-induced climate change, as well as widespread air and water pollution, loss of soil structure and nutrients and loss of terrestrial, freshwater and coastal biodiversity. Furthermore, some livestock production systems have been linked to increased risk of zoonotic diseases and are associated with rising antimicrobial resistance. There are also animal welfare concerns as tens of billions of sentient animals are raised and slaughtered every year.

A number of approaches of varying feasibility and potential impacts have been proposed to address the environmental impacts of the livestock

sector. These include investing in smaller-scale, extensive or regenerative livestock farms; direct interventions to reduce emissions from animal agriculture, such as feed additives; promoting reduced meat consumption in favour of whole plant sources of protein such as beans and lentils; and discouraging consumption of animal products with taxes or other policy levers. Thus far, such interventions have been limited, and are not achieving the desired impacts at the scale or speed necessary in the regions and amongst populations where such changes are most needed.

An additional approach that has attracted attention from policymakers and investors in recent years is to advance the development of novel alternatives such as novel plant-based, fermentation-derived or cultivated ASF **products.** These products have a sensory profile (i.e. appearance, taste, smell and texture) similar to or even indistinguishable from conventional ASF. These alternatives include:



- **Novel plant-based products,** made from plant protein (typically from soy or pea) combined with fats, vitamins, minerals and water to closely imitate the sensory profile of meat.
- Cultivated meat, which is real meat made from animal cells grown in bioreactors.

• Fermentation-derived products, including:

- Biomass fermentation-derived products, which are protein-rich foods created using the rapid growth of microorganisms that are themselves the primary ingredients; and
- Precision fermentation-derived products, which use microorganisms to produce ingredients, including particular proteins, flavours, vitamins and fats, to be added to a final food product.

widely. Projections for its share of total meat consumption range from 4 to 60 per cent by 2040, while projections for the market share occupied by each category of alternative also vary. This illustrates the inherent uncertainty of making predictions of uptake at this early stage of the industry's development. Significant technological advances are still required for these foods to become available at wider scale and to compete with conventional ASF on taste and price.

Assessing the environmental lifecycle impacts of novel ASF alternatives is difficult, as data is scarce, parts of the industry are not yet operating at scale and further developments are expected. **However, novel ASF alternatives already show strong potential for reduced environmental impacts compared to many conventional animal products.** From a GHG emissions perspective, the novel alternatives considered in this report compare especially favourably to beef, which is particularly high-emitting. Nevertheless, some novel products, including cultivated meat, can be energy-intensive to produce. Realizing their full emission reduction potential is therefore contingent on the use of low-carbon energy.

Targeted research is needed to comprehensively assess the public health implications of novel ASF alternatives as they develop. Both traditional plant-based foods and novel ASF alternatives are associated with reduced



risk of zoonoses emergence and anti-microbial resistance. Diets that emphasize minimally processed, plant-based foods are generally associated with reduced risks of premature mortality and non-communicable diseases. However, novel plant-based products currently tend to be highly processed and have high amounts of salt, though opportunities to enhance their nutrient quality exist. Evidence on the health impacts of ASF alternatives using fermentation or cultivated from animal cells is limited.

Understanding the potential socioeconomic implications of novel ASF

alternatives also requires further research. Nevertheless, it is clear that high uptake would disrupt current food systems with both positive and negative impacts for different stakeholders. Policymakers could help maximize beneficial outcomes by taking steps to safeguard food security, jobs, livelihoods, social and gender equity and culture.

ASF alternatives, including the novel forms discussed in this report, have the potential to drastically reduce harm to animals in the food system. Plant- and fermentation-based alternatives avoid the use of animals. Cultivated meat still involves the use of animals to obtain stem cells (through biopsies) and, in some cases, animal serum (for growth media). However, vastly fewer animals would be needed to support cultivated meat production, and companies are working towards eliminating the use of animal serum, with some proven successes.

The policy and regulatory environment for novel ASF alternatives is evolving rapidly, with many governments formulating and implementing new



policies and policy instruments. Many countries and regions—including Brazil, China, the European Union, India, Israel, Singapore and the United States of America—have invested in the production of novel ASF alternatives. Some countries, including Australia, Brazil and Denmark, have provided incentives to producers, with tax exemptions, subsidies and support for energy and market development, while some countries, including China, India and the Netherlands, are also investing in research, human resources, curricula development and the promotion of sustainable practices in this emerging sector. In contrast, in 2023 Italy approved a draft bill that would ban production, import and export of food grown in laboratories, including cultivated meat. Ways through which governments can support novel alternatives to become commercially viable include providing funding for research—in particular open-source research—and commercialization. Governments can also develop regulatory and approval frameworks that ensure food safety in a transparent and streamlined manner.

A shift away from unsustainable forms of production and consumption of conventional ASF and towards novel alternatives presents various uncertainties. Government decisions could facilitate increased environmental, social and health benefits through proactive policymaking to promote a just and sustainable transition. Governments could consider reducing and/or redistributing subsidies or other forms of support currently in place for industrial animal agriculture to ensure food prices reflect associated health and environmental costs.

International collaboration, including through joint research, development and harmonization of standards and international support, can also advance the uptake of novel alternatives, alongside other approaches for meeting global food security and nutritional needs.

Overall, novel ASF alternatives, if supported by appropriate regulatory regimes and governance instruments, can potentially play an important role in a shift towards food systems that are more sustainable, healthier

and less harmful to animals, with likely regional differences. Equitable, evidence-informed policies are needed to ensure positive outcomes. Understanding of the implications of these technologies and their interactions with other environmental, health and social systems continues to evolve, highlighting the need for more research, especially open-source research. Policymaking will benefit from additional independent assessments of the environmental, health and socioeconomic implications of novel food technologies, as well as a better understanding of which policies are most effective in regulating and/or promoting them, and in what geographical, socio-economic and, in some cases, cultural contexts they are best deployed.



1. Introduction

Globally, food systems are responsible for about 30 per cent of the current anthropogenic GHG emissions driving climate change (Intergovernmental Panel on Climate Change [IPCC] 2021). In turn, animal products (including feed, direct emissions, land-use change and supply chains), account for almost 60 per cent of food-related emissions (Machovina, Feeley and Ripple 2015; Poore and Nemecek 2018; Zabel *et al.* 2019; Xu *et al.* 2021). More than three-quarters of the world's farmland is taken up by production of animal products (United Nations Environment Programme [UNEP] 2021b), but the environmental impact varies with the type of meat, modes of production (land, labour and capital intensity), production practices and the nature and magnitude of support (such as subsidies) from governments.



A shift away from land-intensive animal protein production systems, especially cattle farming, could free up land and water, dramatically reduce their carbon footprint and leave space for the restoration of degraded ecosystems, contributing to climate change mitigation and biodiversity protection (Machovina, Feeley and Ripple 2015). However, this is an enormous challenge made more complex by the economic, social and cultural importance of the sector. ASF can provide nutritional health benefits but also carry risks, depending on the food type and the context in which they are produced and consumed. For example, red meat is an important source of protein, vitamins (e.g. vitamins D, B6 and B12), minerals (e.g. iron, selenium and zinc) and other nutrients such as essential amino acids (e.g. lysine, threonine and methionine). At the same time, studies have generally found that high levels of red and processed meat consumption are associated with increased risk of non-communicable diseases (NCDs), particularly cardiovascular disease, type II diabetes and certain cancers (Clark et al. 2019; Figure 2.3). Increased risk of the emergence of zoonotic infectious diseases, such as avian flu, and of anti-microbial resistance is also associated with animal farming (Morand 2020; Wiebers and Feigin 2020; World Health Organization [WHO] 2017).

Changes in the way we produce, distribute and consume ASF must be steered in ways that ease, not exacerbate, the food insecurity that still plagues many parts of the world. About 800 million people are currently affected by hunger and are undernourished, and more than two billion suffer food and nutritional insecurity (Food and Agriculture Organization of the United Nations [FAO], International Fund for Agricultural Development [IFAD], United Nations Children's Fund [UNICEF], World Food Programme [WFP] and WHO 2022). Although both men and women play a significant role in smallholder food production, women's food security is more vulnerable to environmental and economic pressures (Rao *et al.* 2019).

In developed countries, growing numbers of people—from 'flexitarians' cutting down on meat to vegetarians eliminating it and vegans shunning animal products altogether—are reducing their intake of animal protein (Ajena *et al.* 2021). But these trends are projected to be outpaced by overall growth in ASF consumption globally, and most people continue to consume animal products, often for their flavour or for cultural or social reasons. Per capita consumption of ASF however remains low in many low- and middle-income countries, particularly among nutritionally vulnerable populations such as children under five years, and women of reproductive age (United Nations, 2021). The significant impacts of animal agriculture raise questions of whether conventional ASF could be replaced with novel alternatives that appeal directly to meat-eaters at least in some regions. Classic plant-based ingredients such as tofu and seitan have a long history of being used as protein sources. However, these foods do not necessarily match the taste and texture of animal products that appeal to many consumers. New technologies are being used to address this, and include novel plant-based products, cultivated meat and products derived from biomass and precision fermentation (see Box 1.1).

In industrialized countries, retail sales of alternative proteins surpassed US\$5 billion in 2021 (CE Delft and The Good Food Institute [GFI] 2021; Figure 3.4). This is modest compared with the value of the meat industry globally estimated at about US\$900 billion in 2021 (Shahbandeh 2022), but significant for an emerging industry.

Already, plant-based sources provide 57 per cent of protein that humans consume globally (UNEP 2021c). Some predict that new ASF alternatives will provide a significant additional portion of the global protein supply within the next decade or two (Gerhardt *et al.* 2019; UNEP 2021b; GFI 2023a), though others are cautious about their potential market penetration



Box 1.1 Definitions

Novel plant-based foods aim to replicate the sensory experience of animal products by combining plant protein (typically from soy or pea) with fats, vitamins, minerals, water and other additives. This grouping does not include more traditional plant-based meat alternatives such as tofu, tempeh, seitan, mushrooms and jackfruit.

Cultivated meat is meat produced directly from animal cells. This is done by extracting cells from a living animal and growing them in bioreactors. Cells can be differentiated into muscle, fat and other cell types to create products with a three-dimensional structure and organoleptic properties similar or identical to those of conventional meat products.

Fermentation-derived products are foods produced using biomass or precision fermentation. Biomass fermentation is the process of using microorganisms to make protein-rich food, where the microorganisms produced are themselves the primary ingredient. Precision fermentation uses microorganisms to produce specific functional ingredients, including

proteins, vitamins and flavour molecules. These can be used in novel plant-

based food to improve taste or texture, and in cultivated meat to enable

more efficient growth.

(Chriki and Hocquette 2020; Humbird 2021; IPES-Food 2022). An important question is whether, and to what degree, these new products can replace conventional ASF, or if they will merely complement them.

Amid rising public interest and debate, governments are beginning to respond. An increasing number of countries have begun to invest in research and in the production of alternatives to animal products (Section 4). On the other hand, a few countries are responding with bans and restrictive regulations, citing concerns such as impacts on current animal agriculture, farming communities and local culinary traditions (DeSoucey 2010; Sabelli 2023). Realizing the potential benefits depends on reducing the production and consumption of conventional ASF and significant substitution with new alternatives.

This report aims to advance the discussion of the potential role of novel ASF alternatives in contributing to a more environmentally sustainable, healthy and socially and morally acceptable food system. It focuses on three products-novel plant-based products, biomass and precision fermentation products and cultivated meat—and the production technologies involved. It seeks to assess available evidence on the impacts of these innovations and identify possible policies, further research and safeguards that could support them, if governments choose that course.

The report does not address more traditional vegetarian and vegan products (e.g. tofu, tempeh, mushrooms) or insects. Neither does it address fish and other aquatic animals, even though novel technologies may also have a role in shaping the development of the fishing sector.

The rest of this report is structured as follows:

Section 2 examines in more detail the current and projected regional and global demand for ASF and the environmental, human health, socioeconomic and animal welfare implications, recognizing the diversity of production systems. It discusses the need for a system-wide transformation to achieve a more sustainable, healthy and equitable food system, and briefly discusses the potential to decrease the environmental impact of the current food system through existing approaches.

Section 3 discusses three types of novel alternatives to conventional animal products: novel plant-based products, fermentation-derived products and cultivated alternatives (described in Box 1.1). It also examines the available evidence on required inputs, current status and projected uptake of these novel foods, and considers their potential environmental, health, socioeconomic and animal welfare implications compared to conventional meat and dairy at varying levels of uptake.

Section 4 identifies and discusses policies that are already being implemented to regulate or stimulate novel ASF alternatives, and what options could be considered if governments and societies want to further support their uptake.

Section 5 presents the report's key conclusions and highlights remaining knowledge gaps.

Feeding a growing population with affordable, healthy, nutritious and safe food in an environmentally sustainable, socially acceptable and morally responsible manner is one of the greatest challenges of the 21st century.

This report is intended to help shed light on both the potential and the limits

of novel alternatives to conventional animal products in helping to meet this



2. Animal source foods significantly impact the environment, human health, socioeconomic dynamics and animal welfare



2.1 Global demand for animal source foods continues to increase

Rising demand is exerting immense pressure on food systems, with the livestock sector generally growing faster than other sectors. Global production of cereals increased by 14 per cent over the last decade, while production of meat, milk and eggs increased by about 15 per cent, 18 per cent and 22 per cent, respectively (FAO 2022). Much of the increase in demand for cereals (and some other crops) is linked to their use as livestock feeds (Organization for Economic Cooperation and Development [OECD] and FAO 2021). Over the past 60 years, total meat consumption has increased fivefold and per capita consumption has almost doubled (Figure 2.1). Factors including income growth, urbanization and globalization are driving the rising demand for livestock products, especially in emerging and lower income economies (Milford et al. 2019; Fukase and Martin 2020; Latino et al. 2020; Reardon et al. 2021). Population growth is also a key factor in less affluent regions, including many countries in Africa, where, incidentally, per capita intakes of ASF remains very low (Latino, Pica-Ciamarra and Wisser 2020). With global per capita meat consumption expected to increase by 0–0.5 per cent per year (Henchion et al. 2021) and global population expected to rise by 1 per cent per year (United Nations Department of Economic and Social Affairs [UNDESA] 2022) global meat consumption in 2050 is projected to be about 50 per cent or more greater than present.

Within the global food system, patterns of production and consumption vary considerably across and within regions. For example, per capita consumption of animal meat (the sum of beef, pork, sheep meat, goat and poultry) in Europe and North America is up to eight times that in Asia and Africa (Table 2.1 and Figure 2.1 show recent estimates of regional meat consumption). ASF can be a vital component of healthy, diverse diets, particularly for nutritionally vulnerable populations in low- and middle-income countries, e.g. children and women of reproductive age (Alonso, Dominguez-Salas and Grace 2019; International Livestock Research Institute [ILRI] 2019). On the other hand, health issues related to poor diet quality and the prevalence of obesity are increasing steadily in both rich and poor countries alike (Branca *et al.* 2020).

In addition to between-country variation, there is also significant within-country variation in the consumption of ASF. In some countries, specific consumer groups consume meat in amounts that exceed dietary protein requirements (Behrens *et al.* 2017).

Table 2.1 Meat consumption per capita (kg per year)¹

OECD-FAO Outlook	
34.7	
95.3	
70.7	
64.8	
59.6	
27.0	
12.7	
	OECD-FAO Outlook 34.7 95.3 70.7 64.8 59.6 27.0 12.7





Regional meat supply for consumption per capita in 2020



Poultry meat

9

Regional meat supply for consumption in 2020

(million tons)





Source: (OECD and FAO 2022; FAO 2023a; FAO 2023b; UNDESA 2023).

Meat production also varies considerably by region, with Asia being the largest producer (about 43 per cent of global supply), followed by Europe (about 21 per cent), North America (about 15 per cent, South America (about 14 per cent) and Africa (about 7 per cent) (Ritchie, Rosado and Roser 2019).

Government policies have contributed significantly to the development of current meat production systems. Decades of investment, incentives (e.g. subsidies) and international trade in industrial livestock products have come with a large cost to the environment (Chandel, Lal and Kumari 2019). Poultry, pork, mutton, beef and dairy are among the food products that benefit the most from government subsidies, a significant share of which are allocated to industrial production (Stoll-Kleemann and O'Riordan 2015). Moreover, environmental and health costs of meat production and consumption are not reflected in the price or labelling of ASF that reach the consumer. Consequently, animal products, especially those from intensive production systems, are often significantly under-priced (Mosquera 2018; McCormack 2021; Funke *et al.* 2022).

2.2 The current animal agriculture system contributes to climate change, pollution and biodiversity loss

The current food system is a significant contributor to human-induced climate change. It also contributes to the unsustainable use of natural resources and the pollution of land, air and water, and is a significant driver of biodiversity loss through the degradation and conversion of terrestrial, freshwater and marine ecosystems (Springmann *et al.* 2018).

The environmental impact of meat production varies with the type of meat, modes of production (land, labour and capital intensity), production practices and the nature and magnitude of support from governments. A shift away from land-intensive protein production systems could free up land and water, dramatically reduce their emissions, and create conditions for ecosystem restoration (Hayek *et al.* 2021; IPCC 2021; IPCC 2023).

2.2.1 Greenhouse gas emissions from animal agriculture significantly contribute to climate change

While the energy sector (production and use of fossil fuel energy, including for transportation) is responsible for over 70 per cent of anthropogenic GHG emissions (Gerber *et al.* 2013; IPCC 2022), the way people use land contributes around 30 per cent of total emissions, mostly from food production (Clark *et al.* 2020; Jackson *et al.* 2020; IPCC 2021; Eisen and Brown 2022).

Overall, agricultural land use contribute around 70 per cent of food system emissions, with supply chain activities such as processing, retail and waste management contributing the rest (Crippa *et al.* 2021). IPCC (2022) estimated that agriculture, forestry and other land uses contributes 13 GtCO₂eq, 22 per cent of global GHG emissions. Regionally, Asia and the Americas (North, Central and South) are the largest contributors of food system emissions (FAO 2022).

Animal products—including animal feed, changes in land-use and energy intensive global supply chains—account for almost 60 per cent of food system GHG emissions (including methane, nitrous oxide and carbon dioxide), or between 14.5-20 per cent of total GHG emissions (Gerber *et al.* 2013, Xu *et al.* 2021).² Average emissions per kilogram of meat vary by up to a factor of 10, with beef having the biggest emissions footprint, and poultry the smallest (Xu *et al.* 2021). Emissions from cattle also vary widely, with significantly higher average emissions from beef cattle than dairy cattle (Xu *et al.* 2021). The largest producers of beef are Brazil, China, the European Union, India and the United States of America, and the largest exporters are Argentina, Australia, Brazil, India and the United States of America (Hocquette *et al.* 2018).

Direct emissions from livestock—through enteric fermentation and manure only—contribute 30-35 per cent of agricultural emissions (FAO 2021; UNEP and Climate and Clean Air Coalition [CCAC] 2021; FAO 2022), or about half of total animal product emissions. Cattle are the predominant contributors of methane emissions from enteric fermentation while pigs are the main source of nitrous oxide emissions from manure (UNEP and CCAC 2021).



Table 2.2 Major contributions to GHG from different sources³

Source	Contribution to total GHG emissions (per cent)
Energy (production and use)	~70
Agriculture/food systems	~30
Animal products (incl. feed, direct emissions, land-use change and supply chains)	14.5-20
Direct animal emissions	7.5-10

Methane, which is the second most emitted anthropogenic GHG after carbon dioxide, has a much more powerful per molecule warming effect in the atmosphere (Jackson *et al.* 2020; Saunois *et al.* 2020; UNEP and CCAC 2021), which coupled with its short atmospheric lifetime makes its emissions from livestock a focus of climate mitigation strategies (Arndt *et al.* 2022; Scoones 2022). The food system accounts for 60% of all methane emissions, with half of those coming from livestock farming and primarily from enteric fermentation (ClimateWorks Foundation 2023).

² Additional sources include a lower figure of 11 per cent, derived from the FAO's third version of the online Global Livestock Environmental Assessment Model (GLEAM), however, the underlying methodology used by the GLEAM is yet to be published and peer-reviewed. In contrast, Reisinger and Clark 2018, which uses a different carbon modelling methodology, mentions a higher estimate of 23 per cent (FAO 2023d). ³ These numbers are estimates with a range of uncertainty.

2.2.2 Some animal agriculture systems contribute to air and water pollution and soil degradation

Beyond emissions, animal agriculture also contribute to other environmental issues. Livestock feed production accounts for more than 40 per cent of total agricultural water use (Heinke *et al.* 2020). The water footprint per unit of product is higher for animal products (especially beef) than for crops (Mekonnen and Hoekstra 2011; Gerbens-Leenes, Mekonned and Hoekstra 2013).

There is considerable variability in the impacts that different livestock products, production systems and supply chains have on the environment (Poore and Nemecek 2018; ILRI 2019; Adesogan *et al.* 2020). Extensive livestock production systems use fewer inputs, such as feed and chemicals, suggesting low environmental impacts. They can however be considered inefficient, compared to intensive systems, in their use of resources, given their low levels of ASF output (Bosire *et al.* 2015; Mottet *et al.* 2017; Willett *et al.* 2019; van Zanten, Muller and Frehner 2022). On the other hand, intensive cropping and industrialized livestock production systems are in many parts of the world driving nutrient mining, loss of soil structure (through compaction and soil erosion), and wide-spread air and water pollution (Sakadevan and Nguyen 2017; Hamza and Anderson 2005)

Run-off of fertilizers, pesticides and other chemicals to ground and surface waters have increased as crop production for food and livestock feed has intensified (Gerber *et al.* 2005; da Silva *et al.* 2010; Wang *et al.* 2017). Further, nutrient losses (particularly of nitrogen and phosphorus) have increased as a direct impact of livestock production expansion in many emerging countries (Gerber *et al.* 2005; Liu *et al.* 2017).

In intensive livestock systems such as those commonly found in the Global North, poor manure management and low nutrient use efficiency are associated with substantial loss of nutrients from the farming system to the environment (Gerbens-Leenes, Nonhebel and Krol 2010; Rothwell *et al.* 2020).

However, livestock can also deliver benefits. For instance, the use of animal manure in crop agriculture in many developing countries reduces the need for chemical fertilizers. Recycling of animal manure is widely practiced and contributes up to 24 percent of the nitrogen input into crop agriculture in the mixed crop-livestock systems of the developing world (Herrero et al., 2013). Livestock are also useful in the conversion of crop residue and plant protein and energy from sources typically not eaten by humans into high-value forms for human consumption (Varijakshapanicker *et al.* 2019; Barbieri *et al.* 2022). In fact, 86 per cent of food consumed by livestock is not currently eaten by humans and includes by-products, crop residues and grasses or fodder (Mottet *et al.* 2017).

2.2.3 Land use, land-use change, pollution and climate change associated with animal agriculture systems are important drivers of biodiversity loss

Land use and land-use change are the most important drivers of terrestrial biodiversity loss (Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services [IPBES] 2019; Scherer et al. 2020), and food systems are the biggest land user. While food systems occupy half of the habitable land (habitable land is 71 per cent of the land surface, not including glaciers and barren land), 77 per cent of agricultural land is used for the livestock sector. A significant percentage of land used for livestock (which includes land for growing crops, animal feed as well as pastures used for grazing) cannot be used for arable crops (Mottet et al. 2017, UNEP 2019, ILRI, International Union for Conservation of Nature [IUCN], FAO, World Wide Fund for Nature [WWF], UNEP and International Land Coalition [ILC] 2021) (Figure 2.1). Meat from ruminants like cattle and sheep has a particularly high land-use intensity (Poore and Nemecek 2018). Such land use also threatens Key Biodiversity Areas (sites critical for the persistence of global biodiversity), with beef production and pastures as main contributors to potential species loss (Sun et al. 2022). In areas where wild herbivores have disappeared, grazing livestock may partly replace their function within the ecosystem, subject to good management practices. However, such benefits to biodiversity are rare (Godfray et al. 2018).

Figure 2.2 Global land use for food production Earth's surface Land

71% Ocean





With rising demand for meat and dairy products, livestock production is increasing notably in developing, tropical countries that are rich in biodiversity, and this trend is projected to continue (Machovina, Feeley and Ripple 2015). The resulting biodiversity loss threatens food system resilience. For example, the expansion of agriculture is also a major driver of pollinator decline, and a considerable share of global crop production depends on animal pollination (IPBES 2016).

GHG emissions, to which livestock contribute considerably (as described above), contribute to climate change, thereby causing further loss of biodiversity. When climatic conditions, such as air and water temperature, precipitation and streamflow, exceed the niche limits to which a species is adapted, the geographical area in which it can live may shift or be reduced, and the more species are affected, the more biodiversity is threatened (Trisos, Merow and Pigot *et al.* 2020; Barbarossa *et al.* 2021; Scherer *et al.* 2023). A key issue is whether species can migrate fast enough to survive (IPBES 2019), assuming appropriate terrain is available.

Pollution from agrochemicals, such as fertilizers and pesticides used in crop production for human food, animal feed production and other uses also affects biodiversity (Bouwman *et al.* 2013; Nordborg *et al.* 2017). Emissions of nutrients such as phosphorus and nitrogen following fertilizer application can cause eutrophication of freshwater and marine water bodies, deplete oxygen and/or generate harmful algae blooms, and subsequently drive biodiversity loss (Glibert 2017). Oxygen depletion leads to so-called dead zones in coastal ecosystems that recently covered a total area the size of the United Kingdom of Great Britain and Northern Ireland (IPBES 2019). The toxic effects of pesticides harm more species than those targeted. Like fertilizers, they can eventually reach water bodies and reduce their biodiversity (Beketov *et al.* 2013). Moreover, pesticides enter the food chain, affecting also vertebrate species (Garcês, Pires and Rodrigues 2020).



NVIRONM

The second

Environmental impacts of current meat production

Agriculturalland

or the world's harvest is

used to reed livestock

and demand is increasing

to reed a growing population.

espercent of teed consumed

by livestock is currently not edible

by humans (Motter et al. 207.

Gerhard ex al. 2027

Unsustainable animal agriculture is one of the major causes of habitat loss, biodiversity decline and land degradation

OF the world's agricultural

land is dedicated

to the livestock sector

Includes graxing land tor animals

and arable land used for animal

reed production

Livestock farming sometimes involves deforestation and the clearing of natural habitats to create space for grazing or feed crops, leading to habitat loss and biodiversity declines. (Thornton and Herrero 2010)

Crops for human

Consumption

The diversity of local livestock breeds is threatened by the increasing replacement of a few international breeds – putting almost a fifth of breeds at risk of extinction. (Sponenberg et al. 2018)

Overgrazing and the land clearances for feed crop production can lead to soil erosion and land degradation. (FAO 2009)

The cultivation of monoculture feed crops leads to the loss of biodiversity in the surrounding area. (Mulvany 2021)

Biodiversity loss, through large pollinator declines, threatens food system resilience, as a significant share of global crop production depends on pollinators. (IPBES 2016)

Plant functional diversity may even be more sensitive to grazing pressure than plant species **richness**, with the risk of decreasing the plant communities' resilience to periodic disturbances, thus affecting ecosystem functionality. (Ren et al. 2017)

Crop production, a significant proportion of which is specifically targeted to industrial livestock, leads to chemical pollution of water and soils

Fertilizer and pesticide overuse to grow feed crops can degrade soil, pollute water and cause eutrophication, which negatively impacts both plant and animal species. (Steinfeld 2006; Delgado et al. 2008; Galloway et al. 2010)

Plastic coated fertilizers and pesticides contaminate soils and can impact aquatic species when they enter water bodies. (UNEP 2022)

Gases such as methane, ammonia and hydrogen sulphide emitted from animal waste can contribute to air and water pollution, with negative impacts on human and animal health as well as on climate change. (Park et al. 2020)

Increasing amounts of water are used to produce animal feed

Water resources can be depleted by irrigation to grow feed crops. (Richter et al. 2020)

The livestock sector stands among the major causes of food-related greenhouse gas emissions

X

Animal products, including animal feed, account for almost 60 per cent of food-related GHGs emissions (including methane, nitrous oxide and carbon dioxide). (Poore and Nemecek 2018)

Transport of livestock and feed along with other products in the supply chain also contribute to greenhouse gas emissions. (Rojas-Downing et al. 2017; Thornton and Gerber 2010)

Animal welfare considerations are increasingly prominent

Intensive livestock production is associated with major animal welfare issues that deprive farmed animals of basic physical and psychological needs. Extensive animal farming can also raise a range of welfare problems (Poletto and Hötzel 2012).

Also, improper handling, loading and transport methods can cause animals to experience psychological stress, fatigue and physical harm. (Fisher et al. 2009)

23% of production in the meat industry is wasted or lost

The overuse and misuse of antibiotics has negative impacts on human health and also leads to antimicrobial resistance in the environment

Ultilization of antibiotics to support animal growth, their inclusion in animal feed, and the subsequent discharge of antibiotics to the environment through manure and wastewater exert a selection pressure on bacteria, potentially contributing to antibiotic resistance, and potentially exposing humans to otherwise treatable diseases. (UNEP 2023)

A study estimated that in the meat industry approximately 23% of production is wasted or lost.

6 Holo

ar consumption

POoo

ar manuracruring

This incurs a substantial

environmental footprint.

(Karnonska er al. 2027)

ar distribution

3.500

ar the primary

production and

post-harvest levels

2.3 Animal agriculture is associated with both benefits and harms for public health



In the past 50 years, globalization and intensification of food production systems—combined with urbanization, rising incomes, expansion of processed food markets by transnational corporations and other societal trends—have led to a homogenization of dietary patterns across the world towards a 'Westernized' diet consisting of relatively high consumption of ASF (Table 2.1), processed foods, refined grains and sugars and fats (Cordain *et al.* 2005; Cena and Calder 2020; Vermeulen *et al.* 2020; Wood *et al.* 2021; da Costa *et al.* 2022;). However, large variations in the types and amounts of ASF consumption still exist between and within countries (Miller *et al.* 2022).



Consumption of ASF can provide health benefits and risks, depending on the context and food type. ASF can be an important source of nutrients, especially for young children, pregnant or lactating women, and for people on low incomes with limited access to alternative nutrient-dense foods (FAO 2023e). In many developing countries ASF consumption is limited and nutrient intake remains suboptimal, reinforcing the need to tailor dietary guidelines to different regions and age groups to prevent exacerbating existing public health challenges (Fite et al. 2022; Miller et al. 2023). Nevertheless, it is possible to obtain a sufficient intake of essential nutrients without eating meat if a wide variety of other foods is available and consumed (Godfray et al. 2018). At the same time, in high-income countries with relatively high meat intake such as the United States of America and those in western Europe, epidemiological studies and meta-analyses have generally found that high intakes of red meat and processed meat are associated with increased risks of obesity, NCDs including cardiovascular disease, type II diabetes, certain cancers (with the strongest evidence to date for colorectal cancer) and premature death (Godfray et al. 2018; Clark et al. 2019; Swinburn et al. 2019; Santo et al. 2020; Farvid et al. 2021; Kazemi et al. 2021). The International Agency for Research on Cancer (IARC) of the World Health Organization classifies red meat as 'probably carcinogenic' and processed meats as a known cause of cancer for humans (Bouvard et al. 2015). Afshin et al. (2019) estimated that in 2017 around 155,000 premature deaths among adults worldwide were attributable to high intakes of red and processed meats.

To date, these epidemiological studies have predominantly been conducted in high-income Western countries where individuals with relatively lower meat consumption may also typically lead a more health-conscious lifestyle. Further research is needed to comprehensively assess the health outcomes of ASF consumption in other populations that consume different quantities and types of meat and have dissimilar confounding factors (such as lifestyle and methods of food preparation), especially in low- and middle-income countries where the average baseline meat intakes are much lower than those found in high-income Western countries (Godfray *et al.* 2018; Clark *et al.* 2019).

Compared with processed and red meats, the evidence of health outcomes is less conclusive for eggs, dairy products and poultry. Some dose-response and substitution studies have found that higher intake of these products is significantly associated with higher risks of incident cardiovascular disease and premature mortality (Zhong *et al.* 2019; Zhong *et al.* 2020; Zhong *et al.* 2021), whereas others find no statistically significant relationships or even reduced risks (Dehghan *et al.* 2018; Clark *et al.* 2019; Drouin-Chartier *et al.* 2020).

Beyond the nutritional dimension, ASF consumption and production are also associated with other public health implications (Santo *et al.* 2020). ASF are a reservoir for many foodborne pathogens such as *Salmonella*, *Campylobacter* and *E. coli* (Lianou, Panagou and Nychas 2017; Heredia and García 2018). Li *et al.* (2019) estimated that ASF were responsible for 35 per cent of the global foodborne disease burden in 2010.

The widespread and excessive use of antibiotics in animal agriculture has been linked to the rise of antimicrobial resistance in both animals and humans



(Van Boeckel *et al.* 2015; Talebi Bezmin Abadi *et al.* 2019); globally, 73 per cent of all antimicrobials sold are used in animal agriculture (Van Boeckel *et al.* 2019). Animal agriculture expansion and industrialization, and wildlife trade and consumption, have also been linked to increased risk of zoonoses emergence (Jones *et al.* 2013; Bernstein and Dutkiewicz 2021). The majority (70 per cent) of emerging infectious diseases and almost all known pandemics (e.g. influenze, HIV/AIDS, COVID-19) are zoonoses (WHO 2020; IPBES 2020). For example, there is evidence to suggest that the SARS-CoV-2 virus was transmitted from bats to other wild animals, subsequently spilling from these into humans in live animal markets (Wegner *et al.* 2022). Without preventative strategies to address the underlying causes that bring wildlife, livestock and people into close contact, pandemics may emerge more often and with potentially more devastating impacts (IPBES 2020; Wegner *et al.* 2022).

Intensive livestock farming can lead to harmful health impacts for industry workers and local communities. Occupational hazards include exposure to pathogens and toxic air pollutants, injuries due to handling of animals and operation of heavy machinery, and psychological stress in slaughterhouses (Myers 2010; Douglas *et al.* 2018; Ceryes and Heaney 2019; Slade and Alleye 2023). Neighbouring communities are at increased risks of developing respiratory illnesses, stress and other adverse health outcomes due to exposure to air, water, soil, noise and odour pollution from, for example, open animal waste pits in countries without adequate regulatory regimes (Greger and Koneswaran 2010; Casey *et al.* 2015; Johnston and Cushing 2020).

Figure 2.4 Health and environmental impacts of different foods

Foods associated with improved health (whole grain cereals, fruits, vegetables, legumes, nuts and olive oil), have among the lowest environmental impacts.







Food choices are shifting globally in ways that are negatively affecting both human health and the environment. Thus, dietary transitions toward greater consumption of healthier foods would generally improve environmental sustainability. These findings could help consumers, policy makers and food companies to better understand the multiple health and environmental implications of food choices. Meat has both positive and negative health effects — this figure highlights the adverse effects of excess consumption. Foods associated with the largest negative environmental impacts unprocessed and processed red meat—are consistently associated with the largest increases in disease risk. Foods high in sugars harm health but can have relatively low environmental impacts.

Source: Adapted from Clark et al. 2019

2.4 Animal agriculture operates within broader food system inequalities, while meat can carry important cultural and social significance

Food production is the world's single largest economic activity. Livestock, which contributes about 40 per cent of the global value of agricultural output and supports the livelihoods and food and nutritional security of almost 1.3 billion people (World Bank 2022), also has important cultural significance. Nevertheless, current patterns of food production and consumption, including those for ASF, are also associated with a range of negative social impacts.

Concerns have been raised about the loss of local food cultures, increase in food inequalities, power imbalances in food systems and the globalized procurement of foods, which increase food trade and distances across the value chain. Food justice and food sovereignty movements highlight class inequalities and power asymmetries in the food system that affect people's rights to culturally appropriate foodways (Motta 2021). While improved food systems have pulled millions of people from precarious food-insecure lives into healthy productive lives, they have also come with higher food and nutritional inequities, resulting from increasing environmental costs, social disruption and economic inequality (Ruttan 2005; Steffen *et al.* 2015; Krall 2022). The rise of obesity and diabetes in many countries has occurred in parallel with the globalization of food systems (Swinburn *et al.* 2019).

Animal agriculture has changed dramatically in recent decades, especially in



industrialized countries. In the United States of America, small-scale farms have been replaced by factory farms that are often operated to satisfy corporate contracts (Hribar and Schultz 2010) so that an estimated 99 per cent of farm animals are raised in factory farms (Athnis 2019). Factory farming practices are currently also being exported to developing economies, such as India and South Africa, where the food produced is not necessarily for local consumption but for export (Jankielsohn 2015; Lam *et al.* 2017). Factors contributing to the increase of factory farming in developing countries include rising local demand, lower production costs, and potentially weaker environmental regulations (Lam *et al.* 2017).

The meaning of meat is significantly mediated by culture. Different animals may be sacred in one culture but widely consumed in another. A prime steak is a luxury in the United Kingdom of Great Britain and Northern Ireland and the United States of America for example, but the cow is sacred amongst Hindus (Chigateri 2010).

Specific meat dishes can be recognized as culturally significant at the national level (e.g. haggis in Scotland and hamburgers in the United States of America) and economically significant (e.g. beef in Argentina and lamb in New Zealand). In addition to this, meat consumption is gendered, and several studies have found that commonly men eat more meat than women (e.g. Raty and Carlsson-Kanyama 2010; Rosenfield and Tomiyama 2021). Meat and perceived masculinity are currently interconnected, and studies have found that men not only eat more meat than women but are also more reluctant to reduce their meat intake and can develop an increasing attachment to meat

when they experience a threat to their perceived masculinity (Nakagawa and Hart 2019). Additionally, a high proportion of vegetarians in Western societies are female, potentially due to societal and cultural narratives surrounding

gender and meat consumption (Modlinska et al. 2020).

2.5 Animal agriculture raises and kills tens of billions of sentient animals annually



Scientific evidence indicates sentience in terrestrial vertebrates, including the birds and mammals in our food system, meaning that they can feel and suffer (Proctor 2012; Browning and Birch 2022). The boundaries of what we consider worthy of moral consideration have historically expanded over time (Crimston *et al.* 2016), leading to increasing public concern for non-human animals. Such concern is reflected in some laws that recognize animal sentience (Blattner 2019; Browning and Birch 2022).

Currently, tens of billions of land animals are slaughtered yearly for meat (FAO 2022). Many farmed animals are kept in confined intensive systems. By the early 2000s, most pork, poultry meat and eggs came from intensive systems (Steinfeld, Wassenaar and Jutzi 2006). Yields per animal, as well as milk and other meat, have continued increasing (FAO 2022), indicating further intensification. Intensive systems deprive animals of some of their most basic physical and psychological needs (Poletto and Hötzel 2012). High stocking densities, poor housing conditions and rapid growth of the animals can lead to aggression-induced injuries and lameness (Madzingira 2018). Even under extensive conditions, animals can experience welfare loss from, for example, hunger and thirst, thermal stress and painful husbandry practices; some diseases are also more likely than under intensive conditions (Temple and Manteca 2020). Pre-slaughter handling and transport to the slaughterhouse can also cause stress (Rioja-Lang et al. 2019; Benincasa et al. 2020). Consuming meat from smaller animals (e.g. chicken instead of beef) might worsen animal welfare by affecting a substantially larger number of animals (Scherer et al. 2018; Mathur 2022).

Higher animal welfare does not always conflict with efficient farming. For example, improved animal welfare can lead to lower mortality and better product quality while justifying sales at a premium, all of which can have financial benefits (Dawkins 2017).



2.6

Interventions have been proposed to reduce the adverse environmental effects of the current animal agriculture system

Several different approaches have been proposed to address the environmental, health, social and animal welfare impacts of the current global livestock system. They include: reducing loss and waste of ASF; smaller-scale, extensive or regenerative approaches to livestock farming; direct interventions to reduce GHG emissions from animal agriculture such as feed additives; promoting the reduction of meat consumption in favour of whole plant sources of protein; and reducing consumption of meat through the use of a meat tax or similar policy lever.

Table 2.3 describes selected approaches to mitigating the impacts of livestock systems, many of which could be pursued in parallel. Assessing their potential goes beyond the scope of this report. However, the success of these approaches has so far been limited, suggesting more action is necessary to tackle the adverse impacts of the current livestock industry.

Table 2.3 Selected proposed interventions to reduce adverse environmental impacts from animal agriculture



Reducing food loss and waste

Reducing losses in production and along the supply chain, and waste at the retail, service and household level would address a significant factor in the environmental impacts of the global food system. Currently about 23 per cent of global meat production goes uneaten, of which 64 per cent is wasted at the consumption level, 20 per cent in manufacturing, 12 per cent in distribution and about 3.5 per cent at the primary production and post-harvest levels (Karwowska *et al.* 2021). Reducing this waste of resources and inputs along the supply chain would ease challenges related to land and water use, nutrient and yield loss, biodiversity and GHG emissions (FAO 2014; Scialabba 2019; UNEP and CCAC 2021).



Smaller-scale, extensive or regenerative livestock farms

Such farms, including integrated crop-livestock systems, have gained favour as a means to make meat production more sustainable. These approaches aim to reduce or eliminate antibiotic use, support more biodiversity, and conserve more natural resources compared to industrialized animal agriculture. The evidence in support of extensive livestock farming having a lower environmental impact is mixed, especially given the range of farming systems included and ways to measure their impact. Extensive, 'pasture-based' systems may cause lower carbon dioxide and nitrous oxide emissions than more intensive systems because they do not require feed grain (Cusworth et al. 2022). One study estimates that regenerative multi-paddock systems could reduce GHG emissions from livestock production by 35 per cent compared with industrial-conventional systems (Kleppel 2020). However, other studies show higher emissions associated with grassfed models due to longer finishing time and lower finishing weight (Lupo et al. 2013). Additionally, land requirements for extensive systems are much greater, with one lifecycle assessment indicating 2.5 times more land is needed for a multispecies pasture rotation system than a conventional, commodity system (Rowntree et al. 2020).

	These have been posited as a way to suppress bovine methane emis- sions from enteric fermentation. However, the impact of these additives on emissions would be limited to animals held in feedlots; globally, many cattle (likely most cattle outside Australia, Canada and the United States of America) do not spend time on feedlots (Bechtel 2018; McBee 2022). In the United States of America, beef cattle that are grain-finished spend only 41 per cent of their 18-month-long lives on feedlots (Hayek and Garrett 2018)
Reducing meat con- sumption in favour of whole plant protein	This approach has the potential to address multiple problems caused by the current livestock system. Transitioning to plant-based diets could reduce diet-related land use by 76 per cent, diet-related GHG emissions by 49 per cent, and green and blue water use by 21 per cent and 14 per cent, respectively (Gibbs and Cappuccio 2022). The EAT-Lancet Commission on Healthy Diets from Sustainable Food Systems has advocated for diets that are' flexitarian' – largely plant-based, but optionally including modest amounts of fish, meat and dairy foods – to boost both human and plan- etary health (Willett <i>et al.</i> 2019). However, per capita meat consumption continues to rise (OECD and FAO 2021). Examples of advocacy for dietary shifts attempted to date include countries from China to Spain urging citi- zens to eat less meat, and programmes like 'Meatless Mondays' in school. (Dutkiewicz 2021). Some interventions show potential promise, such as appealing to animal welfare concerns and individual lifestyle coaching

(Bianchi et al. 2018; Mathur et al. 2021).

A 'meat tax' similar to those levied on alcohol and tobacco has garnered



support from some advocates. A meat tax in high-income countries could reduce demand and help to address the negative environmental and health-related externalities associated with meat consumption and production (Funke *et al.* 2022). Studies have looked at taxing meat products according to the intensity of their environmental impacts, including emissions, disease risk and impacts on biodiversity (Edjabou and Smed 2013; Vergeer *et al.* 2020). One concern around a meat tax is that it might unfairly affect low-income populations, but this could be addressed by a revenue recycling program (Vergeer *et al.* 2020). There are also concerns that a meat tax would not be politically feasible in all settings (Bonnet *et al.* 2020).

Against this backdrop, this report focuses on another potential intervention to address the challenge of rising demand for ASF: their substitution with novel alternatives. Proponents of novel alternatives argue that they can provide a similar or identical sensory experience to conventional meat and dairy products, while having fewer negative impacts on the environment, human health and animal welfare. The development of novel alternatives can be pursued alongside other strategies such as those described in Table 2.3 in efforts to achieve net-zero emissions and other global goals.

The remainder of this report focuses on the potential implications of the uptake of novel meat and dairy products as an alternative to conventional ASF.

3. New technological solutions are being developed to provide an alternative to animal source foods



Concerns about sustainability, food security and the environmental and public health impacts of industrial animal agriculture have spurred efforts to develop novel alternatives to conventional animal products that appeal to mainstream consumers. This has led to significant and ongoing innovations during the last decade, with today's options primarily appealing to the fast-growing segment of 'flexitarian' consumers across geographies including Australia, Brazil, China, Europe, the Russian Federation and the United States of America (Mattson 2017; BENEO Global Plant-Based Survey 2022). The novel alternatives sector aims to broaden this market to include a wider set of consumers.

This section dives deeper into three types of novel ASF alternatives; describes the production process for each product type; discusses prospects for, and challenges to, broader consumer and market uptake of these products; and considers the environmental, health, socioeconomic and animal welfare implications of these products, in comparison to conventional meat and dairy products.

3.1 Novel meat and dairy alternatives aim to mimic the taste and texture of animal source food

There are currently three main categories of novel alternatives to conventional meat and dairy products:

- Novel plant-based foods aim to replicate the sensory experience of animal products by combining plant protein (typically from soy or pea) with fats, vitamins, minerals, water and other additives (Figure 3.1). This grouping does not include more traditional plant-based meat alternatives such as tofu, tempeh, seitan, mushrooms and jackfruit.
- Cultivated meat is meat produced directly from animal cells. This is done by extracting cells from a living animal and growing them in bioreactors. Cells can be differentiated into muscle, fat and other cell types to create products with a three-dimensional structure and organoleptic properties similar or identical to those of conventional meat products (Figure 3.2).
- Fermentation-derived products are foods produced using biomass or precision fermentation. Biomass fermentation is the process of using microorganisms to make protein-rich food, where the microorganisms produced are themselves the primary ingredient. Precision fermentation uses microorganisms to produce specific functional ingredients, including proteins, vitamins and flavour molecules. These can be used in novel plantbased food to improve tests or texture, and in cultivated most to enable

based food to improve taste or texture, and in cultivated meat to enable more efficient growth (Figure 3.3).

Each of these technologies has seen significant recent and ongoing technological progress, with potential for further developments.

3.1.1 Novel plant-based meat uses new technological approaches to closely imitate the sensory profile of meat

The concept of plant-derived alternatives to ASF is centuries old and their prevalence and variety has steadily increased (Shurtleff and Aoyagi 2014). Many of these foods were designed with vegetarian consumers in mind, and early plant-based alternatives, such as tofu, tempeh and seitan, did not always try to replicate conventional meat. The early generations of 'veggie burgers' introduced in the 1960s and 1970s also were primarily aimed at vegetarians (Shurtleff and Aoyagi 2014) and their flavour profile did not necessarily seek to replicate that of meat. By contrast, novel plant-based meats are designed to match the organoleptic experience (i.e. flavour, aroma, texture, bite, moisture, mouthfeel, appearance and colour) of their animal-based counterparts as closely as possible (Kinney, Weston and Bauman 2019).

Plant proteins are the main ingredients of plant-based meat. How they are mixed and processed along with fats, and other nutrients as well as binders, flavourings (including salt) and colourants, significantly influences a product's properties (Kyriakopoulou, Keppler and van der Goot 2021). The development of a novel plant-based product involves three main steps: crop development, ingredient optimization and processing, and end-product formulation and manufacturing (see Figure 3.1 and Box 3.1 for more details).



Current novel plant-based meats on the market include beef, pork and chicken substitutes, among others. While sales of these products have been concentrated in North America and Western Europe to date, the largest sales growth is seen in Eastern Europe, Asia Pacific and Australasia and Latin America (GFI 2022a).

Figure 3.1 Novel plant-based meat



These products typically more closely resemble the sensory experience of meat than classic products, like tofu, using new scientific insights to imitate the sensory profile of meat more closely and complete the "meat-like" experience. Inputs include pea or soy protein, combined with oils and other functional ingredients, like plant-based binders, vitamins, spices or fermentation-derived ingredients, such as hemoglobin.





Box 3.1 Plant-based meat production steps

Crop development: Plant ingredients, primarily proteins (Hertzler *et al.* 2020), serve as the raw material for novel plant-based meat. While animal protein structures and functions differ significantly from those of plant proteins, plant proteins have properties that can create similar end products. For example, mung bean and chickpea proteins have superior gelling properties compared with lentil, lupin and fava bean proteins (Kyriakopoulou, Keppler and van der Goot 2021) that can improve textures in plant-based meat products. Targeted crop breeding can help further improve plant proteins' nutritional value, functionality and palatability. Until recently, most plant-based products relied on proteins from commodity crops, such as soy, wheat or pea. Other sources such as sorghum, seaweed and chickpea are now being utilized and explored, expanding the options for product development and supply chain diversification.

Ingredient optimization and processing: Processing crops to create ingredients for food products has traditionally focused on oil and carbohydrate extraction. Protein fractionation and functionalization are relatively underexplored areas that could yield innovations useful in the development of novel plant-based foods (Sim *et al.* 2021). Improvements in the ingredient processing cycle are currently being made in areas including seed dehulling and milling (Vishwanathan, Singh and Subramanian 2011) and the kinds of fractionation employed (Galves *et al.* 2019; Vogelsang-O'Dwyer *et al.* 2020). Innovations in ingredient processing can influence taste and price and make downstream processes, such as formulation and texturization, less resource- and time-intensive.

End-product formulation and manufacturing: Producers strive to transform raw ingredients into products that replicate the cookability and organoleptic properties of restructured or whole-muscle conventional meat to closely reproduce the experience of diverse meat dishes and ingredients widely consumed globally. Formulations must therefore incorporate flavourings and other ingredients that deliver the sensory palette and nutritional qualities of animal meat. In some cases, product developers may alter the nutritional profile of plant-based meat to give it an advantage over conventional meat, for example by incorporating plant fibre. Once formulated, the ingredients of a product must be structured and shaped using manufacturing processes such as stretching, kneading, shear-cell processing, press forming, folding, layering, 3D printing and extrusion (Kinney, Weston and Bauman 2019).

Proponents of novel plant-based alternatives identify several ways to

improve the products and reduce costs. These include:

- Optimizing crops for protein content, quality and function. There is potential to develop crop varieties geared specifically towards animal product alternatives. For example, crop strains could be gene-edited for traits such as protein content and functionality (Ku and Ha 2020; Nagamine and Ezura 2022). New technologies enabling high-throughput screening of crops for desired traits can accelerate the process.
- Finding uses for co-products. Rising demand for plant protein may create a surplus of plant starch and fibre. Finding uses for these co-products could reduce disposal costs and environmental impacts and generate additional revenue (Lima and Mellinger 2022). One use for co-products being explored is feedstock for protein production using fermentation technology.
- Scaling plant protein fractionation. Plant protein fractionation, or the extraction of protein from the rest of the plant, needs to scale in line with demand for plant-based meat. Recently, there has been significant investment made in scaling plant protein fractionation. For instance, several companies have set up large-scale pea protein factories to service growing global demand for ingredients optimized for novel plant proteins (Watson 2021).
- Developing and scaling whole-muscle meats. The development of plantbased alternatives to whole-muscle meat products (such as steaks or fillets), which are widely consumed across the world, is a challenge because of their fibrous textures and integrated pockets of fat. Newer manufacturing methods like 3D printing and shear cell technology are helping the industry



to move toward this goal (Rubio, Xiang and Kaplan 2020).

3.1.2 Cultivated meat is made from animal cells grown in a culture medium

Cultivated meat is animal meat that is produced by cultivating animal cells in vitro. Producers of cultivated meat aim to arrange the same cell types in an identical or similar structure to animal tissues, thus replicating the sensory and nutritional profiles of conventional meat. Dutch scientist Mark Post unveiled the first cultivated meat burger on live television in 2013. Since then, the industry has grown to include more than 100 companies across the world backed by over US\$1.9 billion in investments, while dozens more companies have formed to create technology solutions along the value chain (GFI 2022a). Cultivated chicken is the first and only product as of June 2023 to be available commercially, sold by one company in Singapore and the Unites States of America and one company in the United States alone (Firth 2020; United States Food and Drug Administration [FDA] 2023a).

The manufacturing process for cultivated meat (see Figure 3.2) begins with acquiring and banking stem cells from an animal. These cells are then grown in bioreactors filled with an oxygen-rich culture medium made of nutrients such as amino acids, glucose, vitamins and inorganic salts, and supplemented with growth factors and proteins. The result is differentiated cells, which fuse together and form muscle tissue. Cells are then harvested and prepared into final meat products.



such as fillets and chunks, many approaches to producing cultivated meat use some form of 3D scaffolding. This provides structure to the final product, facilitates nutrient, oxygen and waste transport, and provides cues that can help the cells differentiate and mature as desired.

Bioreactor

Port

000

OUTOUTS

Chicken breast

Steat

MUTTON & GOax

The bioprocess for cultivated meat encompasses production lines of bioreactors outfitted with sensor equipment, integrated with cell-harvesting and food-processing equipment, and designed with automation in mind. at high densities and volumes.

 Cells are fed an oxygen-rich cell culture medium made of nutrients, and supplemented with growth factors and proteins.

Cells are stimulated to increase protein

The result is differentiated cells, which then start to fuse together and form muscle tissue. Cells are then harvested and prepared into final meat products. This process generally takes between 2-4 weeks (Specht 2020).

* These represent different potential inputs to create cultivated meat, and not all of them will be used in products that will be commercialized. For example, fetal bovine serum (FBS) is being phased out and is likely not to be used at scale given cost concerns.

Source: adapted from Santo et al. 2020







Major technological advances will be needed before cultivated meat becomes a mass market product. Efforts to overcome these barriers are underway across four main areas:

- Cell line development: For cultivated meat to replicate the variety of conventional meat products on the market, high quality cell lines (populations of cells that can be maintained for an extended period) from many species will be required. Researchers are working to develop and characterize new cell lines and to better understand the properties of different cell types that will determine their suitability for cultivated meat. Developing livestock cell sources that possess the necessary proliferative capacity, differentiation potential and efficiency for cultivated meat production is a key technical requirement for commercial scale-up (Reiss, Robertson and Suzuki 2021).
- Culture media: Cell culture media contains the nutrients and growth factors needed to cultivate cells outside an animal's body. It is a significant driver of the cost of cultivated meat. While animal-based cell culture media such as fetal bovine serum (see Section 3.6) have been in use for roughly a century, they have not been specifically designed with the requirements of cultivated meat in mind. A common industrial use of animal cell culture is currently the production of therapeutic monoclonal antibodies, which command much higher prices than meat products. To be viable at scale, cultivated meat production requires media that is low cost, animal ingredient-free,





food-grade and food-safe, and that can regulate large-scale cell proliferation and differentiation. New approaches based on conventional culture media applications as well as industrial processes are being evaluated to advance media optimization for cultivated meat, with some success (O'Neill *et al.* 2020). There have been developments in research to develop animal ingredient-free formulations (Kolkmann *et al.* 2022; Messmer *et al.* 2022), with one product using serum-free media receiving regulatory approval in Singapore in January 2023 (GOOD Meat 2023).

- Scaffolding and other approaches to creating structure: To reproduce whole-muscle meat products such as fillets and chunks, scientists are developing the use of three-dimensional scaffolding to support and guide tissue formation. Scaffolding can also facilitate nutrient, oxygen and waste transport during growth, and provide cues to help cells differentiate and mature. Scaffolding technologies have already been developed for use in biomedical tissue engineering. However, cultivated meat production comes with constraints related to the scale and cost of production as well as the necessary attributes of the final product, such as texture and food safety. While several promising avenues have been identified, including the use of precision fermentation to produce scaffolding (Singh *et al.* 2022), more research is needed to identify scaffolds capable of supporting the growth of high-quality meat while minimizing production costs (Bomkamp *et al.* 2021).
- Production process design: The production process for cultivated meat typically envisages lines of bioreactors outfitted with sensor equipment, integrated with cell-harvesting and food-processing equipment, and

designed with automation in mind. Industry knowledge is still evolving regarding the best-suited bioreactors and technologies to develop a spectrum of cultivated meat product types (Vergeer, Odegard and Sinke 2021; Tuomisto, Allan and Ellis 2022). A variety of methods with different benefits and caveats offer potential for bringing cultivated meat production up to pilot and commercial scales (Meyer, Minas and Schmidhalter 2016). Several facilities for pilot-scale manufacturing opened globally in 2021 (GFI 2022a), and their performance could help inform industry efforts to further develop models for large-scale production.



3.1.3 Fermentation-derived products produce protein-rich food using microorganisms

Fermentation has been used in food production for millennia, for example as a preservation technique, to make alcoholic beverages and to improve the nutritional value of foods like kimchi and tempeh. Over the past century, fermentation has found a much broader range of applications including in industrial chemistry, biomaterials, therapeutics and medicine, fuels and advanced food ingredients.

The vast diversity of microbial species, coupled with significant advances in synthetic biology, suggests that fermentation has significant potential for the development of novel alternatives to conventional animal products (Specht 2020). Fermentation is used in two primary ways to create such alternatives:

• **Biomass fermentation:** With this approach, the rapid reproduction and high protein content of many microorganisms is leveraged to efficiently produce large quantities of protein. The microbial biomass itself can serve as the main ingredient of a food product or as one of several ingredients in a blend. A range of microorganisms are being explored for their potential use in biomass fermentation, from yeasts to fungi and microalgae (Banks *et al.* 2022). Several companies use filamentous fungi species, which have



an inherent meat-like fibrous texture and amenable flavour profile, as the base for their products (GFI 2023b). Some companies are also focused on microalgae that is grown without sunlight and instead fed sugar (GFI 2022a). Additionally, several companies are commercializing gas-fermented microbes that use carbon dioxide, methane or hydrogen as a carbon source to produce single-cell biomass protein (Marcellin *et al.* 2022).

• Precision fermentation: This approach refers to using microbes to produce specific functional ingredients (Teng *et al.* 2021). Precision fermentation rests on technology that is already used widely in the production of chymosin (the major enzyme in calf rennet) for cheese (Johnson 2006). In meat alternative applications, genetically modified microbes are used to produce ingredients including enzymes, flavouring agents, vitamins, natural pigments, proteins, fats and amino acids. These ingredients could improve functional attributes or sensory characteristics of cultivated or plant-based meats and have potential as scaffolding material or serum replacements in cultivated meat (Singh *et al.* 2022). Several companies are utilizing the technology to produce proteins such as casein and whey for plant-based cheeses and milks, or egg proteins such as ovalbumin for bakery applications. Some companies are producing iron-containing heme proteins, such as leghemoglobin and myoglobin, which give plant-based meats a taste and aroma similar to that of conventional meat.

Both biomass and precision fermentation involve the following high-level steps: feedstock optimization, strain development and target selection, and final product formulation and manufacturing (see Figure 3.3 for more

detail). Potential innovations in these areas include the use of high-throughput methods of strain selection to find new candidates for microbial protein production and the use of solid-state fermentation platforms to reduce risks and costs associated with scaling (GFI 2023b).

While the three categories of novel alternatives to conventional ASF have distinct characteristics, products utilizing more than one of these novel technologies may emerge. For example, plant-based meat could be combined with a small amount of cultivated meat in a product with a more meat-like sensory profile than plant-based meat at a lower cost than pure cultivated meat (Rubio, Xiang and Kaplan 2020). Hybrid products could also include conventional animal meat to increase their consumer appeal (Grasso and Jaworska 2020).

Figure 3.3 **Types of fermentation**



Fermentation approaches, both ancient and modern, provide important ingredients for our food systems. Microbes take input feedstock and nutrients and create complex outputs such as proteins, fats, flavouring agents or nutrient ingredients in a short period of time. These fermented products can range from ingredients for flavour and texture, center of plate protein portions, and even growth factors to support cultivated meat production.

Feedstock (raw ingredients)

Fermentation has been used in food production for millennia, for applications such as preservation, alcoholic beverages and improving the nutritional value and bioavailability of foods ranging from kimchi



1000 CTCS

to tempeh.



Precision fermentation refers to generating synthetic "cell factories" to produce specific functional ingredients (Teng et al. 2021). In alternative protein applications, ingredients produced using precision fermentation may be enzymes, flavouring agents, vitamins, natural pigments, proteins, fats, scaffolding material, amino acids/protein additives and serum replacements (Singh et al. 2022).

Source: adapted from Singh et al. 2022



3.2 There is a wide range of estimates for market share and consumer uptake of novel alternatives



The market for novel alternatives to conventional animal products has expanded since the first novel plant-based alternatives went on sale in 2016 (Strom 2016). Globally, as of 2021, the plant-based milk retail market had grown to US\$17.8 billion, while the plant-based meat retail market has grown to US\$5.6 billion (GFI 2022a).

In the United States of America, in 2021, plant-based meat sales were US\$1.4 billion, up 45 per cent from 2019 and amounting to 1.4 per cent of total retail meat sales. Plant-based milk, a more mature category, accounted for over 15 per cent of all American retail milk sales (GFI and SPINS 2021). These novel products show some indications of displacing their animal-based counterparts, but more research is needed to confirm to what extent this is the case. Growing consumer demand for plant-based milk may be causing cow milk sales to drop somewhat faster than they otherwise would occur (Stewart *et al.* 2020). While novel plant-based meat is less mature, in one study at an American university, adding plant-based meat to menus resulted in it comprising 11 per cent of entrée sales, while the proportion of animal-based entrée sales decreased by 9 per cent, suggesting possible displacement (Malan 2020).

In Asia, the appetite for novel plant-based meat is forecast to grow significantly, with demand expected to increase by 200 per cent in markets like China and Thailand between 2020 and 2025, and an expected 25 per cent increase more widely across the region over the same time period (Arwanto *et al.* 2022; GFI 2022a). Plant-based meat can be prepared to suit diverse food cultures and palates in markets such as Singapore, India, Brazil and Japan. Areas of the world with smaller (but growing) markets for plantbased meat are Latin America and the Caribbean, the Middle East and Africa and Eastern Europe (GFI 2023a).

Early adopters of novel meat and dairy alternatives globally tend to be 'flexitarian' — omnivores who are reducing their consumption of animal meat due to a number of reasons, including environmental concerns (Szejda, Bryant and Urbanovich 2020). However, for the majority of consumers, these reasons will not be enough for them to regularly purchase novel meat alternatives, as taste and price factors will need to be addressed (Szenderak, Fróna, and Rákos 2022). Taste and price are consistently ranked as the top food purchase drivers, above health and environmental sustainability (International Food Information Council 2023). For plant-based meat specifically, consumers stated that taste was the most important factor, followed by price (Parry and Szejda 2019).

Perceptions of cultivated meat among early adopters are positive, with younger, internet-connected, flexitarian consumers even in countries with lower affinity for meat and food technology showing high levels of acceptance for cultivated meat (Bryant *et al.* 2019). However, it is important to note that not all consumers and governments are expected to react pos-

itively to cultivated meat, with food neophobia and uncertainties around safety and health potentially dissuading some consumers (Pakseresht, Kaliji and Canavari 2022). The Italian government has proposed a draft bill that would ban the production and commercialization of cultivated meat, with the stated goals of protecting human health and preserving the national agrifood heritage (Camera dei Deputati 2023; Sabelli 2023).
As shown in Figure 3.4, forecasts for the share of alternative meat sales (across novel plant-based, cultivated and fermentation-derived products) in total global meat sales (conventional and alternative) by 2040 vary greatly and range from 4 per cent⁴ (US\$90 billion) (Powell *et al.* 2019) to 60 per cent⁵ (US\$1.1 trillion) (Gerhardt *et al.* 2019). Similarly, estimates of the share of the market occupied by each product category vary significantly. One foresees cultivated meat and plant-based meat respectively occupying 35 per cent and 25 per cent of the total meat market by 2040 (Gerhardt *et al.* 2019). Another suggests the categories of novel alternative meats will comprise 16 per cent (plant-based), 5 per cent (fermentation-derived) and 1 per cent (cultivated meat) respectively, for a total of 22 per cent of the total meat market by 2035 (Witte *et al.* 2021).

It is worth noting that predictions to date about the development and uptake of plant-based and cultivated meat have tended to be too optimistic (Dullaghan 2021; Nogueira 2023). Cultivated meat in particular faces challenges to significantly scale-up production, such as bioreactor and media costs. Consumer acceptance levels will also impact market share (Bryant and Barnett 2018; Humbird 2021; Negulescu *et al.* 2022).

for 2025–2040 (-3 per cent compound annual growth rate for conventional meat, +9 per cent for novel plant-based alternatives, and +41 per cent for cultivated meat).

Figure 3.4 Global alternatives to conventional meat industry forecasts by year



Source: GFI 2023c

Notes: *Some forecasts projected share of the total meat market rather than the industry size in dollars. For those forecasts, we estimated the dollar size of the alternative protein sector using EY's forecast for the total 2030 meat market. (Maeder *et al.* 2023)

The wide range of these estimates reflects the difficulty of making predictions in the early stages of the development of these novel technologies. Forecasts of significant uptake are predicated on a range of drivers across the areas of scientific and technological innovation, commercial and consumer trends and acceptance, and regulatory and government action. Workforce development is also important for the sector's growth (Specht *et al.* 2020).

Increased government support could help these products achieve taste and price parity with conventional equivalents and improve their accessibility to customers. A recent report estimated that US\$10.1 billion in annual government support for research and manufacturing could see novel ASF alternatives take a 50 per cent market share by 2050 (ClimateWorks) Foundation and Foreign, Commonwealth and Development Office, UK Government 2021). A techno-economic assessment of cultivated meat found that at a production cost of US\$2.92 per pound (US\$6.43 per kg), cultivated meat could be cost-competitive with some conventional meats by 2030 and serve as an affordable ingredient for plant-based and cultivated meat blends. Achieving this, however, assumes that producers can hit multiple process-optimization targets, and build facilities with the capacity to produce 10,000 tons of cultivated meat annually with an estimated cost of USD 450 million each (Swartz 2021; Vergeer, Odegard and Sinke 2021). Other analyses indicate that for plant-based meat to capture six per cent of the global meat and seafood market by 2030 (25 million metric tons of annual plant-based meat production), the industry would need to operate at least 800 model manufacturing facilities—each producing on average 30,000 tons of product annually—at a cost of at least US\$27 billion (GFI 2023b). For reference, in 2022 alone, the United States of America invested US\$1 billion to expand processing capacity for independent meat processors (United States, The White House 2022).

⁴ Low end of the Jeffries forecast (Powell et al. 2019) assumes low overall growth in meat consumption (2 per cent annually from 2018–2040) due to health concerns, weak consumer confidence in novel alternatives due to product recalls, production techniques constrained by regulators, and no enactment of a meat tax.

⁵ A.T. Kearney forecast (Gerhardt et al. 2019) assumes +3 per cent annual growth for overall meat consumption

Figure 3.5 Breakdown of venture investments by company's country, technology type and type of product



Source: FAIRR Initiative 2022. Note: The data refers to deals for the period 2020-2021. **Plant-based**

By contrast, factors that could nudge novel meat and dairy alternatives onto a slower growth path include a lack of public and private investment in innovation and production to solve persistent technical challenges, a lack of clear regulation that impedes innovation, food safety issues or product

recalls impacting consumer sentiment, and obstacles related to labelling or other roadblocks to technological progress or consumer acceptance (Gerhardt *et al.* 2019). Regulatory paths to market are particularly relevant, while global regulatory harmonization and development of standards will be important in obtaining religious certifications (e.g. halal, kosher). Additionally, cultural acceptance and trust of novel alternatives will be key to their adoption. For example, targeted gender-responsive communications may help foster trust and acceptance of this technology. As young men are more open to try cultivated meat in some countries, and men also comprise most meat consumers, this could be one avenue for driving behavioural change (Boereboom *et al.* 2022). Women may be more willing to opt for cultivated meat if informed of food safety advantages, for example if a product is described as free of antibiotics (Piochi, Micheloni and Torri 2022).

Reaching price parity will be a key driver of customer adoption. Price parity depends not only on reducing production costs for alternatives, but also on market developments that affect conventional meat costs. Higher input (feed) costs, meatpacker labour issues, wage inflation, avian influenza outbreaks and supply chain interruptions have recently drawn attention to the volatility—and price instability—of conventional meat (Ijaz *et al.* 2021; United States Department of Agriculture Economic Research Service 2022). Given the importance of reaching price parity, some cultivated meat companies are expected to initially launch hybrid products with plant-based meat to reduce costs (Watson 2020; Sawers 2022).

3.3 Novel meat and dairy alternatives have potential environmental benefits compared to conventional counterparts

3.3.1 Alternatives could reduce greenhouse gas emissions and use less land and water

Comparing the potential environmental impacts of novel meat and dairy alternatives to those of conventional farming is not straightforward. This is in part due to the variety of systems that could be considered 'conventional' farming—from extensive pasture-fed systems to intensive feedlots (Santo et al. 2020)—and their very different inputs and impacts. The assessments considered in this sub-section generally make comparisons primarily based on life cycle assessment (LCA) data of the typical conventional production systems of specific regions. These include intensive industrialized, extensive free-range and organic systems, and various combinations of these. Most of the assessments examine industrial and free-range systems in Europe and North America, but do not generally assess nondominant systems, such as smallholder farming systems, nomadic pastoralism and agropastoralism.

As discussed in Section 2, the role of different livestock production systems in generating different environmental outcomes is under discussion, with regenerative farming systems offering potential for some environmental gains. It thus matters what farming system the novel alternative product is being compared to. Overall, however, it is important to underline that diets in nations with high per-capita meat consumption tend to largely depart from the regenerative farming model, and that such models cannot meet the demand associated with current levels of meat consumption.



Another challenge for comparison is that the environmental impacts of meat and dairy, as well as of their novel alternatives, differ from one product to another. Much existing research focuses on ruminant farming and GHG emissions, where impacts from beef production are greater than for other livestock (de Vries and de Boer 2010), and the potential environmental gains from replacement, therefore, stand to be greater.

Furthermore, while various novel plant-based products are already available, and their environmental impacts relatively well understood, there are difficulties assessing the impacts of emerging technologies, particularly cultivated meat, where environmental analyses are currently limited. However, efficient feed conversion is a key characteristic that novel alternatives to animal products share, meaning that they require fewer crops per unit of end product (Sinke et al. 2023). ASF face greater conversion losses because feed also contributes to the energy needed to move, forage, stay warm and alive and grow non-meat organs (e.g. bones). Less feed translates to less land, fertilizer and pesticide use, resulting in lower scores on several environmental impact categories. To varying degrees, alternative meat and dairy products may require more energy than conventional meat and dairy to produce, so that the source of this energy-fossil fuels or low-carbon-becomes important in comparing impacts.



novel alternatives to conventional animal products, existing insights are





3.3.2 Novel plant-based meat

While plant-based meat products typically incur a greater degree of processing and associated energy use than conventional plant-based foods, life cycle assessment (LCAs) suggest that, compared to conventional beef, they could use 30–50 per cent less energy whilst offering reductions of 86–97 per cent in land use and 67–89 per cent in GHG emissions (Rubio, Xiang and Kaplan 2020, Saget *et al.* 2021). Conventional pork and chicken also show substantially greater land use than novel plant-based meats (Rubio, Xiang and Kaplan 2020).

Given the significant land-use savings, replacing conventional meat with novel plant-based alternatives could make more land available for biodiversity conservation and restoration, renewable energy generation or less intensive agro-ecological approaches (Balmford *et al.* 2019; Newton and Blaustein-Rejto 2021; Tuomisto, Allan and Ellis 2022; Sinke *et al.* 2023). If used for restoration of natural habitats, it could contribute to achieving the goals of the post-2020 Global Biodiversity Framework as well as contributing to climate change mitigation and adaptation.

Regarding water, footprints for both conventional and plant-based alternatives are highly variable. For plant-based meat it is highly dependent on their main sources of protein and the manner of processing (Fresán *et al.* 2019,

Potter *et al*. 2020).

It is important to note that novel plant-based meat comprises a range of products, each with specific characteristics and production contexts that will determine their impacts. The processing needed for some novel plant-based alternatives means that unless their energy mix is decarbonized, they can have a similar—or potentially higher—GHG footprint to some conventional meats, such as chicken and pork (Goldstein *et al.* 2017; Rubio, Xiang and Kaplan 2020). On the other hand, if low-carbon energy sources are used in the production of alternatives, GHG emissions tend to be lower than for conventional animal products (Tuomisto, Ellis and Haastrup 2014; Sinke *et al.* 2023).



3.3.3 Cultivated meat

Current research suggests that cultivated meat will have greater climate impacts than plant-based ASF alternatives particularly if low-carbon energy sources are not used (Tuomisto et al. 2022), and that they will have higher resource use (Smetana et al. 2015; Smetana et al. 2023). However, grow-ing meat from cells could be more sustainable than conventional livestock rearing in terms of land and water and produce fewer GHG emissions if low carbon energy is used (Table 3.1).

To date, only seven anticipatory LCAs of cultivated meat have been published in peer-review scientific articles to quantify these possible benefits. Six of these LCAs are included here (Table 3.1).⁶ The wide variation in projected

impacts between these studies can be attributed to the methods employed, including the choice of system boundaries, data and hypothesized system processes. Smetana et al. (2015), for example, calculated the highest GHG emissions and energy requirements of all the studies due to changing the system design as outlined by Tuomisto and Teixeira de Mattos (2011) to cultivate the feedstock and cyanobacteria in a closed system rather than the previously modelled open-air pond. They also adopted a cradle-to-plate system boundary which includes some of the environmental impacts past the factory gate, such as transportation, packaging, storage, and cooking, which all other studies exclude. Also, Tuomisto and Teixeira de Mattos (2011) calculated cultivated meat's water footprint using both blue water (surface and ground) and green water (rainwater), whereas Mattick et al. (2015b) and Smetana et al. (2015) chose to exclude water use from their system boundaries entirely. Another methodological consideration is whether the varying nutritional content of different novel products are accounted for (in the LCAs in Table 3.1, only Smetana et al. (2015) included nutritional metrics).

⁶ One of the 7 cultivated meat LCAs, Sinke and Odegard (2021) was updated and republished in 2023 as Sinke *et al.* (2023). Only the peer-reviewed, 2023 version has been included in this report.

For instance, where one kg of certain plant-based alternatives may contain less protein than one kg of animal-based meat, meaning additional volumes of the plant-based product would be required to achieve nutritional parity, there is not enough evidence currently to assess comparison when made on the basis of nutrient-based LCA.

Due to the rapid development of cultivated meat technology and the lack of any proven commercial-scale production processes or facilities, accurately modelling the environmental impacts of high-volume cultivated meat production remains a challenge. However, LCA studies suggest that if renewable energy sources are used with the most carbon efficient production techniques, GHG emissions could achieve 2.3 kg CO_2 -eq per kg of meat⁷ (median; range 1.9-4.88 kg CO2-eq per kg of meat) on a cradle-to-gate basis (Tuomisto and Teixeira de Mattos 2011; Tuomisto, Ellis and Hastrup 2014; Mattick *et al.* 2015b; Tuomisto, Allan and Ellis 2022; Sinke *et al.* 2023). This is around 40 times less that estimates of emissions from conventionally produced beef of an estimated 26.2 to (99 kg CO_2 -eq per kg of meat), and around a quarter of the global average emissions for conventionally produced chicken and pork (9.87 and 12.31 kg CO_2 -eq per kg of meat, respectively) (Poore and Nemecek 2018).

The LCAs included here show high variations in GHG emissions volumes (Table 3.1) which, in conjunction with other methodological differences, can



be largely attributed to their modelling of production systems with different energy sources and mixes. Energy sourcing has a substantial impact on the overall environmental impacts of cultivated meat. As emissions from cultivated meat are primarily anticipated to be in the Scope 1 and 2 categories, single actions such as powering facilities with renewable instead of conventional energy mixes are significantly more effective in attaining reductions than these actions could be for conventional animal agriculture, where most emissions are categorized as Scope 3,⁸ therefore requiring a range of tools and actions to achieve similar emission reductions (*Sinke et al.* 2023).

As discussed in Section 3.1, cultivated meat products may include hybrid products, containing both cultivated meat and plant-based proteins. Such products could reduce the environmental impacts of alternatives, particularly during the cultivated meat industry's initial stages of technological development and commercial expansion. This potential has been outlined in a recent LCA (Kim *et al.* 2022) on a hybrid burger. The product, of which cultivated meat constituted 16.9 per cent of the total mass, was found to have life cycle environmental impacts comparable to those of commercially available novel plant-based burgers.

Compared to conventional meat production, all of the LCAs for cultivated meat foresee drastic reductions in land use of 97–99 per cent per kg for beef, 60–99 per cent per kg for pork and 43–98 per cent per kg for chicken (Table 3.1). Applying a more specific measure of land-use efficiency, the feed conversion ratio (FCR) of cultivated meat is estimated to be three times more efficient than conventionally farmed chicken, currently the most environmentally efficient form of livestock production (Sinke *et al.* 2023).

⁷ In this section, the word 'meat' refers to a generic, ground cultivated meat product. Meat types, such as chicken or beef, are not specified in the existing LCAs. Further detail can be found in Table 3.1.
⁸ Scope 1 indicates GHG emissions from sources owned or controlled by the reporting entity; Scope 2 indicates indirect emissions associated with the production of electricity, heat or steam purchased by the reporting entity; Scope 3 indicates all other indirect emissions, i.e. emissions associated with the extraction and production of purchased materials, fuels and services, including transport in vehicles not owned or controlled by the reporting entity, etc. (IPCC 2014).

Authors	Year	Institution	Title	System boundary	Meat type	Greenhouse gas emissions ⁹ (CO ₂ - eq per kg)	Energy use (MJ per kg)	Land use (m ² a per kg ¹⁰)	Water Use (L per kg)
Sinke, Swartz, Sanctorum, van der Giesen and Odegard	2023	CE Delft; GFI; GAIA	Ex-ante life cycle assessment of commercial scale cultivated meat pro- duction in 2030	Cradle-to- gate	Cultivated meat product from land- based animal with 20–30 per cent dry matter content and 18–25 per cent pro- tein content	2.21–24.8 (ReCiPe 2016 ¹¹)	116.48– 481.70 (Cumulative Energy Demand (CED))	2.25–3.59 (ReCiPe 2016)	60-150 (Blue water and green water)
Tuomisto, Allan and Ellis	2022	University of Helsinki; University of Bath	Prospective life cycle assessment of a bioprocess design for cultivated meat production in hollow fiber bioreactors	Cradle-to- gate	Cultivated meat consisting of skele- tal muscle tissue of unspecified species with 30 per cent dry matter and 20 per cent protein content	4.88–25.19 (ReCiPe 2016)	94.09– 532.78 (CED)	1.84–6.89 (ReCiPe 2016)	120—540 (Blue water)
Smetana, Mathys, Knoch and Heinz	2015	German Institute of Food Technologies; University of Vechta	Meat alterna- tives: Life cycle assessment of most known meat substitutes	Cradle-to- plate	Unspecified	23.9–24.64 (ReCiPe 2008)	290.7–373 (ReCiPe 2008)	0.39–0.77 (ReCiPe 2008)	Not reported
Mattick, Landis, Allenby and Genovese	2015	University of Texas; Clemson University; Arizona State University; University of Minnesota	Anticipatory life cycle analysis of in vitro biomass culti- vation for cultivated meat production in the United States	Cradle-to- gate	Cultivated meat product	3.15–22.28 (CML 2001 ¹²)	43.46– 315.88 (CED)	2.92–8.47 (Ecological footprint)	Not reported
Tuomisto, Ellis and Haastrup	2014	European Commission Joint Research Centre; University of Bath	Environmental impacts of cul- tivated meat: Alternative produc- tion scenarios	Cradle-to- gate	Ground cultivated meat product with 30 per cent dry matter content and 19 per cent protein content	2.3–3.4 (GWP ¹³ 100y - IPCC 2006)	38.7–60.9 (Energy Conversion Factor calculations)	0.46 (Not stated)	516.4 (Blue water – Pfister et al. (2009) water scarcity method)
Tuomisto and Teixeira de Mattos	2011	University of Oxford; University of Amsterdam	Environmental impacts of cul- tivated meat production	Cradle-to- gate	Ground cultivated meat product with 30 per cent dry matter content and	1.9–2.2 (GWP 100y - IPCC 2006)	26–33 (Not stated)	0.19–0.23 (Not stated)	367-521 (Water Footprint

Table 3.1 Summary of different life cycle assessments for cultivated meat



content

(Hoekstra, et al. 2009)

⁹ Large ranges for GHG emission estimates within individual studies exist primarily due to the modelling of production scenarios with different energy compositions, including renewable sources, nonrenewable sources and mixes of both.

¹⁰ In LCA, land use is expressed as 'm²a crop-eq' which accounts for both land-use occupation and land-use transformation.

¹¹ A method in life cycle impact assessment that uses characterization factors, which indicate the environmental impact per unit of stressor.

¹² A database that contains characterization factors for life cycle impact assessment.

¹³ Global warming potential.



3.3.4 Fermentation-derived products

Studies of the environmental impacts of novel food products from precision and biomass fermentation (Table 3.2) suggest that, overall, they have lower land use and water impacts than conventionally produced proteins, while potential reductions in greenhouse gas emissions are highly dependent on the source of energy used in their production (Sillman *et al.* 2020; Järviö *et al.* 2021; Perfect Day Inc. 2021); the gains can be significant if low-carbon energy is used.

Promising but not yet commercially available techniques such as methanotrophic and hydrogen-oxidizing chemosynthetic bacteria have the potential to deliver further land use and climate benefits in mycoprotein production, with the magnitude again dependent on the availability of low-emission energy sources (Sillman *et al.* 2020; Järviö *et al.* 2021; Humpenöder *et al.* 2022). Perunit replacement of ruminant meat with more conventional techniques, such as from mycoprotein, already has the potential to reduce water and land use by 90 per cent and GHG emissions by 80 per cent (Hashempour-Baltork *et al.* 2020; Rubio, Xiang and Kaplan 2020), though the environmental benefits of replacing pork and poultry with mycoprotein are less clear (Smetana *et al.* 2015; Souza Filho *et al.* 2019; Rubio, Xiang and Kaplan 2020).

Authors	Year	Institution	Title	System boundary	Product	GHG emissions (CO ₂ -eq per kg)	Energy use (MJ per kg)	Land use (m ² a per kg ¹⁴)	Water use or scarcity (L per kg)
Behm, Nappa, Aro, Welman, Ledgard, Suomalainen and Hill	2022	VTT Technical Research Centre of Finland; Fonterra Research and Development Centre; AgResearch Ruakura Research Centre; Riddet Institute, Masey University	Comparison of carbon footprint and water scar- city footprint of milk protein produced by cel- lular agriculture and the dairy industry	Cradle-to- gate	Precision fermented microbial milk protein (beta-lac- toblobulin)	5.5–17.6	Not reported	Not reported	88–5031 (water scarcity)
Perfect Day Inc.	2021	Perfect Day Inc.	Comparative life cycle assessment of Perfect Day whey protein produc- tion to dairy protein	Cradle-to- gate	Precision fermented whey protein (Trichoderma reesei)	2.71	56.3	Not reported	73.9 (water use, scar- city not reported)
Järviö, Maljanen, Kobayashi, Ryynänen and Tuomisto	2021	University of Helsinki; Natural Resources Institute Finland	An attributional life cycle assessment of microbial protein production: A case study on using hydrogen-oxidizing	Cradle-to- gate	Biomass fermented hydrogen-ox- idising bacteria (Xanthobacter)	0.52–7.83	101.2- 240.2	0.003- 0.047	1–2.5 (water scarcity)

_	-	_	-	_	_	-	

bacteria

Sillman,	2020	Lappeenranta-Lahti	A life cycle environmental	Cradle-to-	Biomass	0.82-1.15 63.54	0.029-	0.8-3.8
Uusitalo,		University of Technology	sustainability analysis of	gate	fermented		0.085	(water use)
Ruuskanen,		(LUT) University; VTT	microbial protein produc-		hydrogen-ox-			
Ojala,		Technical Research Centre	tion via power-to-food		idising bacteria			
Kahiluoto,		of Finland	approaches		(Cupriavidus			
Soukka and					necator)			
Ahola								

¹⁴ In LCA, land use is expressed as 'm²a crop-eq' which accounts for both land-use occupation and land-use transformation.

3.3.5 Transparency is important for realizing all potential environmental benefits

As with all nascent technologies, it is difficult to accurately assess all the potential gains or risks resulting from the expansion of novel ASF alternatives until production operations are scaled up. On the one hand, bigger efficiency gains than those foreseen in current LCAs could result from greater scale and further investment in research and development. Other promising avenues to reduce life cycle impacts include the utilization and valorisation of by-products and waste streams, for example, as feedstocks for fermentation processes, for producing sugars for use in cultivated meat; or for conversion into ingredients for other foods (Koutinas *et al.* 2014; Smetana *et al.* 2018). Additional opportunities include diversifying feedstock to use locally available resources and reduce transportation distances (GFI 2022a).

On the other hand, while promissory narratives are necessary to attract investment in emerging technologies, they may also run the risk of overpromising. Transparency from the sector to share the assumptions behind their data is vital. Guthman (2022) invites reflection on the promise of greater resource efficiency that biotechnological solutions offer against the similar promises of industrial agriculture. Biotechnological solutions hold promise for the environment, but it is not a given. Ultimately, every step of the supply chain for ASF alternatives—from feed and energy production through to consumption and the processing of waste—must be scrutinized and the necessary safeguard measures put in place to fully realize potential benefits (Section 4 explores how policy can help to secure more sustainable outcomes).

3.4

Alternatives could reduce the risk of zoonotic diseases, antimicrobial resistance and foodborne illnesses; further research on nutritional implications is needed

Comparing the public health implications of novel ASF alternatives with those of their conventional counterparts should be considered across multiple dimensions, including risks of zoonotic diseases, antibiotic resistance, and foodborne illnesses, nutritional implications and impacts on workers and communities.

3.4.1 Reduced risk of zoonotic diseases, antimicrobial resistance and foodborne illnesses

A reduction in ASF production and consumption in favour of ASF alternatives is likely to lower the risks of zoonotic disease transmission, given decreased human-livestock interactions, habitat destruction and biodiversity loss (Santo et al. 2020). In general, livestock production contributes more to biodiversity loss than crop production for direct human consumption (Machovina, Feeley and Ripple 2015; Sun et al. 2022). ASF reductions will also reduce the risk of human exposure to antibiotics as well as the development of antimicrobial resistance. This is because crop agriculture uses much lower levels of antibiotics than livestock agriculture and aquaculture, accounting for an estimated 0.26–0.5 per cent of total agricultural antibiotic consumption in the United States of America, for example (Taylor and Reeder 2020). Meat alternatives created under sterile lab conditions should, in theory, also require much lower levels of antibiotic use than in industrial livestock operations (even if some antibiotic use during cell line development is still expected), whilst also reducing the risks of foodborne illnesses (Santo et al. 2020; FAO and WHO 2023).

3.4.2 Nutritional implications

The picture for nutritional health is more nuanced and unclear. Given the limited evidence to date on novel plant-based alternatives, further research is urgently needed to elucidate their full short- and long-term public health implications, including the potentially differentiated outcomes based on factors such as sex and age, in order to inform relevant policies and consumer choices. While ample evidence from longitudinal studies and randomized controlled trials (RCTs) conducted in adult populations suggests that dietary patterns that generally emphasize minimally processed, healthful plantbased foods (i.e. whole grains, vegetables, fruits, nuts, seeds and legumes) are associated with reduced risks of premature mortality and NCDs, including cardiovascular disease, type II diabetes, obesity and some cancers (Schwingshackl *et al.* 2017; Kim, Caulfield and Rebholz 2018; Clark *et al.* 2019; Kim, Willett *et al.* 2019; English *et al.* 2021), significant knowledge gaps remain in understanding the nutritional health implications of replacing intake of conventional ASF with novel plant-based alternatives (Santo *et al.* 2020; Wickramasinghe *et al.* 2021). In particular, two key lines of enquiry with respect to consumer health warrant further investigation: nutritional quality and production formulation and their respective non-communicable disease (NDC) outcomes.

Animal and plant proteins are both nutritious, and protein quality and quantity are often not compromised when shifting to a well-designed diet that is predominantly or wholly plant-based (Tso and Forde 2021). Still, the majority of novel meat alternatives rely on soy, wheat or pea protein isolates as their primary protein source; it is unknown whether this form of protein would offer similar nutritional benefits or NCD reductions as whole legumes (Hu, Otis and McCarthy 2019; Flint et al. 2023). In addition to protein quantity and quality, there is much else to consider regarding the nutritional quality and interchangeability of novel alternatives compared to traditional ASF, such as macronutrients (protein, carbohydrate, fat); micronutrients (vitamins and minerals); the absence, low quantity or reduced bioavailability of certain important micronutrients that are naturally found in traditional ASF such as iron, zinc and vitamin B12; metabolite composition; the content of health-sensitive nutrients (fat, sugar and sodium) and other added ingredients; and the method of preparation required for consumption (Tso and Forde 2021; van Vliet et al. 2021; Flint et al. 2023; Nolden and Forde 2023).



Novel meat alternatives can have comparable levels of protein, saturated fat and sodium to some traditional meat products, but no cholesterol and higher levels of dietary fibre (Bohrer 2019). To date, variations in both the study methods and in the available alternative products mean that the limited published evidence is inconclusive regarding the nutritional profile of novel meat alternatives compared to traditional ASF (Flint *et al.* 2023).

In general, novel plant-based products that mimic the sensorial qualities of ASF tend to be highly processed¹⁵ and can have high amounts of sodium and saturated fats (Gehring et al. 2021; Tso and Forde 2021; Wickramasinghe et al. 2021; Flint et al. 2023). While food processing is not inherently bad and can help to improve product palatability and enable fortification, there is strong evidence linking consumption of ultra-processed foods with greater caloric intake and weight gain, as well as increased risks of a range of adverse health outcomes including cardiovascular disease and premature death (Hall et al. 2019; Lawrence and Baker 2019). High sodium intake is also the leading dietary health risk worldwide, especially amongst young men (Afshin et al. 2019). However, there is no evidence linking these adverse health outcomes to processed, novel plant-based alternatives, and the few RCTs that have been conducted to date found positive health outcomes. Results from a 16-week RCT conducted in California, United States of America with 36 participants suggest that replacing ASF intake with novel plant-based alternatives may lower several cardiovascular disease risk factors and body weight in healthy adults (Anderson and Bradley 2020; Crimarco et al. 2020). Another recent RCT found positive changes in the gut microbiome when substituting several meat-based meals weekly for modern plant-based alternatives (Toribio-Mateas, Bester and Klimenko 2021).

¹⁵ The NOVA system classifies all foods and food products into four groups according to the extent and purpose of the industrial processing they undergo (Monteiro *et al.* 2019). Under this system, many plant-based alternatives can be classified as 'ultra-processed foods (UPFs)', which are defined as 'formulations of substances derived from foods, such as starches, sugars, fats and protein isolates, with little, if any, whole food, and often with added flavours, colours, emulsifiers and cosmetic additives' (Wickramasinghe *et al.* 2021). However, some have argued that the NOVA classification system is too simplistic and does not adequately evaluate the nutritional attributes of certain foods (Messina *et al.* 2022).

Even though fermentation has been used to process food for millennia and is known to generally increase its nutritional value and gut-microbiome-enhancing properties (Xiang et al. 2019), there is limited research on the health effects of foods produced via the more novel techniques of biomass and precision fermentation. Consumption of mycoprotein derived from fungal biomass fermentation has been associated with certain health benefits such as improved blood levels of cholesterol, sugar and insulin (Ahmad et al. 2022; Souza Filho et al. 2019). However, some commercial vegan or vegetarian products utilizing mycoprotein can be highly processed and contain high levels of salt (Bohrer 2019). Some plant-based meat alternatives contain heme iron produced by precision fermentation of genetically engineered yeast; while its health effects have not yet been studied, high intake of molecularly identical heme from red and processed meat has been associated with increased risks of type II diabetes, cardiovascular disease and cancers (Santo et al. 2020). That said, heme iron is only naturally found in ASF and is more bioavailable than non-heme iron found in plant-based foods; consumption of heme iron can therefore help to address iron deficiency concerns especially for women of reproductive age and pregnant and lactating women (Zimmermann and Hurrell 2007).

Since cultivated meat alternatives are not yet widely commercially available, there is little evidence available about their public health implications, including, for example, whether they will be biochemically similar to traditional meat given possible differences between post-mortem transformations in conventional meat and post-harvest transformations in cultivated meat (Fraeye *et al.* 2020; Santo *et al.* 2020).





In the absence of sufficient evidence, it would be incorrect to assume that all plant-based foods are inherently healthy. Some researchers have raised concerns that the established health benefits of traditional plant-based foods are being conflated with positive messaging around animal welfare and sustainability benefits for the newer alternatives, many of which come in the form of fast-food product categories such as burgers, nuggets, meatballs and sausages (Tso and Forde 2021; Flint et al. 2023; Nolden and Forde 2023). At the same time, these alternatives designed to mimic the sensory attributes of meat are more likely to displace or reduce traditional ASF rather than whole-plant foods in a person's diet (Gastaldello et al. 2022). Traditional ASF can also come in heavily processed forms and with high levels of sodium, and there is strong evidence linking high intake of red and processed meats to adverse health outcomes (see Section 2.3). Therefore, despite knowledge gaps about the nutritional health implications of novel ASF alternatives, it would also be incorrect to assume that all conventional ASF products are healthier.

Ultimately, the nutritional health implications of all novel ASF alternatives will vary with each product's specific formulation, nutritional composition and density, and what they are replacing in an individual's diet, as well as with physiological differences in the consumer such as age and sex

(Hu, Otis and McCarthy 2019; Tso and Forde 2021; Marwaha, Beveridge and Phillips 2022). Further research is also needed to assess the specific impacts on pregnant women, children and unborn children. Opportunities and further technological innovations exist for developers to enhance the nutrient quality (such as in terms of macronutrient and micronutrient composition and density) while minimizing or eliminating undesirable substances (such as cholesterol, unhealthy fats, salt and sugar, antibiotics and hormones) and to diversify the formulation of all novel ASF alternatives (Flint *et al.* 2023; Nolden and Forde 2023).

3.4.3 Impacts on workers and communities

To date, little is known about the existing or potential occupational health risks in the modern plant-based food industry (including novel plant-based foods, cultivated meat and fermentation); conditions are likely to be less hazardous than those in an industrial meat processing factory, but workers could still be exposed to dangerous chemicals used in the manufacturing process, such as hexane used in soy protein isolate production (Xiang *et al.* 2019; Santo *et al.* 2020). In terms of the health risks for nearby communities, issues associated with water contamination from nutrient runoff and use of pesticides that exist in today's crop production would persist (ibid). Still, modern plant-based alternatives are expected to generate less waste compared to ASF production and require fewer inputs (e.g., water, pesticides, fertilizers) for the same amounts of calories and protein (Rubio, Xiang and Kaplan 2020; Santo *et al.* 2020).

3.5 Socioeconomic impacts will depend on the degree

A transition from conventional ASF to novel alternatives would likely be accompanied by significant socioeconomic impacts, including in areas such as rural employment and food security. The nature and magnitude of these potential impacts depends greatly on factors including the speed and degree of adoption of novel alternatives and the extent to which they replace or complement conventional ASF (see Section 3.2 for uncertainties in this regard).

of uptake of alternatives; more research is needed to fully understand these implications



To date, only a limited amount of research has evaluated the socioeconomic implications of novel ASF alternatives. Moreover, the available studies disproportionately focus on ethics and consumer acceptance (Stephens *et al.* 2018). Nevertheless, some researchers have begun to consider the wider impacts. Possible challenges include lost market and job opportunities for incumbents, agribusiness consolidation and further globalization of the food system, and unanswered questions about intellectual property rights of new technologies (Santo *et al.* 2020; Howard 2022; IPES-Food 2022). The literature also discusses opportunities these industries could bring, including new jobs and the redistribution of production to new areas and countries (Treich 2021).

Promoting a just transition will require policies that consider the distribution of these challenges and opportunities, paying particular attention to vulnerable and disadvantaged groups such as women, ethnic minorities, Indigenous Peoples and migrants, so inequalities are not further widened (Treich 2021; Verkuijl *et al.* 2022; see also Section 4). Inclusive planning and support involving policymakers, workers, communities, industry actors and other relevant stakeholders will be key in designing such policies (ibid).

Growth of the plant-based and cultivated meat industries may create more high-skilled jobs in certain regions (Treich 2021). ClimateWorks Foundation and the Foreign, Commonwealth and Development Office, UK Government (2021), suggests that, under a high innovation scenario, alternative proteins (including alternative dairy) could capture roughly half the global market for animal protein, have sales of US\$1.1 trillion and employ about 10 million people globally by 2050. At the same time, changes in the distribution of labour may result in increases in unemployment in some sectors with impacts on livelihoods (Mason-D'Croz *et al.* 2022). Specifically, a significant reduction in conventional ASF production could drive major losses of employment in the animal farming and meat processing sectors. This will affect livestock and animal feed producers, farmworkers, meat processors and meat packing workers, among others (Mouat and Prince 2018; Stephens *et al.* 2018; Verkuijl *et al.* 2022).

Still, growth in the plant-based meat sector could create additional demand for various crops as sources of plant proteins, in turn creating opportunities for farmers that currently grow those crops or who could adopt them into their rotations (Verkuijl *et al.* 2022). Additionally, novel alternatives might in some cases create new opportunities for conventional livestock farming, for example to supply meat for blended products or raise livestock to supply cell lines for cultivated meat production (ClimateWorks Foundation and Foreign, Commonwealth and Development Office, UK Government 2021).

Rural populations could suffer if plant-based and cultivated meat shifts food production to cities, leading to losses in livelihoods, culture and traditions for rural producers and communities (Johnson and Lichter 2019; Pender *et al.* 2019; Morais-da-Silva *et al.* 2022). At the same time, alternative ASF production could create new jobs and opportunities in urban and peri-urban areas.

There are many uncertainties about how these novel food industries would be structured. They could present new opportunities for small businesses that target local food preferences (van der Weele and Driessen 2013; van der Weele and Tramper 2014; Stephens *et al.* 2018; Jönsson 2020; Treich 2021). However, potentially high barriers to entry (e.g. capital costs, intellectual property rights) and significant economies of scale in production could provide powerful incentives for industry consolidation (Morrison Paul 2003; Langemeier and Boehlje 2017; MacDonald 2017).

Across the conventional ASF sector there will be disparities in the adaptive capacity to respond to a shift towards novel alternatives. Most cultivated meat and plant-based product companies are owned by agribusinesses or start-ups headquartered in industrialized countries (Mouat, Prince and Roche 2019). Their success could further entrench economic and political power disparities between the Global North and Global South (Howard 2022; IPES-Food 2022). Additionally, large food companies, that have already invested heavily in the novel foods sector, could secure significant intellectual property rights related to novel foods, further concentrating their market power. The number of food patents filed globally for meat substitute products has been rising in recent years, mostly filed by large firms in the Global North (ibid). The extent of intellectual property rights that will be involved in the cultivated meat sector is unclear, though developments in seed patenting may offer some insights (Santo *et al.* 2020).

Novel alternatives have the potential to affect global supply chains, changing comparative advantages and the direction of trade flows of agricultural commodities. Trade in conventional ASF has grown significantly in recent decades, but is dwarfed by that in animal feed (Galloway *et al.* 2007). Nations where conventional animal products and feed make up a large portion of exports or of the economy could be vulnerable to a shift in global demand for agricultural commodities, in particular countries that rely on agricultural exports for foreign currency (Mancini and Antonioli 2022; Morais-da-Silva *et al.* 2022). By contrast, countries with little domestic animal agriculture may be able to replace some protein imports with domestically produced novel ASF alternatives and even benefit from export opportunities, with countries such as Israel and Singapore already investing in alternatives (Kamalapuram, Handral and Choudhury 2021; Morach *et al.* 2021; GFI 2023a).

The consequences of the adoption of novel alternatives for food security is difficult to assess *ex ante*. Some studies suggest that they could increase the availability of protein-rich foods to meet the nutritional needs of more people (Wan, Tai and Du 2021; Zhu and Begho 2022). However, poverty is a major driver of food insecurity (Sen 1983; FAO 2022), and as previously noted novel alternatives may disrupt the livelihoods of rural populations that are disproportionately poor and work in agriculture (Castañeda et al. 2018; Davis, Lipper and Winters 2022). Further research is needed to understand potential food security implications of novel ASF alternatives more fully, as outcomes may depend on the specific dynamics of local supply chains. Complementary agricultural and social support policies may be needed to support communities to adapt to change. This could include increased investment in areas including social safety nets, agricultural research and extension services (Chichaibelu et al. 2021), as well as incentives to ensure that freed-up resources such as land and water are used for activities that benefit the environment and food security.

3.6 Alternatives promise significantly reduced animal welfare issues

Replacing traditional ASFs with alternatives would significantly address the animal welfare issues associated with conventional livestock production (discussed in Section 2), by reducing the current need to raise, keep and slaughter billions of animals every year to supply products such as meat and milk. This is particularly the case for novel plant-based and fermentation-derived alternatives, which do not require the use of animals for their production.

In contrast to the other two categories of novel alternative, cultivated meat has, in its early development stages, relied on the use of animals, thereby raising a number of animal welfare concerns.

Cells for cultivated meat so far predominantly originate from biopsies performed on living animals. According to Melzener *et al.* (2020), a biopsy to obtain stem cells for cultivated beef muscle may involve removing small amounts of muscle tissue from an immobilized, sedated and locally anaesthetized cow with a needle or by making an incision. The animal's discomfort from a needle will probably be comparable to that from the taking of a blood sample, while for an incision biopsy, the level "will likely be greater, but still not substantial". It remains to be seen how many samples are taken per session, and how frequently an animal will be subjected to biopsies. In addition, competitive pressures could potentially lead to some similar forms of maltreatment to those commonly experienced by animals in the conventional livestock industry. It is also possible that animals from the conventional livestock industry would be among the providers of these biopsies.

Another ethical issue is the fate of animals no longer able to deliver effective cell samples. Research suggests that the muscle tissue of young (3-monthold) bulls yields significantly more stem cells than older or female animals (Melzener *et al.* 2021). A strong emphasis on animal welfare could involve allowing active and retired donor animals to live out their natural lives in "sanctuary-like" habitats (Dutkiewicz and Abrell 2021). However, economic and environmental incentives might motivate people to slaughter them before this, either to harvest their remaining useful cells or for conventional meat (Melzener *et al.* 2021).

Crucially, however, the number of animals required to meet global meat demand would be dramatically smaller. In theory, cells taken from a single donor animal during twenty biopsies over their lifespan could yield cultivated beef equivalent to the meat of 400 cattle. Optimizing the cell proliferation process could see a single biopsy eventually replace millions of cattle (Melzener *et al.* 2021). Researchers are seeking to develop "immortalized" cell lines that can proliferate indefinitely (Reiss, Robertson and Suzuki 2021; Soice and Johnston 2021b) and could eliminate the need for repeated biopsies.

Cultivated meat production has also relied on fetal bovine serum (FBS) as a growth medium. FBS is taken from the fetus of a slaughtered pregnant cow through cardiac puncture without anaesthesia. It affects both the pregnant cows who are transported and slaughtered, and the fetuses who may already be mature enough to feel pain during and after the cow's slaughter (Jochems *et al.* 2002; Weber *et al.* 2021). Researchers are developing animal-free growth media for reasons of animal welfare as well as high costs and technical challenges associated with FBS (Messmer *et al.* 2022). In January 2023, a United States of America-based start-up received approval in Singapore to produce cultivated chicken based on serum-free media (GOOD Meat 2023).

Overall, novel ASF alternatives from plants and fermentation avoid the massive-scale animal welfare issues of conventional animal products. Cultivated meat has the potential to be similarly beneficial, though it still faces animal welfare challenges that companies are working to address through ongoing research and development (Chen *et al.* 2022).

Figure 3.6 Number of people a cow can feed for a year



Conventional animal farming Around 5 people: a family

UN ENVIRONMENT PROGRAMME FRONTIERS 2023 SPECIAL ISSUE





Cultivated meat, early development Around 2,000 people: a small village



Cultivated meat, mature technology Around 100 million people: a large country



optimized, although this number could even be substantially higher (Melzener et al. 2020).

4. Policy and regulatory environments can significantly influence the future of alternatives

Governments, mainly in high- and middle-income countries, have begun to explore policies and regulations to actively govern novel alternatives to conventional animal products. This may entail considering whether and how to accelerate their development and deployment, as well as ensuring that their emergence supports, rather than threatens, progress towards other societal goals. Evidence-based policy processes that include stakeholders are vital to ensure a transition towards more sustainable food systems that is just (Duncan *et al.* 2022; Webb *et al.* 2023).

The emergence of alternative proteins, including novel ASF alternatives, presents economic opportunities for some countries, and may present challenges to others; particularly those that may have constrained access to key inputs including energy, feedstocks, capital and technology necessary for a competitive alternative protein sector. Current plant-based products use crops like peas and soy that certain countries already have a strategic advantage in producing, and the development of a larger, higher-value market for these crops could support local economies and livelihoods. In addition, other crops may become viable inputs to novel plant-based products, with several governments investing in research and development for such new functionalities (Agritech Future 2022; Nutraceutical Business Review 2022; Change Foods 2022). Nations that are small, densely populated, unsuited for agriculture or otherwise dependent on imported food or feed may find promise in cultivated meat and fermentation technologies that enable the production of dietary protein without the associated land-use requirements (Hayek et al. 2021). As a global market develops for alternative proteins and ancillary products, such as bioreactors and cell-growth media, governments may also consider the advantages of fostering the development of a high-value addition sector.

4.1 Current policies impacting novel meat and dairy alternatives

The policy and regulatory environment is evolving quickly in response to the emergence of novel ASF alternatives. Several high- and middle-income countries have begun to take steps to enable and promote the development and uptake of these products. Conversely, other countries have adopted a more cautious or even oppositional stance towards novel ASF alternative, including restrictions or bans on their production or sale. The reasons for these restrictions may be multifaceted, including food safety concerns, societal or cultural resistance, lobbying by conventional ASF producers, the impact on traditional agriculture sectors, or knowledge gaps and uncertainty about these novel food products.

In order to share evidence of safety and help avoid potential misinformation, producers of novel alternatives such as cultivated meat companies can be encouraged to provide open, independently verified research on product safety and nutrition, given that the rigor and transparency of approval processes may vary by country (Holmes *et al.* 2022). Such independent veri-

fication could be funded through a mix of public and private financing.

An overview of recent policy and regulatory developments surrounding novel ASFs (Tables 4.1 and 4.2) indicates how some governments are responding to novel foods, including ASF alternatives.

Several countries have developed approval processes for novel foods, including carefully worded definitions of the term, to provide a pathway for producers to bring their products to market, including an obligatory review of their safety and nutrition by regulatory agencies (Table 4.2). These novel food frameworks typically cover cultivated meat and some fermentation-derived ingredients, but most novel plant-based products are exempt, given that most plants used in these products already have a precedent set for being used as food.

Some countries in both the Global North and Global South have moved actively to understand and foster the potential of novel ASF alternatives, for instance through support for research and pilot production facilities.

Some countries have adopted a more precautionary stance towards alternative ASFs. Plant-based cheeses have been banned in Türkiye (Southey 2022), while Italy has approved a draft bill that, would ban the production and commercialization of cultivated meat. The bill was presented as a measure to ensure "the protection of human health and the interests of citizens as well as preserving the agri-food heritage and its strategic importance for national interest" (Senato della Repubblica 2023; Sabelli 2023).

Table 4.1 **Policy instruments** targeting alternatives to conventional animal products

Funding open-access science and/or	* * *			★** **	
research centres	Australia	Brazil	Canada	China	Denmark
		_		_	
	**** * * ***				X
	The European Unio	n France	Germany	India	Israel
				(***	
	Japan	The Netherlands	Norway	Singapore	Republic of Korea

Example countries

Remarks

Organizations implementing research and development initiatives are mainly national or provincial public science and technology organizations and public-private partnerships in food technology research.



The United of Great Britain States and Northern Ireland of America

Funding industry and/or manufacturing

Policy instrument

* * * * *	*	★** **		**** * * ***
Australia	Canada	China	Denmark	The European Union
			XX	
Finland	France	Germany	Israel	Japan
	¥		(*** **	
The Netherlands	Oman	Qatar	Singapore	Republic of Korea
Spain	Sweden	The United Th Arab Emirates an	e United Kingo of Great Britai d Northern Irel	lom The United n States and of America <i>(including state governments)</i>

These policy instruments include direct public production or investments in public-private production ventures and information campaigns as well as legal and market facilitation mechanisms. Government investments largely focus on strategic domestic industries, e.g. pulse-based ASF alternatives production in Denmark and Canada and fermentation infrastructure in Israel and Finland. A number of state-owned sovereign wealth funds, e.g. in Singapore, Oman and Qatar, are also making sizable investments in domestic novel alternative companies and/or projects.

Regulation



Labelling, market transparency and consumer

trust-building campaigns are handled by public sector agencies. Other regulatory instruments include developing novel food approval processes and banning the production or sale of certain alternatives.

Financial incentives and instruments



Tax exemptions and other incentives to producers of novel alternatives, such as subsidies and support for inputs (e.g. protein-rich crops), market development.

Workforce development

These policy instruments include professional training and the development of curricula to build knowledge and skills regarding novel ASF production and sustainable agri-food systems.

Table 4.2 **Regulation of novel foods** in selected countries

Brazil

The Ministry of Health is the main authority responsible for food regulation and safety assessments. It performs this role via the autonomous National Agency of Sanitary Surveillance (Agencia Nacional de Vigilancia Sanitaria [ANVISA]), which evaluates the safety of additives and ingredients in foods and regulates labelling.

Australia and New Zealand

A joint regulatory body, the Food Standards Australia New Zealand (FSANZ), maintains food standards that apply in both countries. The Australia New Zealand Food Standard Code regulates areas including labelling, additives, processing aids, nutritive substances, contaminants and the approval of new foods, including novel foods.

Pre-market approval by ANVISA is required for novel foods and ingredients. Approvals are valid for five years and may be renewed (ANVISA 2022). Regulations in Brazil and other members of Mercosur, the South American regional organization, are increasingly aligned. Brazil is gradually replacing many of its food standards with official Mercosur standards, which are in turn influenced by European and US standards as well as the global Codex Alimentarius.

Canada

The Food Directorate, a federal authority under the government's Health Canada department, is responsible for setting standards, policies and regulations and providing advice and information on the safety and nutritional value of food.

Canadian regulations (Government of Canada 2023) define a novel food as:

- **a** a substance, including a microorganism, that does not have a history of safe use as a food;
- **b** a food that has been manufactured, prepared, preserved or packaged by a process that
 - has not been previously applied to that food and
 - causes the food to undergo a major change; and
- **c** a food that is derived from a plant, animal or microorganism that has been genetically modified such that:
 - the plant, animal or microorganism exhibits characteristics that were not previously observed in that plant, animal or microorganism,
 - the plant, animal or microorganism no longer exhibits characteristics that were previously observed in that plant, animal or microorganism, or
 - one or more characteristics of the plant, animal or microorganism no longer fall within the anticipated range for that plant, animal or microorganism.

Pre-market approval is required for a new substance, for an extension of the use of a permitted food additive,

FSANZ defines a novel food as a non-traditional food that requires an assessment of the public health and safety considerations having regard to:

- **a** the potential for adverse effects in humans; or
- **b** the composition or structure of the food; or
- c the process by which the food has been prepared; or
- **d** the source from which it is derived; or
- e patterns and levels of consumption of the food; or
- **f** any other relevant matters. (Australian Government 2016)

Non-traditional food is defined as a food or substance that has no history of human consumption in the two countries (Australian Government 2016).

FSANZ approved soy leghemoglobin, a fermentation product used in plant-based meat, as a nutritive substance and a genetically modified food in 2020. The regulator received its first application for pre-market approval of cultivated meat in early 2023 (FSANZ 2023).

China

The Ministry of Health is responsible for drafting laws, regulations, policies and standards governing food and for supervising enforcement.

Novel or 'new resource' foods include raw materials or ingredients that have no significant history of consumption in the country (GFI 2023a).

'New food additives' require pre-market approval from the ministry. Legislation defines a new food additive as one not covered by the national food safety standards or approved for a permitted use by the ministry, or whose scope of use or dosage is increased (GFI 2023a).

European Union

for permission to change the maximum level of a permitted food additive, or for authorization to add a new organism to the list of permitted sources of enzymes used as a food additive. The submission requires evidence that the food is safe for consumption.

In 2020, the Canadian Food Inspection Agency, which is part of Health Canada, sought public input on the labelling of plant-based meat alternatives. The resulting guidelines require labels to use the phrase 'simulated' followed by the meat or poultry type and the phrase 'contains no meat' or 'contains no poultry' in close proximity (Government of Canada 2020).

United States of America

US authorities do not specifically define or regulate novel foods. The Food and Drug Administration (FDA) considers any new food ingredient either as a food additive (requiring its pre-market approval) or 'generally recognized as safe' for specific uses (as classified by a panel of experts and determined independently of the FDA).

The FDA completed its first pre-market consultation for human foods made from cultivated animal cells in 2022 (FDA 2022). Such consultations are a step toward the formal approval of cultivated meat products, which also usually involves the USDA's Food Safety and Inspection Service. Terrestrial meat, poultry and catfish are considered livestock species, and therefore fall under the jurisdiction of the USDA, while cultivated fish of other species only requires FDA clearance. In 2023, USDA approved labelling and issued grants of federal inspection for cultivated meat for the first time, allowing two companies to begin sales of cultivated chicken.

United Kingdom of Great Britain and Northern Ireland

The European Union updated its food regulations in 2015 to reflect developments in food science and technology. The new regulations define novel food as 'food that has not been consumed to a significant degree by humans in the EU before 15 May 1997', when the first regulation on novel food was published. In addition, a food product should be considered novel food 'when it results from a production process not used for food production within the Union before 15 May 1997, which gives rises to significant changes in the composition or structure of a food affecting its nutritional value, metabolism or level of undesirable substances' (Official Journal of the European Union 2015).

The EU regulation expanded the categories of novel foods to include, among others, whole insects and their parts, 'food with a new or intentionally modified molecular structure, as well as food from cell culture or tissue culture derived from animals, plants, microorganisms, fungi or algae, food from microorganisms, fungi or algae and food from material of mineral origin' (Official Journal of the European Union 2015, European Food Safety Authority Panel on Dietetic Products and Allergies 2016).

In the regulation, it is specified that the main purpose is to ensure a high level of protection of human health and consumers' interests.

Food businesses are responsible for determining whether their products are novel. The European Food Safety Authority (EFSA) carries out risk assessments on novel foods and the European Commission decides on authorizations (Lähteenmäki-Uutela et al. 2021). As of mid-2023, the authority has not received any applications for cultivated meat products (Sabelli 2023).

Singapore

The Singapore Food Agency's Guidance on Safety Assessment of Novel Foods (2023) specifies the information required to seek approval for a cultivated meat product.

The United Kingdom of Great Britain and Northern Ireland defines novel foods as those 'not used for human consumption to a significant degree' in the United Kingdom or the European Union before 15 May 1997 (Food Standards Agency 2023).

Examples include:

- new foods, such as phystosterols and phytostanols used in cholesterol-reducing spreads
- Foods traditionally eaten in other countries, such as chia seeds
- Foods produced through new processes

There are two routes for new foods to gain approval for sale:

- Traditional food notification: a simplified route to authorize products that have been continuously used for 25 years by a significant number of people in a country outside the United Kingdom of Great Britain and Northern Ireland or European Union.
- Full application: a more complex route for foods that don't have a history of consumption in other countries. A number of documents need to be presented and should follow European Food Safety Authority guidance on the preparation and presentation of an application for authorization of a novel food.

The regulator received its first application for pre-market approval of cultivated meat through its novel foods process in mid-2023 (Cervera 2023).

The agency defines novel foods as 'foods and food ingredients that do not have a history of safe use' (Singapore Food Agency 2023). Substances with a history of safe use are those that have been consumed as an ongoing part of the diet by a significant human population (e.g. the population of a country), for a period of at least 20 years and without reported adverse human health effects.

In 2020, Singapore became the first country to approve the sale of a cultivated meat product.

South Africa

The Department of Agriculture, Land Reform and Rural Development (DALRRD) regulates plant-based meat products through the Food Safety Agency. The Department of Health's food control section additionally ensures food safety as an essential public health function and conducts risk assessments related to agricultural chemicals and food produced through biotechnology (DALRRD 2022; South Africa, Department of Health 2022).

In August 2022, DALRRD (2022) published new regulations regarding the classification, packaging and labelling of processed meat products intended for sale. The regulations define 'meat analogue' (also known as a meat substitute, mock meat or imitation meat) as products that:

- a approximate the aesthetic qualities (primary texture, flavour and appearance) and/or chemical characteristics of a specific type of meat; and
- **b** are made from non-meat ingredients and are available in different forms such as coarse ground meat analogues, emulsified meat analogues and loose fill.

Regulators face challenges regarding how novel foods are marketed. Whether novel ASF alternatives should be allowed to carry words such as 'meat', 'steak', 'burger' or 'milk' has sparked intense public and political discussion in several countries, and policymakers have reached widely varying decisions (Gleckel 2020; Taylor 2022; Demartini *et al.* 2022). Initial evidence suggests that labelling plant-based alternatives with these words does not cause consumers to think these alternatives come from animals, as some proponents of labelling restrictions have claimed, and that using these labels reduces confusion for consumers about their taste, potentially improving uptake (Gleckel 2020). Regarding cultivated meat and precision fermented animal proteins, it will be important to have clear labelling for those with allergies, given that those who are allergic to a certain type of ASF may also be allergic to its cultivated meat or fermented counterpart (Bryant 2020).

The regulation of cultivated meat has drawn considerable public scrutiny. Though designed to be (closely) identical to conventional meat at the cellular level, whether it is considered meat in law is a question that Singapore and the United States of America have so far answered in the affirmative. In 2020, Singapore became the first country to approve the sale of a cultivated meat product – cultivated chicken (Ives 2020). In the United States of America, oversight of cultivated meat from terrestrial animals is shared between the Food and Drug Administration (FDA) and the Department of Agriculture (USDA). As of June 2023, two producers have received approval to sell cultivated chicken products in the United States of America, having completed a premarket consultation with FDA, received approval on proposed labelling from USDA, and received a standard grant of inspection from USDA (Aubrey 2023; FDA 2023). The United Kingdom's Food Standard Agency, the Switzerland Federal Food Safety and Veterinary Office and Food Standards Australia New Zealand, the joint food regulator for the two countries, have all established that cultivated meat should be regulated through novel foods frameworks and have received applications for cultivated meat product through this process (United Kingdom, Food Standards Agency 2023; Food Standards Australia New Zealand [FSANZ] 2023; Morrison 2023). Government agencies in other countries have yet to determine how to regulate it. In Japan, authorities are developing a regulatory framework for cultivated meat while working with the private sector on industry standards to create consumer confidence (Derbes 2021). Canada, the European Union and Israel are developing pathways for cultivated meat products within their novel food frameworks (Table 4.2).

Precision fermentation-derived products may be covered within existing regulatory frameworks, given that the technology has a history of safe use in food and pharmaceutical applications (e.g. production of rennet and human insulin), and modern products are commonly ingredients with an established history of safe consumption (e.g. dairy and egg proteins). Nevertheless, precision fermentation-derived ingredients may also require approval as novel foods in some cases.

4.2 Potential policies to support novel alternatives to conventional animal source foods

Many countries have expressed an interest in promoting novel foods. These are predominantly high- and middle-income countries. Governments in countries including (but not limited to) Australia, Canada, China, the Netherlands, Israel, Singapore and the United Kingdom of Great Britain and Northern Ireland have declared support for and devised strategies to develop new food technologies, with stated policy goals ranging from food security to sustainability to economic opportunity (Mok, Tan and Chen 2020; Teng 2020; Canada, Agriculture and Agri-Food Canada 2022; Australian Government 2022; Ettinger 2023a; TU Delft 2022; United Kingdom Research and Innovation 2022). China and the United States of America—the world's two largest economies and significant players in the global livestock industry—have both expressed support for these technologies. China has explicitly included cultivated meat and other 'future food' in the Five-Year Plan of its Ministry of Agriculture and Rural Affairs (People's Republic of China Ministry of Agriculture and Rural Affairs 2021) and expressed plans to further develop these technologies (Liang and Lee 2022). The United States of America included alternative proteins in its national planning through the March 2023 'Bold Goals for U.S. Biotechnology and Biomanufacturing' report, in which USDA and the Department of Energy identify them as priorities to address climate, sustainability and food security issues (United States, The White House 2023).

There are several complementary approaches that governments can use to promote these technologies and their uptake. A range of policy options are discussed below. Some are already being pursued by a number of governments, while others are not yet widespread but could be considered by policymakers as part of a broader toolkit to support more sustainable and just food systems.

4.2.1 Policies targeting producers

Higher production and uptake of novel ASF alternatives depends on food producers perceiving the market as profitable and reliable, and then delivering products that can compete with conventional equivalents in terms of taste and price (Section 3.2). Governments can help producers reach these goals through research funding, commercialization funding and the development of appropriate regulatory frameworks.

Government support for research and development: research can be performed by government research agencies or contracted out to other institutions and should deliver open-access results. Examples of research priorities include breeding or engineering crops for higher protein yields, developing novel methods for texturizing plant-based proteins and developing bioreactors capable of supporting high-density and large-volume cell cultures (see also Section 3.1). For example, in 2022, France's national research agency called for proposals to investigate novel uses for French-grown crops in plant-based meat and to develop new functionalities for microbes, algae and insects – priorities that would benefit France's economy and food systems (France, Agence Nationale de la Recherche 2022). In the United States of America, the USDA granted US\$10 million to Tufts University to develop a centre of excellence in cellular agriculture (Nicholas and Silver 2021). Meanwhile Israel, among other projects, has provided a grant to a private cultivated meat company to develop an open-access screening system for cultivated meat inputs to reduce duplicative research and accelerate the pace of innovation (Shoup 2022).

Research performed to date, including that funded by governments, has resulted in a number of commercially successful products and ingredients, lower production and consumer costs and foundational technological developments. However, further research is needed to maintain progress in the field and to reach taste and price parity with conventional animal products. Governments around the world dedicated at least US\$180 million in 2022 to research and development to improve novel ASF alternatives and associated production technology (GFI 2023a). However, to realize the potential for innovation to yield high quality alternative proteins, one recent analysis estimates that governments need to commit US\$4.4 billion per year on research and development (for classic and novel plant-based products, fermented products, cultivated meat and insects) (ClimateWorks Foundation and Foreign, Commonwealth and Development Office, UK Government 2021). While private companies often invest in research and development, open-access public sector research enables the state to shape the market by focusing on areas that could benefit multiple actors or serve public interest, for example by identifying uses for domestic resources in alternative ASF products, developing alternatives to high-cost inputs, or devising lower-emission production processes (Holmes et al. 2022).

Government support for commercialization: governments can assist producers in establishing production facilities and infrastructure and in marketing newly developed products. Subsidies and tax concessions can reduce costs and incentivize actors across the supply chain, including farmers, equipment makers and distributors as well as manufacturers. For example, Denmark has announced a five-year programme that will pay 580 million kroner (US\$81 million) in bonuses to farmers who grow input crops for plantbased products (Denmark, Ministry of Food, Agriculture and Fishers 2021).

Policy instruments like tax rebates, direct financial investments and loan guarantees to enhance producer interest in the space can also promote competition by reducing barriers to entry, for instance by reducing up-front capital costs of production facilities.

The aforementioned ClimateWorks Foundation and Foreign, Commonwealth and Development Office, UK Government (2021) analysis estimates that governments worldwide need to commit US\$5.7 billion per year to commercialization in order to realize the potential benefits of alternative proteins.

Developing appropriate regulatory frameworks: legal and regulatory frameworks are critical for allowing companies to bring novel foods such as ASF alternatives to market in a timely fashion. Novel foods approval processes differ by jurisdiction but usually require producers to prove the safety of their products. This can include performing animal testing, convening independent boards of experts or running full-scale production runs before market approval. Given that some of today's ASF alternative products are unique among foods in their formulation and production methods, review processes can take significant time. However, as the sector develops, regulators can consider adapting approval processes to support efficiency, while ensuring that safety standards are maintained. Producers of novel ASF alternatives are encouraging governments to work with industry experts to determine how to evaluate the safety of cultivated meat in ways that support innovation and reduce uncertainty in this area (GFI 2022).

In the longer term, governments could consider developing approval processes tailored to specific types of products (e.g. cultivated meat) outside of their respective novel foods processes. This would allow applicants to better understand the relevant approval processes and regulators to specialize in this category. For example, FSANZ in Australia and New Zealand and Health Canada in Canada already collaborate on genetically modified foods, with one agency performing the safety assessment and the other serving as a peer reviewer, reducing duplicative efforts for both regulators and applicants (FSANZ 2022). Similar opportunities to streamline approval processes for ASF may emerge. Beyond developing more streamlined approval processes, governments can play a critical role in setting up independent mechanisms to analyse and compare environmental and health claims for novel ASF alternatives.

4.2.2 Policies targeting consumers

As well as supporting the supply of ASF alternatives, government policy can stimulate consumer demand.

Labelling: information on food labels can make alternative products more attractive. Currently, most consumers choose their foods based on cost and taste (Danley 2020). However, many novel ASF products have not yet reached price and taste parity with conventional animal products. Until then, consumers will need some other reason, such as sustainability, nutritional value or animal welfare, to choose these products. Moreover, the importance of environmental and ethical considerations to consumers may grow, for instance if price and taste differences between conventional ASFs and novel alternatives shrink. Governments who want to promote more sustainable consumption could look to develop clear labelling standards, in particular to facilitate comparison between similar products, where labels have been shown to be most effective in informing consumer choice (Bauer and Reisch 2019).

Public information campaigns: beyond labelling, consumer acceptance of novel ASF products will depend in part on the information people are exposed to (Post et al. 2020). Governments could coordinate public information campaigns (potentially in partnership with industry) to inform consumers. Specifically, they could provide scientifically grounded information on the relative environmental impacts of ASF and various alternatives to guide consumer choices towards more sustainable products. They could also choose to address consumer concerns around the nutritional value of novel ASF alternatives. There is precedent for this type of campaign; for example, in the United States of America, programmes run through the USDA exist for beef, dairy and numerous other commodities, using industry funds to sponsor consumer marketing programmes (USDA 2011). The European Commission provides tens of millions of euros each year to promote agricultural products from the EU, including animal meat products (Boffey 2020). Working with the media can be valuable in this regard: there are indications that media attention can shape public perceptions of food technologies, including cultivated meat (Bryant 2020). Nevertheless, these activities may face competing messaging from parts of the food industry (Sexton et al. 2019). Governments should work to ensure that consumers receive accurate, complete and relevant information on new technologies and products to reduce the potential for misleading or incorrect information.

As well as stimulating private demand, governments can use public procurement to support the purchase of novel alternatives in settings such as public schools, hospitals and government offices. This could include advance market commitments to purchase given quantities of products in advance of their commercial availability (GFI 2021; Systemiq 2023). Like any significant societal transition, a transition from conventional ASF towards novel alternatives presents various uncertainties and risks regarding its scale, speed and impacts. Government decisions and industry actions can help to maximize environmental, social, health and animal welfare benefits and minimize harms from the rise of this new industry.

Environmental measures: there are likely several environmental benefits associated with shifts away from conventional ASF consumption towards novel ASF alternatives, as detailed in Section 3.3. These include potential reductions in the use of resources such as land and water. Whether this potential is fully realized will depend on how the food system and overall economy respond to the structural changes created by the adoption of alternative proteins, and if freed resources are reallocated to other economic activities (Mason-D'Croz *et al.* 2022). To increase the likelihood of realizing environmental gains, governments can strengthen policies to encourage the conservation and sustainable use of land and other resources. Such approaches would be consistent with agreed international targets on climate change and biodiversity.

Furthermore, there remain a range of potential, non-negligible environmental impacts from the alternatives industry itself. These include impacts

from inputs (e.g. farming of ingredients such as soy, peas, coconut oil and cocoa butter requiring land, fertilizer, chemicals, water and energy), to waste streams generated by alternative protein production processes (Coca 2023; Holmes *et al.* 2022). The novel alternatives industry has an incentive to be seen as a leader on sustainability and could work to secure a leadership role in sustainable supply chains. However, major alternative protein companies currently lack sustainable sourcing policies (Coca 2023).

In addition to voluntary actions by industry, government measures can play an essential role in supporting more sustainable outcomes. For example, as discussed in Section 3, alternatives to ASF are likely to offer slight to significant GHG emission reductions depending on production methods and supportive public infrastructure. Government policies can help to ensure potential emission reductions are realized by, for instance, transitioning electricity grids to renewable sources and encouraging sustainability-focused innovation in food technology development. Such shifts would also be consistent with those needed to achieve the goals of the Paris Agreement.

Social measures: shifting away from a farmed animal-centred food system to one that features alternative ASF will bring changes to the agricultural workforce. Moreover, (rural) communities may be affected by significant changes in incomes and livelihoods as well as in food cultures and social norms. Consumers may also be affected by food prices changes, depending on how both novel and conventional food markets develop.

The principles of a just transition encourage governments to work with

stakeholders to design policies that will help to minimize disruptions and maximize benefits for people affected. While the need for a just transition towards a sustainable energy system and recommendations for ecologically and socially just transitions from the prevalent food-energy-water system (Giampietro *et al.* 2013) have been articulated, the need to act immediately has gained traction more recently (Anderson 2019; Verkuijl *et al.* 2022; Just Rural Transition 2023).

Structural changes associated with novel alternatives may favour urban economies at the expense of rural ones more dependent on agriculture and conventional ASF production. Policies can be targeted to help alleviate some of these disruptions. For example, governments can provide more robust social safety nets that can increase citizens' resilience and adaptive capacity, while helping to ensure economic access to healthy diets (Barrett *et al.* 2022). Farmers and landowners facing economic losses could be supported in return for the provision of public goods such as carbon storage, biodiversity conservation and valued landscapes (Verkuijl *et al.* 2022). Investments and policies can pay special attention to vulnerable population groups, including ethnic minorities and migrant workers, who make up a significant part of the workforce of the meat supply chain in many countries (Eurofound 2019; Fremstad, Rho and Brown 2020). Governments can support research designed to gain a better understanding of how the proliferation of novel alternatives will affect food security in different contexts (Section 3.5) and tailor policies accordingly.

As some of the world's largest corporations enter the alternative ASF space through investment or acquisition, there are concerns about further consolidation of the food system in the hands of a few powerful actors who can strongly influence products and prices, while smaller producers are further marginalized (Treich 2021; Howard 2022; IPES-Food 2022). There are also concerns that centralized production of ASF alternatives may be vulnerable to supply chain disruptions and deliver homogenized products that fail to meet local dietary preferences (Soice and Johnston 2021a).

Governments could proactively explore options to prevent strong consolidation of the market, including actions to restrict takeovers that could

endanger innovation and fair competition (de Streel and Larouche 2015). Governments can also support open-access research to reduce barriers to entry related to intellectual property and encourage a broader landscape of companies to engage with this emerging industry (Holmes 2022). They could also support smaller-scale companies and decentralized production to create potentially more responsive and resilient supply chains (Soice and Johnston 2021a). More broadly, governments could re-evaluate competition policy to assess if it is incentivizing behaviour that contributes to social welfare and environmental objectives (Dolmans 2020; Holmes 2020; Malinauskaite 2022).

Health measures: It is important for government regulators to ensure the application of robust and transparent safety standards and approval procedures to ensure that novel ASF alternatives, some of which involve processes and formulations new to the food industry and require exacting standards of hygiene, are safe to eat and deserving of consumer confidence (see Section 4.1).

As discussed in Section 3, novel alternatives are likely to have certain health benefits over conventional animal products, particularly in areas such as reduced risk of zoonoses and antimicrobial resistance. At the same time, questions remain about the nutritional value of (certain) novel alternatives (Section 3.4) and the consequences of regularly consuming ultra-processed foods (Pagliai *et al.* 2021). Governments can proactively address these concerns by supporting research designed to enhance the nutritional content and monitoring the health consequences of novel foods. For example, the USDA's aforementioned grant to a university will help support research to improve the nutritional content of cultivated meat, among other goals (Nicholas and Silver 2021). Other support could also include government funding for randomized control trials focused on assessing the nutritional implications of alternatives.

Animal welfare measures: There is increasing recognition that justice considerations in food systems transitions must go beyond humans and extend towards non-human animals (Kaljonen *et al.* 2021). Novel plant-based meat and fermentation-derived products do not rely on the use of animals. However, cultivated meat requires on innovations to eliminate the need for animals, including repeated biopsies and animal-based growth media. Replacing FBS as a growth media is a key priority.

Overall, animal welfare benefits from novel alternatives can only be gained if they replace conventional animal products, rather than being consumed alongside them. Investment in novel alternatives could consider current animal welfare impacts in order to prioritize the replacement of conventional products with the most severe animal welfare impacts (Scherer *et al.* 2018).

4.2.4 Policities to support an enabling environment

The current food system has been shaped significantly by longstanding policy choices and market dynamics that tend to favour conventional ASF. These include subsidies for conventional farming, a general absence of pricing for environmental externalities, regulatory exemptions and government-backed advertising support, all of which serve to keep the price of meat artificially low and social and environmental externalities high, thereby distorting the competitive landscape (Boffey 2020; Pieper, Michalke and Gaugler 2020; FAO *et al.* 2021; McCormack 2021). There are several measures governments can take to help address or prevent such distortions and give sustainable foods, including alternatives to conventional meat and dairy, a fair opportunity to establish themselves and gain market share:

Addressing existing fiscal support to conventional animal products: governments can review and revise current fiscal policies to ensure that they are contributing to wider social and environmental objectives. This could involve

48

repurposing subsidies and tax breaks to encourage healthier and more environmentally sustainable outcomes (Gautam *et al.* 2022; Springmann and Freund 2022). Given the links between conventional ASF and negative health and environmental outcomes, this could include phasing out or reducing fiscal support for conventional ASF production and consumption. Poultry, pork, mutton and beef are among the foods that benefit most from government support, with the majority of such support going to industrial production (FAO *et al.* 2021). An analysis of major EU and United States of America agricultural policies between 2014-2020 found that public funding for novel technologies is smaller than that for animal products by factors of 1,200 in the EU and 800 in the US. Approximately 1000 times more public funding went to livestock farmers, compared to plant-based and cultivated meat (Vallone and Lambin 2023).

Reflecting externalities in pricing: governments could incentivize more sustainable and healthy food systems by better internalizing environmental and health externalities in the prices of products (Hendriks et al. 2023). Such interventions could be levied at different parts of the supply chain, including point of sale (e.g. in supermarkets), point of production (e.g. farms), or point of processing (e.g. meat processing companies). Based on data from 2019, the environmental and human health costs of the United States of America's food system have been estimated at close to double the value of food expenditure at market prices, with conventional ASF significant drivers of these external costs (Rockefeller Foundation 2021). Taxes like those on sugary drinks have proven effective in reducing the consumption of unhealthy sodas in Mexico, the United States of America and the United Kingdom of Great Britain and Northern Ireland (Lancet Editorial Board 2020; Scarborough et al. 2020). Modelling studies have also suggested that taxes on red and processed meat could contribute to improved health and environmental outcomes (Springmann et al. 2018), particularly if combined with subsidies and price supports for healthier foods (Springmann et al. 2017). At the same time, as discussed in Section 2, there are risks that taxes targeting environmental costs (e.g. carbon taxes) may introduce inequalities for consumers (Vergeer et al. 2020), or have consequences relevant to animal welfare, if they change relative prices of animal products, encouraging increased consumption of smaller animals such as chickens and fish (Springlea 2022). This underscores the need to consider the impacts of policy interventions holistically across different societal goals.

Avoiding bans and unnecessarily restrictive labelling: emerging alternatives to ASFs face regulatory restrictions in several regions. These include restrictions or bans on the use of terms such as 'meat', 'steak', 'burger', 'sausage' and 'milk' in the labelling and marketing of such products.

Although research is limited, it is conceivable that labelling restrictions will make it more difficult for alternatives to compete with conventional animal products, for instance if this leads to segregated product placement (e.g. in a vegetarian sector in the grocery store) (Piernas *et al.* 2021), or to labelling that is less attractive to consumers. In the case of some regulations, such as production and sales bans, this may be the policy's explicit goal.

4.2.5 Multilateral Cooperation

The development of novel ASF alternatives will likely benefit from international collaboration. Several complementary avenues are available to governments:

Collaboration on research and development: governments can collaborate in the research and development of novel ASF alternatives. For example, the Singapore-Israel Industrial R&D Foundation, a collaboration between the countries' entrepreneurial development agencies, awarded a grant to two companies (one based in each country) to jointly develop 3D-printed cultivated fish (Steakholder Food Ltd 2023). Another example is a 2023 grant funded by research agencies from Sweden and Austria, along with the EU, awarded to two companies (one based in each country) to develop a 3D-printed mycoprotein product (Ettinger 2023b). Attempts to work beyond bilateral efforts with multi-state organizations to coordinate research could further reduce duplicative efforts and accelerate the pace of innovation globally.

International support: If there is demand from recipient countries, collaboration with and support from international financial institutions or bilateral cooperation, in partnership with the private sector, could also be considered to build expertise and/or establish production capacities for novel ASF alternatives in middle- and low-income countries. Such an approach would mirror other forms of international sustainable development support, technology transfer and other forms of cooperation in areas such as energy and climate.

Supportive trade policy: international trade plays a critical role in many agricultural value chains and can make global food systems more resilient to shocks (Baldos and Hertel 2015). Nevertheless, international trade can also facilitate the offshoring of the environmental burden of production (e.g. embedded emissions) and contribute to the globalization of food systems and diets associated with negative health and environmental outcomes (Hawkes 2009). Phytosanitary measures are important protections for human (e.g. food safety) and planetary health (e.g. control of non-native species), but divergent regulatory standards can contribute to delays at borders and sometimes act as protectionist non-tariff barriers (Aginam 2008). International harmonization of these standards would help create larger markets for alternative proteins by making it easier for producers to attain the necessary economies of scale. Policymakers could also consider including novel ASF alternatives in agreements that liberalize trade in environmental goods. Efforts to better align trade policy with environmental objectives could contribute to more sustainable food systems, and in practice would likely favour novel alternatives compared to conventional ASFs.

Development of international food safety standards: Governments can engage with multilateral institutions such as Codex Alimentarius (run jointly by the FAO and WHO), which proposes international standards for food safety and regulation, to support recommendations that ensure timely, efficient and safe approval processes for novel ASF alternatives so that they can compete fairly with conventional equivalents. In April 2023, the FAO and WHO jointly released a report on cultivated meat safety, with the stated goal of equipping authorities (particularly in low- and middle-income countries) with up-to-date scientific knowledge for consideration in potentially important regulatory actions (FAO and WHO 2023).

5. Conclusion

Current rates of ASF production and consumption are a significant contributor to global GHG emissions, land and water use, deforestation, pollution and biodiversity loss. Moreover, the livestock industry is associated with a range of health risks, including the emergence of zoonoses, growing antimicrobial resistance through overuse of antibiotics and increases in several types of NCDs associated with high levels of red and processed meat consumption. ASF production also raises a host of animal welfare issues.

It is important to recognize that the distribution of ASF consumption and production is highly uneven, and the impacts vary depending on a range of factors, including animal product type and rearing methods used, and subsidies to livestock industries and markets. Nevertheless, it is evident that, overall, current global production and consumption risks undermining several critical environmental and health goals. To counter these risks, significant changes are needed in the way the world produces and consumes ASF, and calls are growing for regions with high levels of ASF production and/or consumption to reduce them.

A range of options are available to help mitigate the impacts of ASF on our environment and health, as this report briefly discusses. Examples of these options include regenerative livestock farms, feed additives to reduce emissions, reducing subsidies for the livestock sector, and internalizing externalities into the price of meat. Thus far, such interventions have been limited

and are not achieving the desired impacts at the scale and speed necessary. With meat consumption projected to grow by about 50 per cent or more by 2050 as incomes and population levels continue to rise, such a scenario puts the achievement of several Sustainable Development Goals at risk.

In recent years, there has been growing attention to the role that novel ASF alternatives, including novel plant-based, cultivated and biomass and precision fermentation-derived products, may be able to play in supporting a shift away from ASF consumption by closely mimicking or even replicating the sensory experience of ASF. The private sector has invested significant sums in the sector over the past decade. Governments are also investing in research and development related to novel alternatives to ASF, includ-ing for reasons of environmental protection, food security and economic competitiveness.

As discussed in this report, novel meat and dairy alternatives may be associated with a range of benefits, compared to their conventional counterparts (see Table 5.1):

 Environmental – Compared to their conventional counterparts, novel meat and dairy alternatives may in many cases be associated with reduced GHG emissions; reduced land use; and reduced pollution from animal waste and agricultural chemicals. This may help contribute to several international environmental goals such as those under the Paris Agreement and Kunming-Montreal Global Biodiversty Framework.

 Human health – Novel meat and dairy alternatives can generally be expected to reduce risks associated with the emergence of antimicrobial resistance and zoonoses.

 Animal welfare – Replacing conventional meat and dairy with novel alternatives would significantly reduce the number of animals being reared and killed for food, often in conditions that raise concerns about their welfare. It should be noted that different novel ASF alternatives may have different environmental, health and social implications. These benefits could potentially be significant, but there are also challenges and limitations associated with these novel products, depending on the technology.

For example, the production of some novel ASF products, such as cultivated meat, is highly energy intensive. Realizing the full climate potential of novel ASF alternatives is thus contingent on using low-carbon energy, which in turn depends on governments' ability to transition to a low-carbon economy consistent with their commitments under the Paris Agreement.

In addition, there is currently limited evidence on the nutritional health implications of substituting conventional ASF with novel alternatives. Impacts may depend on the degree of processing and the levels of salt and saturated fats in novel products.

There are a number of practical questions that will impact the potential of novel alternatives to ASF to contribute to a more sustainable, healthy and humane food system. Even if these novel products offer clear and significant benefits, to what degree will they replace or complement conventional products, since the benefits only accrue through substitution? Can they be produced on a large enough scale at a price that is competitive with conventional meat and dairy, especially if the cost of conventional products

remains artificially low because of government subsidies and the failure to internalize their environmental and health costs? Will consumers across vastly different socioeconomic and cultural realities accept the novel products that reach the market?

Important questions surrounding the social implications of novel alternatives have received limited attention and remain to be clarified. For instance, how would a shift from conventional meat and dairy consumption and production towards novel alternatives affect stakeholders, especially in the Global South, small-scale farmers, Indigenous Peoples, and taking into account gender differences? Improving global food security is a policy priority for many countries, yet the ways that novel alternatives may impact food security in different regions remains understudied. And while the rise of alternatives could both displace jobs and create new ones, it may also support the further consolidation of food production in the hands of big corporations, which may create efficiencies but also come at a cost, such as a lack of diversity of products and of local ownership of food systems. Can societies manage such transitions fairly while maximizing benefits and minimizing downsides?

Policies, regulations, incentives and awareness-raising will be needed to achieve positive and just outcomes, while minimizing risks and disruptions. While some governments are actively promoting novel alternatives, including through research and development, others are moving to protect their livestock industries and/or other animal product producers by restricting or banning some novel alternatives. Given the number of important evidence gaps, transparent decision-making and increased government support for research is needed to improve understanding of the environmental, health and social implications of these novel alternatives, and which policies and investments are most effective for governing these technologies and realizing their potential.

Building a more sustainable, healthy and just food system is a priority for the achievement of many of the Sustainable Development Goals. With the right policies in place, novel ASF alternatives may be able to contribute positively to this critical endeavour.

Table 5.1

Overview of the environmental, health, social and animal welfare implications of novel plant-based, cultivated meat and fermentation-derived alternatives, compared to conventional meat and dairy products

This table presents the relative environmental, health, social, and animal welfare implications of selected novel plant-based, cultivated and fermentation-derived meat and dairy alternatives, compared to conventional industrial meat and dairy products, drawing upon the scientific evidence and analysis presented in this report.

- Higher impact
- Equivalent or very similar impact
- Lower impact
- More research needed
- O Not addressed in this report

Health	Novel plant-based products	Cultivated products	Fermentation-derived products
Risk of emerging zoonoses			
Risk of antimicrobial resistance			
Nutritional quality and dietary health outcomes			

Social	Novel plant-based products	Cultivated products	Fermentation-derived products
Food security			
Job gains/losses ⁴			

Changes in inequalities, e.g. global South and North, rural-urban, gender

Source: Authors' summary assessment

¹ The environmental assessments considered for this report are primarily based on life cycle assessment data of the typical, or "conventional", meat and dairy production systems of specific regions, including intensive industrialised farming systems, extensive free-range production, organic systems and various combinations of these. Most utilize a combination of data sources, including both industrial and free-range systems typical of Europe and North America, but do not generally assess nondominant systems such as smallholder farming systems.
² Novel plant-based chicken and pork products have received only limited attention in terms of their water use, and more research would be beneficial.
³ Although most analyses focus on land savings potential, land use is the largest driver of biodiversity loss, and as a result biodiversity gains are therefore likely (Balmford *et al.* 2019).
⁴ It is likely that jobs will be lost in the livestock industry sector, while will be gained in the novel alternatives sectors. More information can be found in Section 3.5.
⁵ If animal-free growth medium is used.

List of references

Adesogan, A.T., Havelaar, A.H., McKune, S.L., Eilittä, M. and Dahl, G.E. (2020). Animal source foods: Sustainability problem or malnutrition and sustainability solution? Perspective matters. *Global Food Security* 25, 100325. https://doi.org/10.1016/j.gfs.2019.100325.

- Afshin, A., Sur, P.J., Fay, K.A., Cornaby, L., Ferrara, G., Salangma, J.S. *et al.* (2019). Health effects of dietary risks in 195 countries, 1990–2017: A systematic analysis for the Global Burden of Disease Study 2017. *The Lancet* 393(10184), 1958–1972. https://doi.org/10.1016/s0140-6736(19)30041-8.
- Aginam, O. (2008). Trade, health or politics? Protectionism, risk assessment and the globalization of food safety. *Food and Drug Law Journal* 63(3), 665–672. http://www.jstor.org/stable/45119339.
- Agritech Future (2022). Vertical Future successfully completes Innovate UK funded research aimed at cutting agrichemical use, 22 November. https:// www.agritechfuture.com/vertical-farming/vertical-future-successfully-completes-innovate-uk-funded-research-aimed-at-cutting-agrichemical-use/. Accessed 3 July 2023.
- Ahmad, M.I., Farooq, S., Alhamoud, Y., Li, C. and Zhang, H. (2022). A review on mycoprotein: History, nutritional composition, production methods and health benefits. *Trends in Food Science and Technology* 121, 14–29. http:// dx.doi.org/10.1016/j.tifs.2022.01.027.

Ajena, F., Alvares, I., Becheva, S., Benning, R., Bernal Rubio, M., Birke, P. et al. (2021) Meat Atlas. Facts and figures about the animals we eat. Heinrich Böll Stiftung, Friend of the Earth Europe. https://eu.boell.org/sites/default/files/2021-09/MeatAtlas2021_final_web.pdf

Alemayehu, G., Berhe, T., Gelan, E., Mokria, M., Jaldessa, J., Molu, J. *et al.* (2022). Animal welfare knowledge, attitudes and practices among livestock
 holders in Ethiopia. *Frontiers in Veterinary Science* 9, 1006505. https://doi.
 org/10.3389/fvets.2022.1006505.

ent Alonso, S., Dominguez-Salas, P. and Grace, D. (2019). The role of livestock products for nutrition in the first 1,000 days of life. *Animal Frontiers* 9(4), 24–31. https://doi.org/10.1093/af/vfz033.

UK Anderson, C.A.M. and Bradley, R. (2020). The potential of novel plant protein foods to improve dietary patterns and markers of cardiovascular health.
The American Journal of Clinical Nutrition 112(5), 1151–1152. https://doi.org/10.1093/ajcn/nqaa253.

Anderson, T. (2019). *Principles for a Just Transition in Agriculture*. Johannesburg: Action Aid. https://actionaid.org/sites/default/files/publications/Principles%20for%20just%20transition%20in%20agriculture.pdf.

Arndt, C., Hristov, A.N., Price, W.J. and Yu, Z. (2022). Full adoption of the most effective strategies to mitigate methane emissions by ruminants can help meet the 1.5°C target by 2030 but not 2050. *Proceedings of the National Academy of Sciences* 119(20), e2111294119. https://doi.org/10.1073/pnas.2111294119.

Arwanto, V., Buschle-Diller, G., Mukti, Y.P., Dewi, A.D.R., Mumpuni, C., Purwanto M.G.M. *et al.* (2022). The state of plant-based food development and its prospects in the Indonesia market. *Heliyon* 8(10), e11062. https:// doi.org/10.1016/j.heliyon.2022.e11062.

Athnis, J. (2019). US Factory Farming Estimates, 11 April. https://www.sentienceinstitute.org/us-factory-farming-estimates.

Aubrey, A. (2023). 'No kill' meat, grown from animal cells is now approved for sale in the U.S. *NPR*, 21 June. https://www.npr.org/sections/ health-shots/2023/06/21/1183484892/no-kill-meat-grown-from-animalcells-is-now-approved-for-sale-in-the-u-s. Accessed 4 July 2023.

Australian Government (2016). Australia New Zealand food standards code standard 1.5.1 – Novel foods. *Federal Register of Legislation*. https://www. legislation.gov.au/Details/F2017C00324. Accessed 18 July 2023.

Australian Government (2022). Australia's A\$56 billion protein sector

Behm, K., Nappa, M., Aro, N., Welman, A., Ledgard, S., Suomalainen, M. *et al.* (2022). Comparison of carbon footprint and water scarcity footprint of milk protein produced by cellular agriculture and the dairy industry. *International Journal of Life Cycle Assessment* 27, 1017–1034. https://doi.org/10.1007/ s11367-022-02087-0.

Behrens, P., Kiefte-de Jong, J.C., Bosker, T., Rodrigues, J.F.D., de Koning, A. and Tukker, A. (2017). Evaluating the environmental impacts of dietary recommendations. *Proceedings of the National Academy of Sciences* 114(51), 13412–13417. https://doi.org/10.1073/pnas.1711889114.

Beketov, M.A., Kefford, B.J., Schäfer, R.B. and Liess, M. (2013). Pesticides reduce regional biodiversity of stream invertebrates. *Proceedings of the National Academy of Sciences* 110(27), 11039–11043. https://doi. org/10.1073/pnas.1305618110.

BENEO Global Plant-Based Survey (2022). Global consumer survey:
Flexitarians are the driving force behind plant-based foods. Accesswire,
5 April. https://www.accesswire.com/696128/Global-Consumer-SurveyFlexitarians-are-the-Driving-Force-Behind-Plant-Based-Foods.

Benincasa, N.C., Sakamoto, K.S., Silva, I.J.O.D. and Lobos, C.M.V. (2020). Animal welfare: impacts of pre-slaughter operations on the current poultry industry. *Journal of Animal Behaviour and Biometeorology*, *8*, 104-110. http://dx.doi.org/10.31893/jabb.20014.

holds the key to the future of global food security. *Australian Trade and Investment Commission,* 22 July 2022. https://www.austrade.gov.au/en/ news-and-analysis/analysis/au13bn-investment-opportunity-in-australian-protein.html. Accessed 20 September 2023.

Baldos, U.L.C. and Hertel, T.W. (2015). The role of international trade in managing food security risks from climate change. *Food Security* 7, 275–290. https://doi.org/10.1007/s12571-015-0435-z.

Balmford, B., Green, R.E., Onial, M., Phalan, B. and Balmford, A. (2019). How imperfect can land sparing be before land sharing is more favourable for wild species? *Journal of Applied Ecology* 56(1), 73–84. https://doi.org/10.1111/1365-2664.13282.

Banks, M., Johnson, R., Giver, L., Bryant, G. and Guo, M. (2022). Industrial production of microbial protein products. *Current Opinion in Biotechnology* 75, 102707. https://doi.org/10.1016/j.copbio.2022.102707.

Barbarossa, V., Bosmans, J., Wanders, N., King, H., Bierkens, M.F.P., Huijbregts, M.A.J. *et al.* (2021). Threats of global warming to the world's freshwater fishes. *Nature Communications* 12, 1701. https://doi. org/10.1038/s41467-021-21655-w.

Barbieri, P., MacDonald, G.K., Bernard de Raymond, A. and Nesme, T. (2022). Food system resilience to phosphorus shortages on a telecoupled planet. *Nature Sustainability* 5, 114-122. https://doi.org/10.1038/

Bernstein, J. and Dutkiewicz, J. (2021). A public health ethics case for mitigating zoonotic disease risk in food production. *Food Ethics* 6(2), 9. https:// doi.org/10.1007/s41055-021-00089-6.

Bianchi, F., Dorsel, C., Garnett, E., Aveyard, P. and Jebb, S.A. (2018). Interventions targeting conscious determinants of human behaviour to reduce the demand for meat: A systematic review with qualitative comparative analysis. *International Journal of Behavioural Nutrition and Physical Activity* 15, 1-25. https://doi.org/10.1186/s12966-018-0729-6.

Blattner, C. E. (2019) The Recognition of Animal Sentience by the Law. Journal of Animal Ethics,9 (2), 121–36. https://doi.org/10.5406/ janimalethics.9.2.0121.

Boereboom, A., Mongondry, P., de Aguiar, L.K., Urbano, B., Jiang, Z., de Koning, W. *et al.* (2022). Identifying consumer groups and their characteristics based on their willingness to engage with cultured meat: A comparison of four European countries. *Foods* 11(2), 197. http://dx.doi.org/10.3390/ foods11020197.

Boffey, D. (2020). EU spending tens of millions of euros a year to promote meat eating. *The Guardian*, 14 February. https://www.theguardian.com/ environment/2020/feb/14/eu-spending-tens-of-millions-of-euros-a-year-topromote-meat-eating. Accessed 5 July 2023.

s41893-021-00816-1.

Barrett, C.B., Benton, T., Herrero, M., Nelson, R.J., Bageant, E., Buckler, E. et al. (2022). Envisioning four design objectives for 2045–2070. In Socio-Technical Innovation Bundles for Agri-Food Systems Transformation. Cham: Palgrave Macmillan. 55-63. https://doi.org/10.1007/978-3-030-88802-2_4.

Bauer, J.M. and Reisch, L.A. (2019). Behavioural insights and (un)healthy dietary choices: A review of current evidence. *Journal of Consumer Policy* 42, 3–45. https://doi.org/10.1007/s10603-018-9387-y.

Bechtel, W. (2018). Stabilizing Jersey calf prices for beef markets with crossbreeding. *The Packer*, 28 December. https://www.thepacker.com/markets/ know-your-market/stabilizing-jersey-calf-prices-beef-markets-crossbreeding. Accessed 25 November 2022.

Bohrer, B.M. (2019). An investigation of the formulation and nutritional composition of modern meat analogue products. *Food Science and Human Wellness* 8(4), 320–329. https://doi.org/10.1016/j.fshw.2019.11.006.

Bomkamp, C., Skaalure, S., Fernando, G., Ben-Arye, T., Swartz, E. and Specht, E. (2021). Scaffolding biomaterials for 3D cultivated meat: Prospects and challenges. *Advanced Science* 9(3), 2102908. https://doi.org/10.1002/ advs.202102908.

Bonnet, C., Bouamra-Mechemache, Z., Requillart, V. and Treich, N. (2020). Viewpoint: Regulating meat consumption to improve health, the environment and animal welfare. *Food Policy* 97, 101847. https://doi. org/10.1016/j.foodpol.2020.101847. Bosire, C.K., Ogutu, J.O., Said, M.Y., Krol, M.S., de Leeuw, J. and Hoekstra, A.Y. (2015). Trends and spatial variation in water and land footprints of meat and milk production systems in Kenya. *Agriculture, Ecosystems and Environment* 205, 36–47. https://doi.org/10.1016/j.agee.2015.02.015.

Bouvard, V., Loomis, D., Guyton, K. Z., El Ghissassi, F., Benbrahim-Tallaa, L. et al. Carcinogenicity of consumption of red and processed meat. The Lancet Oncology, 16, 1599-1600. https://doi.org/10.1016/S1470-2045(15)00444-1.

Bouwman, L., Goldewijk, K.K., Van Der Hoek, K.W., Beusen, A.H., Van Vuuren, D.P., Willems, J. *et al.* (2013). Exploring global changes in nitrogen and phosphorus cycles in agriculture induced by livestock production over the 1900–2050 period. *Proceedings of the National Academy of Sciences* 110(52), 20882-20887. https://doi.org/10.1073/pnas.1012878108.

Branca, F., Demaio, A., Udomkesmalee, E., Baker, P., Aguayo, V.M., Barquera, S. *et al.* (2020). A new nutrition manifesto for a new nutrition reality. *The Lancet* 395(10217), 8-10. https://doi.org/10.1016/s0140-6736(19)32690-x.

Brazil, Agencia Nacional de Vigilancia Sanitaria (2020). Food – Market authorizations, 29 October. https://www.gov.br/anvisa/pt-br/english/regulation-of-products/food. Accessed 18 July 2023.

Brazil, Agencia Nacional de Vigilancia Sanitaria (2022).

CE Delft and the Good Food Institute (2021). *The Costs and Environmental Impacts of Cultivated Meat*. https://gfi.org/resource/ cultivated-meat-lca-tea-report-analysis/.

Cena, H. and Calder, P.C. (2020). Defining a healthy diet: Evidence for the role of contemporary dietary patterns in health and disease. *Nutrients* 12(2), 334. https://doi.org/10.3390/nu12020334.

Center for International Forestry Research (2021). *Impacts of Taking, Trade and Consumption of Terrestrial Migratory Species for Wild Meat.* Bonn: CMS Secretariat. https://www.cms.int/en/publication/impacts-taking-trade-and-consumption-terrestrial-migratory-species-wild-meat-report. Accessed 5 June 2023.

Cervera, M. (2023). Cultivated meat in the UK? Aleph Farms files Britain's first cell-based steak application. *Food Ingredients First*, 4 August. https://www.foodingredientsfirst.com/news/cultivated-meat-in-the-uk-aleph-farms-files-britains-first-cell-based-steak-application.html. Accessed 14 August 2023.

Ceryes, C.A. and Heaney, C.D. (2019). 'Ag-gag' laws: Evolution, resurgence and public health implications. *New Solutions: A Journal of Environmental and Occupational Health Policy* 28(4), 664–682. https://doi.

Relatório de Análise de Impacto Regulatório sobre a modernização do marco regulatório, fluxos e procedimentos para novos alimentos e novos ingredientes. http://antigo.anvisa.gov.br/documents/10181/6582266/Relat%C3%B3rio+de+AIR+Novos+Alimentos.pdf/ fdc2f762-15a9-4811-a9ea-cc92add0ddce.

Browning, H. and Birch, J. (2022). Animal sentience. *Philosophy Compass* 17(5), e12822. https://doi.org/10.1111/phc3.12822.

Bryant, C. (2020). Culture, meat and cultured meat. *Journal of Animal Science* 98(8), skaa172. https://doi.org/10.1093/jas/skaa172.

Bryant, C. and Barnett, J. (2018). Consumer acceptance of cultured meat: A systematic review. *Meat Science* 143, 8-17. https://doi.org/10.1016/j. meatsci.2018.04.008.

Bryant, C., Szejda, K., Parekh, N., Deshpande, V. and Tse, B. (2019). A survey of consumer perceptions of plant-based and clean meat in the USA, India and China. *Frontiers in Sustainable Food Systems* 3(11). https://doi. org/10.3389/fsufs.2019.00011.

Canada, Government of Canada (2023). C.R.C, c.870. Justice Laws Website. https://laws-lois.justice.gc.ca/eng/regulations/C.R.C.,_c._870/section-B.28.001.html.

Canada, Agriculture and Agri-Food Canada (2022). Government of Canada invests over \$1.4 million in plant-based food production in British Columbia. Government of Canada, 7 July. https://www.canada.ca/en/ agriculture-agri-food/news/2022/07/government-of-canada-invests-over-14-million-in-plant-based-food-production-in-british-columbia.html.

org/10.1177%2F1048291118808788.

Chandel, B.S., Lal, P. and Kumari, B. (2019). Livestock production systems, subsidies and its implications: An investigation through review of literature. *Indian Journal of Dairy Science*, 72(2),121-128. https://doi.org/10.33785/ IJDS.2019.v72i02.001.

Change Foods (2022). Change Foods awarded \$1 million grant for development of sustainable feedstock for animal-free dairy. *Protein Report*, 5 May. https://www.proteinreport.org/newswire/change-foods-awarded-1-milliongrant-development-sustainable-feedstock-animal-free-dairy. Accessed 3 July 2022.

Chen, L., Guttieres, D., Koenigsberg, A., Barone, P.W., Sinskey, A.J. and Springs, S.L. (2022). Large-scale cultured meat production: Trends, challenges and promising biomanufacturing technologies. *Biomaterials* 280, 121274. https://doi.org/10.1016/j.biomaterials.2021.121274.

Chichaibelu, B.B., Bekchanov, M., von Braun, J. and Torero, M. (2021). The global cost of reaching a world without hunger: Investment costs and policy action opportunities. *Food Policy* 104, 102151. https://doi. org/10.1016/j.foodpol.2021.102151.

Chigateri, S. (2010). Negotiating the 'sacred' cow: Cow slaughter and the regulation of difference in India. In *Democracy, Religious Pluralism and the Liberal Dilemma of Accommodation*. Studies in Global Justice. Mookherjee, M. (ed.). Dordrecht: Springer. 137-159. https://doi.

Casey, J.A., Kim, B.F., Larsen, J., Price, L.B. and Nachman, K.E. (2015). Industrial food animal production and community health. *Current Environmental Health Reports* 2(3), 259–271. https://doi.org/10.1007/ s40572-015-0061-0.

Castañeda, D., Newhouse, D., Nguyen, M.C., Uematsu, H., Azevedo, J.P. and the World Bank Data for Goals Group (2018). A new profile of the global poor. *World Development* 101, 250–267. https://doi.org/10.1016/j. worlddev.2017.08.002.

org/10.1007/978-90-481-9017-1_8.

Chriki, S. and Hocquette, J.F. (2020). The myth of cultured meat: A review. *Frontiers in Nutrition* 7(7). https://doi.org/10.3389/fnut.2020.00007.

Clark, M.A., Domingo, N.G., Colgan, K., Thakrar, K., Tilman, D., Lynch, J. *et al.* (2020). Global food system emissions could preclude achieving the 1.5° and 2°C climate change targets. *Science* 370(6517), 705–708. https://doi. org/10.1126/science.aba7357.

Clark, M.A., Springmann, M., Hill, J. and Tilman, D. (2019). Multiple health and environmental impacts of foods. *Proceedings of the National Academy of Sciences* 116(46), 23357–62. https://doi.org/10.1073/ pnas.1906908116. Climate and Clean Air Coalition (2023). Enteric fermentation. The United Nations Environment Programme, Food and Agriculture Organization of the United Nations, Global Research Alliance on Agricultural Greenhouse Gases, New Zealand. https://www.ccacoalition.org/projects/ enteric-fermentation.

ClimateWorks Foundation and Foreign, Commonwealth and Development Office, UK Government (2021). *Global Innovation Needs Assessments: Protein Diversity*. https://www.climateworks.org/wp-content/ uploads/2021/11/GINAs-Protein-Diversity.pdf.

Coca, N. (2023). Plant burgers are way better for the planet than beef, but these 2 ingredients threaten tropical ecosystems. *Vox*, 7 June. https:// www.vox.com/future-perfect/23746037/plant-meat-beyond-impossible-sustainabilility-coconut-cacao. Accessed 10 August 2023.

Cordain, L., Eaton, S.B., Sebastian, A., Mann, N., Lindeberg, S., Watkins, B.A. et al. (2005). Origins and evolution of the Western diet: Health implications for the 21st century. *The American Journal of Clinical Nutrition* 81(2), 341– 354. https://doi.org/10.1093/ajcn.81.2.341.

Crimarco, A., Springfield, S., Petlura, C., Streaty, T., Cunanan, K., Lee, J. et al. (2020). A randomized crossover trial on the effect of plant-based compared with animal-based meat on trimethylamine-N-oxide and cardiovascular disease risk factors in generally healthy adults: Study with appetizing plantfood-meat eating alternative trial (SWAP-MEAT). *The American Journal of Clinical Nutrition* 112(5), 1188–1199. https://doi. org/10.1093/ajcn/nqaa203. Dawkins, M.S. (2017). Animal welfare with and without consciousness. *Journal of Zoology* 301(1), 1-10. https://doi.org/10.1111/jzo.12434.

de Streel, A. and Larouche, P. (2015). Disruptive innovation and competition policy enforcement. OECD Working Paper DAF/COMP/GF(2015)7. https://doi.org/10.2139/SSRN.2678890.

de Vries, M. and de Boer, I.J.M. (2010). Comparing environmental impacts for livestock products: A review of life cycle assessments. *Livestock Science* 128 (1-3), 1–11. https://doi.org/10.1016/j.livsci.2009.11.007.

Dehghan, M., Mente, A., Rangarajan, S., Sheridan, P., Mohan, V., Iqbal, R. et al. (2018). Association of dairy intake with cardiovascular disease and mortality in 21 countries from five continents (PURE): A prospective cohort study. The Lancet, 392(10161), 2288-2297. https://doi.org/10.1016/ S0140-6736(18)31812-9.

Demartini, E., Vecchiato, D., Finos, L., Mattavelli, S. and Gaviglio, A. (2022). Would you buy vegan meatballs? The policy issues around vegan and meat-sounding labelling of plant-based meat alternatives. *Food Policy* 111, 102310. https://doi.org/10.1016/j.foodpol.2022.102310.

Denmark, Ministry of Food, Agriculture and Fisheries (2021). Denmark is to be a world leader in plant-based foods and green proteins. Fact Sheet 4.

Crimston, D., Bain, P.G., Hornsey, M.J. and Bastian, B. (2016). Moral expansiveness: Examining variability in the extension of the moral world. *Journal of Personality and Social Psychology* 111(4), 636–653. https://doi. org/10.1037/pspp000086.

Crippa, M., Solazzo, E., Guizzardi, D., Monforti-Ferrrario, F., Tubiello, F. N. and Leip, A. (2021). Food systems are responsible for a third of global anthropogenic GHG emissions. *Nature Food* 2(3), 198–209. https://doi. org/10.1038/s43016-021-00225-9.

Cusworth, G., Lorimer, J., Brice, J. and Garnett, T. (2022). Green rebranding: Regenerative agriculture, future-pasts and the naturalisation of livestock. *Transactions of the Institute of British Geographers* 47(4), 1009-1027. https://doi.org/10.1111/tran.12555.

da Costa, G.G., da Conceição Nepomuceno, G., da Silva Pereira, A. and Simões, B.F.T. (2022). Worldwide dietary patterns and their association with socioeconomic data: An ecological exploratory study. *Globalization and Health* 18(1), 31. https://doi.org/10.1186/s12992-022-00820-w.

da Silva, V.P., van der Werf, H.M., Spies, A. and Soares, S.R. (2010). Variability

https://en.fvm.dk/fileadmin/user_upload/FVM.dk/Dokumenter/Landbrug/ Fact_Sheet_4.pdf.

Derbes, E. (2021). Cultivated meat is gaining momentum–and pathways for regulatory approval. *The Good Food Institute*, 3 June. https://gfi.org/blog/cultivated-meat-regulation-2021/.

DeSoucey, M. (2010). Gastronationalism: Food traditions and authenticity politics in the European Union. *American Sociological Review* 75(3), 432–455. https://doi.org/10.1177/0003122410372226.

Dolmans, M. (2020). Sustainable competition policy. *Competition Law and Policy Debate CLPD* 5(4) and 6(1). https://doi.org/10.2139/SSRN.3608023.

Douglas, P., Robertson, S., Gay, R., Hansell, A.L. and Gant, T.W. (2018). A systematic review of the public health risks of bioaerosols from intensive farming. *International Journal of Hygiene and Environmental Health* 221(2), 134–173. https://doi.org/10.1016/j.ijheh.2017.10.019.

Drouin-Chartier, J.P., Chen, S., Li, Y., Schwab, A.L., Stampfer, M.J., Sacks, F.M. *et al.* (2020). Egg consumption and risk of cardiovascular disease: Three large prospective US cohort studies, systematic review and updated meta-analysis. *BMJ* 368. https://doi.org/10.1136/bmj.m513.

Dudley, N. and Alexander, S. (2017). Agriculture and biodiversity: A review. *Biodiversity* 18(2-3), 45-49. https://doi.org/10.1080/14888386.2017.1351892.

in environmental impacts of Brazilian soybean according to crop production and transport scenarios. *Journal of Environmental Management* 91(9), 1831-1839. https://doi.org/10.1016/j.jenvman.2010.04.001.

Danley, S. (2020). Taste and price are table stakes for trendy food items. *Food Business News*, 28 October. https://www.foodbusinessnews.net/ articles/17161-taste-and-price-are-table-stakes-for-trendy-food-items. Accessed 5 July 2023.

Davis, B., Lipper, L. and Winters, P. (2022). Do not transform food systems on the backs of the rural poor. *Food Security* 14(3), 729–740. https://doi. org/10.1007/s12571-021-01214-3. Dullaghan, N. (2021). Cultured meat predictions were overly optimistic. *Rethink Priorities*, 15 September. https://rethinkpriorities.org/publications/ cultured-meat-predictions-were-overly-optimistic.

Duncan, J., DeClerck, F., Baldi, A., Treyer, S., Aschemann-Witzel, J., Cuhls, K. *et al.* (2022). Democratic directionality for transformative food systems research. *Nature Food* 3, 183-186. https://doi.org/10.1038/ s43016-022-00479-x.

Dutkiewicz, J. (2021). How do you convince people to eat less meat? *The New Republic*, 28 July. https://newrepublic.com/article/163079/convincepeople-eat-less-meat. Accessed 11 August 2023. Dutkiewicz, J. and Abrell, E. (2021). Sanctuary to table dining: Cellular agriculture and the ethics of cell donor animals. *Politics and Animals* 7. https:// journals.lub.lu.se/pa/article/view/22252/20774.

Edjabou, L.D. and Smed, S. (2013). The effect of using consumption taxes on foods to promote climate friendly diets – The case of Denmark. *Food Policy* 39, 84-96. https://doi.org/10.1016/j.foodpol.2012.12.004.

Eisen, M.B. and Brown, P.O. (2022). Rapid global phaseout of animal agriculture has the potential to stabilize greenhouse gas levels for 30 years and offset 68 per cent of CO2 emissions this century. *PLOS Climate* 1(2), e0000010. https://doi.org/10.1371/journal.pclm.0000010.

English, L.K., Ard, J.D., Bailey, R.L., Bates, M., Bazzano, L.A., Boushey, C.J., Brown, C. *et al.* (2021). Evaluation of dietary patterns and all-cause mortality: A systematic review. *JAMA Network Open* 4(8), e2122277. https:/doi. org/10.1001%2Fjamanetworkopen.2021.22277.

Ettinger, J. (2023a). Israel's PMO committed to 'solidifying' the country's leadership in alternative protein'. *Green Queen*, 3 April. https://www.greenqueen.com.hk/israels-pmo-committed-to-alternative-protein/#:~:tex-t=The%20PMO%20says%20Israel%20is,concerns%20and%20bolster%20 the%20economy. Accessed 5 July 2023.

Fisher, A.D., Colditz, I.G., Lee, C. and Ferguson, D.M. (2009). The influence of land transport on animal welfare in extensive farming systems. *Journal of Veterinary Behavior* 4(4), 157-162. https://doi.org/10.1016/j. jveb.2009.03.002.

Flint, M., Bowles, S., Lynn, A. and Paxman, J.R. (2023). Novel plant-based meat alternatives: Future opportunities and health considerations. *Proceedings of the Nutrition Society* 82(3), 1-16. https://doi.org/10.1017/ s0029665123000034.

Food and Agriculture Organization of the United Nations (2009). Food Insecurity in the World. https://www.fao.org/3/i0876e/i0876e00.htm.

Food and Agriculture Organization of the United Nations (2014). Food Wastage Footprint: Full Cost-accounting. Rome.

Food and Agriculture Organization of the United Nations (2021). *Emissions Due to Agriculture: Global, Regional and Country trends* 2000–2018. FAOSTAT Analytical Brief Series No 18. Rome. https://www.fao.org/3/ cb3808en/cb3808en.pdf.

Food and Agriculture Organization of the United Nations (2022). *Thinking About the Future of Food Safety – A Foresight Report*. Rome. https://doi. org/10.4060/cb8667en.

Ettinger, J. (2023b). Mycorena and Revo Foods hook €1.5 million for 3D mycoprotein seafood. *Green Queen*, 26 June. https://www.greenqueen. com.hk/mycorena-revo-foods-e1-5-million-3d-mycoprotein-seafood/. Accessed 5 June 2023.

Eurofound (2019). The future of manufacturing in Europe. Publication Office of the European Union. https://www.eurofound.europa.eu/en/future-manufacturing-europe-fome#:~:text=Final%20report%20published%20in%20 April,can%20be%20promoted%2C%20even%20regionally.

European Food Safety Authority Panel on Dietetic Products and Allergies (2016). Guidance on the preparation and presentation of an application for authorization of a novel food in the context of Regulation (EU) 2015/2283. *European Food Safety Authority Journal* 14(11), 4594. https:// doi.org/10.2903/j.efsa.2016.4594.

European Union, Official Journal of the European Union (2015). Regulation (EU) 2015/2283 of the European Parliament and of the council of 25 November 2015 on novel foods. https://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:32015R2283.

Farvid, M.S., Sidahmed, E., Spence, N.D., Mante Angua, K., Rosner, B.A. and Barnett, J.B. (2021). Consumption of red meat and processed meat and cancer incidence: a systematic review and meta-analysis of prospective studies. European Journal of Epidemiology, 36, 937-951. https://doi. org/10.1007/s10654-021-00741-9

Food and Agriculture Organization of the United Nations (2023a). Food Balances (-2013, old methodology and population) Database. https://www. fao.org/faostat/en/#data/FBSH. Accessed 12 January 2023.

Food and Agriculture Organization of the United Nations (2023b). Food Balances (2010-) Database. https://www.fao.org/faostat/en/#data/FBS. Accessed 12 January 2023.

Food and Agriculture Organization of the United Nations (2023c). FAOSTAT Database. http://www.fao.org/faostat/en/#data. Accessed 14 February 2023.

Food and Agriculture Organization of the United Nations (2023d). FAO GLEAM v3.0 dashboard. Emissions. https://foodandagricultureorganiza-tion.shinyapps.io/GLEAMV3_Public/ Accessed 7 November 2023.

Food and Agriculture Organization of the United Nations (2023e). Contribution of terrestrial animal source food to healthy diets for improved nutrition and health outcomes – An evidence and policy overview on the state of knowledge and gaps. Rome, FAO. https://doi.org/10.4060/ cc3912en

Food and Agriculture Organization of the United Nations and World Health Organization (2023). *Food Safety Aspects of Cell-based Food*. Rome. https://doi.org/10.4060/cc4855en.

Fedor, P. and Zvaríková, M. (2019). Biodiversity indices. *Encyclopaedia of Ecology* 1, 337-346. https://doi.org/10.1016/B978-0-12-409548-9.10558-5.

Firth, N. (2020). Cultured meat has been approved for consumers for the first time. *MIT Technology Review*, 1 December. https://www.technologyreview. com/2020/12/01/1012789/cultured-cultivated-meat-just-singapore-approved-food-climate/. Accessed 21 August 2023.

Fite, M.B., Tura, A.K., Yadeta, T.A., Oljira, L. and Roba, K.T. (2022). Consumption of animal source food and associated factors among pregnant women in eastern Ethiopia: A community-based study. *PLoS ONE* 17(6), e0270250. https://doi.org/10.1371%2Fjournal.pone.0270250. Food and Agriculture Organization of the United Nations, International Fund for Agricultural Development, United Nations International Children's Emergency Fund, World Food Programme and World Health Organization (2021). *The State of Food Security and Nutrition in the World 2021*. Rome: Food and Agriculture Organization of the United Nations. https://doi. org/10.4060/cb4474en.

Food and Agriculture Organization of the United Nations, International Fund for Agricultural Development, United Nations International Children's Emergency Fund, World Food Programme and World Health Organization (2022). *The State of Food Security and Nutrition in the World 2022*. Rome. https://doi.org/10.4060/cc0639en. Food Standards Australia New Zealand (2020). Approval report – Application A1186: Soy leghemoglobin in meat analogue products, 15 December. https://www.foodstandards.gov.au/code/applications/ Documents/A1186%20Approval%20Report.pdf.

Food Standards Australia New Zealand (2022). Health Canada and FSANZ complete first shared GM food safety assessment, 17 June. https://www.foodstandards.gov.au/science/international/Pages/gm-food-safety.aspx.

Food Standards Australia New Zealand (2023). Applications. https://www. foodstandards.gov.au/code/applications/Pages/default.aspx. Accessed 3 October 2023.

Fraeye, I., Kratka, M., Vandenburgh, H. and Thorrez, L. (2020). Sensorial and nutritional aspects of cultured meat in comparison to traditional meat: Much to be inferred. *Frontiers in Nutrition* 7. https://doi.org/10.3389/ fnut.2020.00035.

France, Agence Nationale de la Recherce (2022). "Développer les protéines végétales et diversifier les sources de protéines" Volet 2 – Protéines de légumineuses- Appel à projets – 2022 https://anr.fr/en/france-2030/ call-for-proposals details/call/6887aaedc17bb0a03ea17b95e73291f6/.

Fremstad, S., Rho, H.J. and Brown, H. (2020). Meatpacking

Gautam, M., Laborde, D., Mamun, A., Martin, W., Piñeiro, V. and Vos, R. (2022). Repurposing agricultural policies and support: Options to transform agriculture and food systems to better serve the health of people, economies and the planet. Washington, D.C.: World Bank. http://hdl.handle. net/10986/36875.

Gehring, J., Touvier, M., Baudry, J., Julia, C., Buscail, C., Srour, B. *et al.* (2021). Consumption of ultra-processed foods by pesco-vegetarians, vegetarians, and vegans: Associations with duration and age at diet initiation. *Journal of Nutrition* 151(1), 120-131. https://doi.org/10.1093/jn/nxaa196.

Gerbens-Leenes, P.W., Mekonned, M.M. and Hoekstra, A.Y. (2013). The water footprint of poultry, pork and beef: A comparative study in different countries and production systems. *Water Resources and Industry* 1–2, 25–36. https://doi.org/10.1016/j.wri.2013.03.001.

Gerbens-Leenes, P.W., Nonhebel, S. and Krol, M.S. (2010). Food consumption patterns and economic growth. Increasing affluence and the use of natural resources. *Appetite* 55(3), 597–608. https://doi.org/10.1016/j. appet.2010.09.013.

Gerber, P.J., Steinfeld, H., Henderson, B., Mottet, A., Opio, C., Dijkman, J. et al. (2013). Tackling Climate Change Through Livestock: A Global Assessment of Emissions and Mitigation Opportunities. Rome: Food and Agriculture Organization of the United Nations. https://www.fao.org/3/ i3437e/i3437e.pdf.

workers are a diverse group who need better protections. Center for Economic and Policy Research. https://cepr.net/ meatpacking-workers-are-a-diverse-group-who-need-better-protections/.

Fresán, U., Marrin, D.L., Mejia, M.A., Sabaté, J. (2019) Water footprint of meat analogs: Selected indicators according to life cycle assessment. *Water* 2019, *11*, 728. https://doi.org/10.3390/w11040728.

Fukase, E. and Martin, W. (2020). Economic growth, convergence, and world food demand and supply. *World Development* 132, 104954. https://doi.org/10.1016/j.worlddev.2020.104954.

Funke, F., Mattauch, L., Bijgaart, I.V.D., Godfray, H.C.J., Hepburn, C., Klenert, D. et al. (2022). Toward optimal meat pricing: Is it time to tax meat consumption? *Review of Environmental Economics and Policy* 16(2), 219-240. https://www.journals.uchicago.edu/doi/10.1086/721078.

Galloway, J.N., Burke, M., Bradford, G.E., Naylor, R., Falcon, W., Chapagain, A.K. et al. (2007). International trade in meat: The tip of the pork chop. *AMBIO: A Journal of the Human Environment* 36(8), 622–629. https://doi. org/10.1579/0044-7447(2007)36[622:ITIMTT]2.0.C0;2.

Galves, C., Stone, A.K., Szarko, J., Liu, S., Shafer, K., Hargreaves, J. *et al.* (2019). Effect of pH and defatting on the functional attributes of safflower, sunflower, canola and hemp protein concentrates. *Cereal Chemistry* 96(6), 1036–1047. https://doi.org/10.1002/cche.10209. Gerber, P., Chilonda, P., Franceschini, G. and Menzi, H. (2005). Geographical determinants and environmental implications of livestock production intensification in Asia. *Bioresource Technology* 96(2), 263–276. https://doi.org/10.1016/j.biortech.2004.05.016.

Gerhardt, C., Ziemßen, F., Warschun, M., Suhlmann, G., Donnan, D., Kühnle, H. (2019). How will cultivated meat and meat alternatives disrupt the agricultural and food industry. A.T. Kearney. https://gastronomiaycia.republica. com/wp-content/uploads/2019/06/estudio_futuro_alimentos.pdf.

Giampietro, M., Aspinall, R., Bukkens, S., Benalcazar, J., Diaz-Maurin, F., Flammin, A. et al. (2013). An Innovative Accounting Framework for the Food-Energy-Water Nexus: Application of the MuSIASEM Approach to Three Case Studies. Environment and Natural Resources Management Working Paper No. 56. Rome: Food and Agriculture Organization of the United Nations. https://www.researchgate.net/publication/273362115_An_Innovative_ Accounting_Framework_for_the_Food-Energy-Water_Nexus_Application_ of_the_MuSIASEM_approach_to_three_case_studies.

Gibbs, J. and Cappuccio, F.P. (2022). Plant-based dietary patterns for human and planetary health. *Nutrients* 14(8), 1614. https://doi.org/10.3390/ nu14081614.

Garcês, A., Pires, I. and Rodrigues, P. (2020). Teratological effects of pesticides in vertebrates: A review. *Journal of Environmental Science and Health, Part B* 55(1), 75–89. https://doi.org/10.1080/03601234.2019.1660562.

Gastaldello, A., Giampieri, F., De Giuseppe, R., Grosso, G., Baroni, L. and Battino, M. (2022). The rise of processed meat alternatives: A narrative review of the manufacturing, composition, nutritional profile and health effects of newer sources of protein, and their place in healthier diets. *Trends in Food Science and Technology* 127, 263–271. https://doi. org/10.1016/j.tifs.2022.07.005. Glibert, P.M. (2017). Eutrophication, harmful algae and biodiversity– Challenging paradigms in a world of complex nutrient changes. *Marine Pollution Bulletin*, 124(2), 591-606. https://doi.org/10.1016/j. marpolbul.2017.04.027.

Gleckel, J. (2020). Are consumers really confused by plant-based food labels? An empirical study. *University of Louisville, Louis D. Brandeis School of Law, Journal of Animal and Environmental Law.* https://ssrn.com/ abstract=3727710. 60

Glibert, P.M. (2017). Eutrophication, harmful algae and biodiversity
– Challenging paradigms in a world of complex nutrient changes.
Marine Pollution Bulletin 124(2), 591–606. https://doi.org/10.1016/j.
marpolbul.2017.04.027.

ClimateWorks Foundation (2023) Global innovation needs assessments: Food system Methane. https://www.climateworks.org/ginas-methane/.

Godfray, H.C.J., Aveyard, P., Garnett, T., Hall, J.W., Key, T.J., Lorimer, J. *et al.* (2018). Meat consumption, health, and the environment. *Science* 361(6399). https://doi.org/10.1126/science.aam5324.

Goldstein, B., Moses, R., Sammons, N. and Birkved, M. (2017). Potential to curb the environmental burdens of American beef consumption using a novel plant-based beef substitute. PLOS ONE. https://doi.org/10.1371/journal.pone.0189029

The Good Food Institute. (2022a). 2021 State of the Industry Report: Plant-Based Meat, Seafood, Eggs and Dairy. The Good Food Institute. https://gfi. org/wp-content/uploads/2022/04/2021-Plant-Based-State-of-the-Industry-Report-1.pdf.

The Good Food Institute (2022b). 2021 State of the Industry Report: Cultivated Meat and Seafood. https://gfi.org/wp-content/ Hall, K.D., Ayuketah, A., Brychta, R., Cai, H., Cassimatis, T., Chen, K.Y. *et al.* (2019). Ultra-processed diets cause excess calorie intake and weight gain: An inpatient randomized controlled trial of ad libitum food intake. *Cell Metabolism* 30(1), 67-77.

Hamza, M.A. and Anderson, W.K. (2005). Soil compaction in cropping systems: A review of the nature, causes and possible solutions. *Soil and Tillage Research*, 82(2), 121-145. https://doi.org/10.1016/j.still.2004.08.009

Hashempour-Baltork, F., Khosravi-Darani, K., Hosseini, H., Farshi, P. and Reihani, S.F.F (2020). Mycoproteins as safe meat substitutes. *Journal of Cleaner Production* 253, 119958. https://doi.org/10.1016/j. jclepro.2020.119958.

Hawkes, C., Chopra, M. and Friel, S. (2009). Globalization, trade and the nutrition transition. In *Globalization and Health: Pathways, Evidence and Policy*.
Labonté, R., Schrecker T., Parker, C. and Runnerls, V. (eds.). New York:
Routledge. 235-262.

Hayek, M.N. and Garrett, R.D. (2018). Nationwide shift to grass-fed beef requires larger cattle population. *Environmental Research Letters* 13(8), 084005. https://doi.org/10.1088/1748-9326/aad401.

Hayek, M.N., Harwatt, H., Ripple, W.J. and Mueller, N.D. (2021). The carbon

uploads/2022/04/2021-Cultivated-Meat-State-of-the-Industry-Report-1.pdf.

The Good Food Institute (2023a). State of Global Policy 2022. https://gfi.org/ wp-content/uploads/2023/01/State_of_Global_Policy_Report_2022.pdf.

The Good Food Institute (2023b). 2022 State of the Industry Report: Fermentation: Meat, seafood, eggs and dairy. https://gfi.org/wp-content/ uploads/2023/01/2022-Fermentation-State-of-the-Industry-Report-1.pdf.

The Good Food Institute (2023c). 2022 State of the Industry Report: Cultivated Meat and Seafood. https://gfi.org/wp-content/ uploads/2022/04/2021-Cultivated-Meat-State-of-the-Industry-Report-1.pdf.

The Good Food Institute and SPINS (2021). 2020 U.S. retail market data for the plant-based industry. https://gfi.org/marketresearch/. Accessed 25 August.

GOOD Meat (2023). GOOD Meat receives approval to commercialize serum-free media. 18 January. https://www.goodmeat.co/all-news/good-meat-receives-approval-to-commercialize-serum-free-media.

Goedkoop, M., Heijungs, R., Huijbregts, M., De Schryver, A., Struijs, J. and Van Zelm, R. (2009). ReCiPe 2008. A life cycle impact assessment method which comprises harmonised category indicators at the midpoint and the endpoint level 1, 1-126. Space and Environment, Ministry of Housing, The Netherlands. https://dvikan.no/ntnu-studentserver/reports/selected%20 sections%20-%20Goedkoop%20etal%20ReCiPe_main_report_final_27-02-2009_web.pdf. opportunity cost of animal-sourced food production on land. *Nature Sustainability* 4(1), 21-24. https://doi.org/10.1038/s41893-020-00603-4.

Heinke, J., Lannerstad, M., Gerten, D., Havlík, P., Herrero, M., Notenbaert, A.M.O. *et al.* (2020). Water use in global livestock production—opportunities and constraints for increasing water productivity. *Water Resources Research* 56(12), e2019WR026995. https://doi. org/10.1029/2019WR026995.

Henchion, M., Moloney, A.P., Hyland, J., Zimmermann, J. and McCarthy, S. (2021). Trends for meat, milk and egg consumption for the next decades and the role played by livestock systems in the global production of proteins. *Animal*, *15*, 100287. https://doi.org/10.1016/j.animal.2021.100287

Hendriks, S., de Groot Ruiz, A., Herrero Acosta, M., Baumers, H., Galgani,
P., Mason-D'Cros, D. *et al.* (2023). The true cost of food: A preliminary assessment. In *Science and Innovations for Food Systems Transformation*. von Braun, J., Afsana, K., Fresco, L.O. and Hassan, M.H.A. (eds.) Cham: Springer. 581–601. https://doi.org/10.1007/978-3-031-15703-5_32.

Heredia, N. and García, S. (2018). Animals as sources of food-borne pathogens: A review. *Animal Nutrition* 4(3), 250–255. https://doi.org/10.1016/j. aninu.2018.04.006.

Herrero, M., Henderson, B., Havlík, P., Thornton, P.K., Conant, R.T., Smith, P. et al. (2016). Greenhouse gas mitigation potentials in the livestock sector. *Nature Climate Change* 6(5), 452–461. https://doi.org/10.1038/nclimate2925.

Grasso, S. and Jaworska, S. (2020). Part meat and part plant: Are hybrid meat products fad or future? *Foods* 9(12), 1888. https://doi.org/10.3390/ foods9121888.

Greger, M. and Koneswaran, G. (2010). The public health impacts of concentrated animal feeding operations on local communities. *Family and Community Health* 33(1),11–20. https://doi.org/10.1097/ fch.0b013e3181c4e22a.

Guthman, J. (2022). The CAFO in the bioreactor: Reflections on efficiency logics in bio-industrialization present and future. *Environmental Humanities* 14(1), 71-88. https://doi.org/10.1215/22011919-9481440.

Herrero, M., Grace, D., Njuki, J., Johnson, N., Enahoro, D., Silvestri, S., & Rufino, M. (2013). The roles of livestock in developing countries: the good, the bad and the knowledge gaps. *Animal*, 7(s1), 3–18. http://dx.doi. org/10.1017/S1751731112001954

Herrero, M., Mason-D'Croz, D., Thornton, P.K., Fanzo, J., Rushton, J.,
Godde, C. *et al.* (2023). Livestock and sustainable food systems:
Status, trends and priority actions. In *Science and Innovations for Food Systems Transformation.* von Braun, J., Afsana, K., Fresco, L.O.
and Hassan, M.H.A. (eds.) Cham: Springer. 375-399. https://doi.
org/10.1007/978-3-031-15703-5_20.
61

Herrero, M., Thornton, P.K., Gerber, P. and Reid, R.S. (2009). Livestock, livelihoods and the environment: Understanding the trade-offs. *Current Opinion in Environmental Sustainability* 1(2), 111-120. https://doi.org/10.1016/j. cosust.2009.10.003.

Hertzler, S., Lieblein-Boff, J.C., Weiler, M. and Allgeier, C. (2020). Plant proteins: Assessing their nutritional quality and effects on health and physical function. *Nutrients* 12(12), 3704. https://doi.org/10.3390/nu12123704.

Hocquette, J.F., Ellies-Oury, M.P., Lherm, M., Pineau, C., Deblitz, C. and Farmer, L. (2018). Current situation and future prospects for beef production in Europe—A review. *Asian-Australasian Journal of Animal Sciences*, *31*(7), 1017. https://doi.org/10.5713/ajas.18.0196.

Hoekstra, A.Y., Chapagain, A.K., Aldaya, M.M. and Mekonnen,
M.M. (2009). Water Footprint Manual. Water footprint network.
Enschede, The Netherlands. https://waterfootprint.org/resources/
TheWaterFootprintAssessmentManual_English.pdf.

Holmes, D., Humbird, D., Dutkiewicz, J., Tejeda-Saldana, Y., Duffy, B. and Datar, I. (2022). Cultured meat needs a race to mission not a race to market. *Nature Food* 3, 785-787. https://doi.org/10.1038/ s43016-022-00586-9. Intergovernmental Panel on Climate Change (2021). Summary for Policymakers. In Climate Change 2021 – The Physical Science Basis: Working Group I Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S.L., Péan, C., Chen, Y. et al. (eds.). Cambridge: Cambridge University Press. 3–32. https://doi.org/10.1017/978100915789 6.001.

Intergovernmental Panel on Climate Change (2022). Climate Change 2022 – Impacts, Adaptation and Vulnerability: Working Group II Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge: Cambridge University Press. https://doi. org/10.1017/9781009325844.

Intergovernmental Panel on Climate Change (2023). *Climate Change 2023: Synthesis Report. Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change.* Core Writing Team, Lee, H. and Romero, J. (eds.). Geneva. https://www.ipcc.ch/ report/sixth-assessment-report-cycle/.

Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (2016). The Assessment Report of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services on Pollinators,

Holmes, S. (2020). Climate change, sustainability and competition law. Journal of Antitrust Enforcement 8(2), 354–405. https://doi.org/10.1093/ JAENFO/JNAA006.

Howard, P.H. (2022). Cellular agriculture will reinforce power asymmetries in food systems. *Nature Food* 3, 198-800. https://doi.org/10.1038/ s43016-022-00609-5.

Hribar, C. and Schultz, M. (2010). Understanding Concentrated Animal Feeding Operations and Their Impact on Communities. Bowling Green, OH: The National Association of Local Boards of Health. https://www.cdc.gov/ nceh/ehs/docs/understanding_cafos_nalboh.pdf. Accessed13 June 2022.

Hu, F.B., Otis, B.O. and McCarthy, G. (2019). Can plant-based meat alternatives be part of a healthy and sustainable diet? *JAMA* 322(16), 1547–1548. https://doi.org/10.1001/jama.2019.13187.

Humbird, D. (2021) Scale-up economics for cultured meat. *Biotechnology and Bioengineering*. 118(8), 3239–3250. https://doi.org/10.1002/bit.27848.

Humpenöder, F., Bodirsky, B.L., Weindl, I., Lotze-Campen, H., Linder, T. and Popp, A. (2022). Projected environmental benefits of replacing beef with microbial protein. *Nature* 605, 90-96. https://doi.org/10.1038/ s41586-022-04629-w.

Ijaz, M., Yar, M.K., Badar, I.H., Ali, S., Islam, M.S., Jaspal, M.H. et al. (2021). Meat production and supply chain under Covid-19 scenario: Current trends and future prospects. *Frontiers in Veterinary Science 8*. https://doi. org/10.3389/fvets.2021.660736. *Pollination and Food Production*. Potts, S.G., Imperatriz-Fonzeca, V.L. and Ngo, H.T. (eds.). Bonn. https://doi.org/10.5281/zenodo.3402856.

Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (2019). *Global Assessment Report on Biodiversity and Ecosystem Services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services*. Brondizio, E.S., Settele, J., Diaz, S. and Ngo, H.T. (eds.). Bonn. https://doi.org/10.5281/zenodo.3831673.

Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (2020). Workshop Report on Biodiversity and Pandemics of the Intergovernmental Platform on Biodiversity and Ecosystem Services. Bonn. https://www.ipbes.net/sites/default/files/2020-12/IPBES%20 Workshop%20on%20Biodiversity%20and%20Pandemics%20Report_0.pdf.

International Food Information Council (2023). 2023 Food and Health Survey. https://foodinsight.org/wp-content/uploads/2023/05/IFIC-2023-Food-Health-Report.pdf.

International Livestock Research Institute (2019). Options for the Livestock Sector in Developing and Emerging Economies to 2030 and Beyond. Meat: The Future Series. Cologny/Geneva: World Economic Forum. https://www3. weforum.org/docs/White_Paper_Livestock_Emerging%20Economies.pdf.

International Livestock Research Institute, International Union for Conservation of Nature, Food and Agricultural Organisation of the United Nations, World Wide Fund for Nature, United Nations Environment Programme and the International Land Coalition (2021). Rangelands Atlas. Nairobi, Kenya. https://cgspace.cgiar.org/handle/10568/114064

Intergovernmental Panel on Climate Change (2006). 2006 IPCC Guidelines for National Greenhouse Gas Inventories. Eggleston, H.S., Buendia, L., Miwa, K., Ngara, T. and Tanabe, K.,(eds.). Bonn. https://www.ipcc.ch/ report/2006-ipcc-guidelines-for-national-greenhouse-gas-inventories/.

Intergovernmental Panel on Climate Change (2014). Climate Change 2014: Mitigation of Climate Change. Working Group III Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge: Cambridge University Press. https://www.ipcc.ch/site/assets/ uploads/2018/02/ipcc_wg3_ar5_frontmatter.pdf. The International Panel of Experts on Sustainable Food Systems (2022). The Politics of Protein: Examining Claims about Livestock, Fish, 'Alternative Proteins' and Sustainability. https://www.ipes-food.org/_img/upload/files/ PoliticsOfProtein.pdf.

Italy, Camera dei Deputati (2023) Conversione in legge, con modificazioni, del decreto-legge 29 Settembre 2023, n. 132, recante disposizioni urgenti in materia di proroga di termini normativi e versamenti fiscali. https://documenti.camera.it/leg19/pdl/pdf/leg.19.pdl.camera.1551.19MSG0062840. pdf. Accessed 24 November 2023. 62

Ives, M. (2020). Singapore approves a lab-grown meat product, a global first. *The New York Times*, 2 December. https://www.nytimes.com/2020/12/02/ business/singapore-lab-meat.html. Accessed 18 July 2023.

Jackson, R.B., Saunois, M., Bousquet, P., Canadell, J.G., Poulter, B., Stavert, A.R. *et al.* (2020). Increasing anthropogenic methane emissions arise equally from agricultural and fossil fuel sources. *Environmental Research Letters* 15(7), 071002. https://iopscience.iop.org/ article/10.1088/1748-9326/ab9ed2.

Jankielsohn, A. (2015). The hidden cost of eating meat in South Africa: What every responsible consumer should know. *Journal of Agricultural and Environmental Ethics*, 28, 1145-1157. https://philpapers.org/rec/JANTHC.

Järviö, N., Maljanen, N., Kobayashi, Y., Ryynanen, T., Tuomisto, H. (2021). An attributional life cycle assessment of microbial protein production: A case study on using hydrogen-oxidizing bacteria. *Science of The Total Environment* 776, 145764. <u>https://doi.org/10.1016/j.</u> scitotenv.2021.145764.

Jochems, C.E.A., van der Valk, J.B., Stafleu, F.R. and Baumans, V. (2002). The use of fetal bovine serum: Ethical or scientific problem? *Alternatives to Laboratory Animals* 30(2), 219–227. https://doi.org/10.1177/02611929020 Kazemi, A., Barati-Boldaji, R., Soltani, S., Mohammadipoor, N., Esmaeilinezhad, Z., Clark, C.C.T. *et al.* (2021). Intake of various food groups and risk of breast cancer: A systematic review and dose-response meta-analysis of prospective studies. *Advances in Nutrition* 12(3), 809– 849. https://doi.org/10.1093/advances/nmaa147.

Kim, H., Caulfield, L.E. and Rebholz, C.M. (2018). Healthy plant-based diets are associated with lower risk of all-cause mortality in US adults. *The Journal of Nutrition* 148(4), 624–631. https://doi.org/10.1093/jn/nxy019.

Kim, H., Caulfield, L.E., Garcia-Larsen, V., Steffen, L.M., Coresh, J. and Rebholz, C.M. (2019). Plant-based diets are associated with a lower risk of incident cardiovascular disease, cardiovascular disease mortality and all-cause mortality in a general population of middle-aged adults. *Journal of the American Heart Association* 8(16), e012865. https://doi.org/10.1161/jaha.119.012865.

Kim, S., Beier, A., Schreyer, H.B. and Bakshi, B.R. (2022). Environmental life cycle assessment of a novel cultivated meat burger patty in the United States. *Sustainability* 14(23), 16133. http://dx.doi.org/10.3390/ su142316133.

Kinney, M.J. (2019). Formulating with Animal-Free Ingredients. The Good Food Institute. https://gfi.org/wp-content/uploads/2021/01/

3000208.

Johnson, K.M. and Lichter, D.T. (2019). Rural depopulation: growth and decline processes over the past century. *Rural Sociology* 84(1), 3–27. https://doi.org/10.1111/ruso.12266.

Johnson, M. (2006). Major technological advances and trends in cheese. Journal of Dairy Science 89(4): 1174-8. http://dx.doi.org/10.3168/jds. S0022-0302(06)72186-5.

Johnston, J. and Cushing, L. (2020). Chemical exposures, health and environmental justice in communities living on the fenceline of industry. *Current Environmental Health Reports* 7(1), 48–57. https://doi.org/10.1007/ s40572-020-00263-8.

Jones, B.A., Grace, D., Kock, R., Alonso, S., Rushton, J., Said, M.Y. *et al.* (2013). Zoonosis emergence linked to agricultural intensification and environmental change. *Proceedings of the National Academy of Sciences* 110(21), 8399–8404. https://doi.org/10.1073/pnas.1208059110.

Jönsson, E. (2020). On breweries and bioreactors: Probing the 'present futures' of cellular agriculture. *Transactions of the Institute of British Geographers* 45, 921–936. https://rgs-ibg.onlinelibrary.wiley.com/doi/ epdf/10.1111/tran.12392.

Just Rural Transition (2023). *Principles for Just Food System Transitions*. https://justruraltransition.org/wp-content/uploads/sites/12/2023/04/JRT_ IFTJune2019Animalfreeformulations.pdf.

Kinney, M.J., Weston, Z. and Bauman, J.D. (2019). *Plant-based Meat Manufacturing by Extrusion*. The Good Food Institute. https://gfi.org/images/ uploads/2019/11/Plant-Based-Meat-Manufacturing-Guide-_GFI.pdf.

Kityo, A., Lee, S.A. and Kang, D. (2023). Total and cause-specific mortality associated with meat intake in a large cohort study in Korea. *Frontiers in Nutrition* 10. https://doi.org/10.3389/fnut.2023.1138102.

Kleppel, G.S. (2020). Do differences in livestock management practices influence environmental impacts? *Frontiers in Sustainable Food Systems* 4, 141. https://doi.org/10.3389/fsufs.2020.00141.

Kolkmann, A.M., Van Essen, A., Post, M., Moutsatsou, P. (2022). Development of a chemically defined medium for *in vitro* expansion of primary bovine satellite cells. *Frontiers in Bioengineering Biotechnology* 10. https://doi.org/10.3389/fbioe.2022.895289.

Koutinas, A. A., Vlysidis, A., Pleissner, D., Kopsahelis, N., Lopez Garcia, I., Kookos, I.K. *et al.* (2014). Valorization of industrial waste and by-product streams *via* fermentation for the production of chemicals and biopolymers. *Chemical Society Reviews* 8, 2587-2627. https://doi.org/10.1039/C3CS60293A.

Ku, H.L. and Ha, S.H. (2020). Improving nutritional and functional quality by genome editing of crops: Status and perspectives. *Frontiers in Plant Science* 11. https://doi.org/10.3389/fpls.2020.577313.

Principles_Report_170423.pdf.

Kaljonen, M., Kortetmäki, T., Tribaldos, T., Huttunen, S., Karttunen, K., Maluf, R.S. *et al.* (2021) Justice in transitions: Widening considerations of justice in dietary transition. *Environmental Innovation and Societal Transitions*, 40, 474-485. https://doi.org/10.1016/j.eist.2021.10.007.

Kamalapuram, S.K., Handral, H. and Choudhury, D. (2021). Cultured meat prospects for a billion! *Foods* 10(12), 2922. https://doi.org/10.3390/ foods10122922.

Karwowska, M., Łaba, S. and Szczepański, K. (2021). Food loss and waste in meat sector—why the consumption stage generates the most losses?. *Sustainability*, *13*(11), 6227. https://doi.org/10.3390/su13116227

Kyriakopoulou K., Keppler, J. and van der Goot, A.J. (2021). Functionality of ingredient and additives in plant-based meat analogues. *Foods* 10(3):600. https://doi.org/10.3390/foods10030600.

Lähteenmäki-Uutela, N., Rahikainen, M., Lonkila, A. and Yang, B. (2021). Alternative proteins and the EU food law. *Food Control* 130, 108336. https:// doi.org/10.1016/j.foodcont.2021.108336.

Lam, Y., Fry, J.P., Hu, E., Kim, B.F., Nachman K.E. (2017). *Industrial Food Animal Production in Low- and Middle-Income Countries: A Landscape Assessment*. Baltimore: John's Hopkins Center for a Livable Future. http://www.jhsph.edu/ research/centers-and-institutes/johns-hopkins-center-for-a-livable-future/_ pdf/projects/IFAP/IFAPLowmid_income_countriesWeb1.pdf. The Lancet Editorial Board (2020). Sweet success: Will sugar taxes improve health? The Lancet Diabetes and Endocrinology 5(4), P235. https://doi. org/10.1016/S2213-8587(17)30070-0.

Langemeier, M. and Boehlje, M. (2017). Drivers of consolidation and structural change in production agriculture. *Economic Review, Special Issue*. 5–23. https://ag.purdue.edu/commercialag/ home/wp-content/uploads/2017/10/201710_LangemeierBoehlje_ DriversOfConsolidationAndStructuralChange.pdf.

Latino, L.R., Pica-Ciamarra, U. and Wisser, D. (2020). Africa: The livestock revolution urbanizes. *Global Food Security* 26, 100399. https://doi. org/10.1016/j.gfs.2020.100399.

Lawrence, M.A. and Baker, P.I. (2019). Ultra-processed food and adverse health outcomes. *BMJ* 365, I2289. https://doi.org/10.1136/bmj.I2289.

Li, M., Havelaar, A.H., Hoffmann, S., Hald, T., Kirk, M.D., Torgerson, P.R. *et al.* (2019). Global disease burden of pathogens in animal source foods, 2010. *PLoS ONE*, 14(6), e0216545. https://doi.org/10.1371/journal.pone.0216545.

Liang, Y. and Lee, D. (2022). Recent progress of cultivated meat in Asia. *Food Materials Research* 2(12). https://doi.org/10.48130/FMR-2022-0012.

Maeder, B., Walter H., Dongoski, R. and Krupke, T. (2023) How alternative proteins are reshaping meat industries. EY Parthenon. https://www.ey.com/ en_gl/strategy/how-alternative-proteins-are-reshaping-meat-industries#:~:text=A%20distinct%20advantage%20of%20plant,can%20be%20far%20 more%20agile.

Malinauskaite, J. (2022). Competition law and sustainability: EU and national perspectives. *Journal of European Competition Law and Practice* 13(5), 336–348. https://doi.org/10.1093/JECLAP/LPAC003.

Mancini, C. and Antonioli, F. (2022). The future of cultured meat between sustainability expectations and socio-economic challenges. In *Future Foods*. Bhat, R. (ed.). London: Academic Press. 331-350. https://doi. org/10.1016/B978-0-323-91001-9.00024-4.

Marcellin, E., Angenent, L.T., Nielsen, L.K., and Molitor, B. (2022). Recycling carbon for sustainable protein production using gas fermentation. *Current Opinion in Biotechnology* 76, 102723. https://doi.org/10.1016/j. copbio.2022.102723.

Marwaha, N., Beveridge, M.C.M. and Phillips, M.J. (2022). Fad, food or feed: Alternative seafood and its contribution to food systems. *Frontiers in Sustainable Food Systems* 6. https://www.frontiersin.org/article/10.3389/

Lianou, A., Panagou, E.Z. and Nychas, G.J.E. (2017). Meat safety—I foodborne pathogens and other biological issues. In *Lawrie's Meat Science*. Toldra, F. (ed.). Duxford: Woodhead Publishing. 521–552. https://doi.org/1 0.1016%2FB978-0-08-100694-8.00017-0.

Lima, J.R. and Mellinger, C.G. (2022). Pulses-derived proteins for the plantbased market: Opportunities to reduce postharvest loss and waste. *Current Opinion in Biotechnology 78*, 102820. https://doi.org/10.1016/j. copbio.2022.102820.

Liu, Q., Wang, J., Bai, Z., Ma, L. and Oenema, O. (2017). Global animal production and nitrogen and phosphorus flows. *Soil Research* 55(6), 451-462. https://doi.org/10.1071/SR17031.

Lupo, C.D., Clay, D.E., Benning, J.L. and Stone, J.J. (2013). Life-cycle assessment of the beef cattle production system for the northern great plains, USA. *Journal of Environmental Quality* 42(5), 1386-1394. https://doi. org/10.2134/jeq2013.03.0101.

Ma, X., Mahecha, M.D., Migliavacca, M., van der Plas, F., Benavides, R. et al. (2019). Inferring plant functional diversity from space: The potential of Sentinel-2. *Remote Sensing of Environment* 233, 111368. https://doi. org/10.1016/j.rse.2019.111368.

MacDonald, J.M. (2017). Consolidation, concentration and competition in the food system. *Economic Review, Special Issue* 85-105. https://www.kansascityfed.org/documents/765/2017-Consolidation,%20Concentration,%20 and%20Competition%20in%20the%20Food%20System.pdf. fsufs.2022.750253.

Mason-D'Croz, D., Barnhill, A., Bernstein, J., Bogard, J., Dennis, G., Dixon, P. et al. (2022). Ethical and economic implications of the adoption of novel plant-based beef substitutes in the USA: A general equilibrium modelling study. *The Lancet Planetary Health* 6(8), e658–e669. https://doi. org/10.1016/S2542-5196(22)00169-3.

Mathur, M.B. (2022). Ethical drawbacks of sustainable meat choices. *Science* 375(6587), 1362. https://doi.org/10.1126/science.abo2535.

Mathur, M.B., Peacock, J., Reichling, D.B., Nadler, J., Bain, P.A., Gardner, C.D. *et al.* (2021). Interventions to reduce meat consumption by appealing to animal welfare: Meta-analysis and evidence-based recommendations. *Appetite* 164, 105277. https://doi.org/10.1016/j.appet.2021.105277.

Mattick, C.S., Landis, A.E., and Allenby, B.R. (2015a). A case for systemic environmental analysis of cultured meat. *Journal of Integrative Agriculture* 14(2), 249–254. https://doi.org/10.1016/S2095-3119(14)60885-6.

Mattick, C.S., Landis, A.E., Allenby, B.R. and Genovese, N.J. (2015b). Anticipatory life cycle analysis of in vitro biomass cultivation for cultured meat production in the United States. *Environmental Science and Technology* 49(19), 11941-11949. https://doi.org/10.1021/acs. est.5b01614.

Mattson (2017). *Five Food and Beverage Macro Trends for 2017*. https:// www.mattsonco.com/wp-content/uploads/2017/04/Mattson_ MacroTrends_2017.pdf.

Machovina, B., Feeley, K.J. and Ripple, W.J. (2015). Biodiversity conservation: The key is reducing meat consumption. *Science of The Total Environment* 536, 419–431. https://doi.org/10.1016/j.scitotenv.2015.07.022.

Madzingira, O. (2018). Animal welfare considerations in food-producing animals. In *Animal Welfare*. Abubakar, M. and Manzoor, S. (eds.). Rijeka: IntechOpen. https://doi.org/10.5772/intechopen.78223.

Malan, H.J. (2020). Swap the meat, save the planet: A community-based participatory approach to promoting healthy, sustainable food in a university setting. Ph.D. Thesis. University of California, Los Angeles. Available at: https://escholarship.org/uc/item/3pq4t2mz. McBee Cattle Company (2022). https://mcbeecattlecompany.com.

McClements, D.J. and Grossmann, L. (2022). Nutritional and health aspects. In *Next-Generation Plant-based Foods: Design, Production and Properties*: McClements, D.J. and Grossmann, L. (eds.). Cham: Springer International Publishing. 227–284. https://download.bibis.ir/Books/Food/2022/ Next-Generation-Plant-based-Foods-Design,-Production,-and-Properties-(David-Julian-McClements,-Lutz-Grossmann)_bibis.ir.pdf. McCormack, S. (2021). Climate change and animal agriculture: Federal actions protect the biggest contributors from the disasters they cause. *Environmental Law* 51(3), 745 – 769. https://www.jstor.org/stable/48628532.

Mekonnen, M.M. and Hoekstra, A.Y. (2011). The green, blue and grey water footprint of crops and derived crop products. *Hydrology and Earth System Sciences*, *15*(5),1577-1600. https://doi.org/10.5194/hess-15-1577-2011

Melzener, L., Verzijden, K.E., Buijs, A.J., Post, M.J. and Flack, J.E. (2020). Cultured beef: From small biopsy to substantial quantity. *Journal of Science of Food and Agriculture* 101(1), 7–14. https://doi.org/10.1002/ jsfa.10663.

Messina, M., Sievenpiper, J.L., Williamson, P., Kiel, J. and Erdman Jr., J.W. (2022). Perspective: Soy-based meat and dairy alternatives, despite classification as ultra-processed foods, deliver high-quality nutrition on par with unprocessed or minimally processed animal-based counterparts. *Advances in Nutrition* 13(3), 726–738. https://doi.org/10.1093/advances/nmac026.

Messmer, T., Klevernic, I., Furquim, C., Ovchinnikova, E., Dogan, A., Cruz, H. et al. (2022). A serum-free media formulation for cultured meat production supports bovine satellite cell differentiation in the absence of serum starvation. *Nature Food* 3(1), 74–85. https://doi.org/10.1038/ s43016-021-00419-1. Morais-da-Silva, R.L., Villar, E.G., Reis, G.G., Sanctorum, H. and Molento, C.F.M. (2022). The expected impact of cultivated and plant-based meats on jobs: The views of experts from Brazil, the United States and Europe. *Humanities and Social Sciences Communications* 9, 297. https://doi. org/10.1057/s41599-022-01316-z.

Morand, S. (2020). Emerging diseases, livestock expansion and biodiversity loss are positively related at global scale. *Biological Conservation* 248, 108707. https://doi.org/10.1016/j.biocon.2020.108707.

Morrison, O. (2023). First application for cultivated meat approval in Europe submitted. *Food Navigator Europe*, 26 July. https://www.foodnavigator.com/Article/2023/07/26/ First-application-for-cultivated-meat-approval-in-Europe-submitted.

Morrison Paul, C.J. (2003). Cost economies: A driving force for consolidation and concentration? *Southern Economic Journal* 70(1), 110-127. https:// doi.org/10.2307/1061634.

Mosquera, J. (2018). Corn, cows, and cash: How farming subsidies work and what they could potentially achieve. *Journal of Land Use & Environmental Law*, 34, 191. https://www.jstor.org/stable/26896702

Motta, R. (2021). Social movements as agents of change: Fighting

Meyer, H.P., Minas, W. and Schmidhalter, D. (2016). Industrial-scale fermentation. In *Industrial Biotechnology: Products and Processes*.
Wittmann, C. and Liao, J.C. (eds.). Weinheim: Wiley-VCH. 1-53. https://doi. org/10.1002/9783527807833.ch1.

Milford, A.B., Le Mouël, C., Bodirsky, B.L. and Rolinski, S. (2019). Drivers of meat consumption. *Appetite* 141, 104313. https://doi.org/10.1016/j. appet.2019.06.005.

Miller, V., Reedy, J., Cudhea, F., Zhang, J., Shi, P., Erndt-Marino, J. *et al.* (2022). Global, regional and national consumption of animal-source foods between 1990 and 2018: Findings from the Global Dietary Database. *The Lancet Planetary Health* 6(3), e243-e256. https://doi.org/10.1016% 2FS2542-5196(21)00352-1.

Miller, V., Webb, P., Cudhea, F., Zhang, J., Reedy, J., Shi, P. *et al.* (2023). Children's and adolescents' rising animal-source food intakes in 1990– 2018 were impacted by age, region, parental education and urbanicity. *Nature Food* 4, 305-319. https://doi.org/10.1038/s43016-023-00731-y.

Modlinska, K., Adamczyk, D., Maison, D., and Pisula, W. (2020). Gender differences in attitudes to vegans/vegetarians and their food preferences, and their implications for promoting sustainable dietary patterns–A systematic review. *Sustainability* 12(16), 6292. https://doi.org/10.3390/su12166292. intersectional food inequalities, building food as webs of life. *The Sociological Review* 69(3), 603-625. https://doi.org/10.1177/003802612 11009061.

Mottet, A., de Haan, C., Falcucci, A., Tempio, G., Opio, C., Gerber, P.J. (2017). Livestock: On our plates or eating at our table? A new analysis of the feed/ food debate. *Global Food Security* 14, 1–8. http://dx.doi.org/10.1016/j. gfs.2017.01.001.

Mouat, M.J., and Prince, R. (2018). Cultured meat and cowless milk: On making markets for animal-free food. *Journal of Cultural Economy* 11(4), 315–329. https://doi.org/10.1080/17530350.2018.1452277.

Mouat, M.J., Prince, R. and Roche, M.M. (2019). Making value out of ethics: The emerging economic geography of lab-grown meat and other animal-free food products. *Economic Geography* 95(2), 136–158. https://doi. org/10.1080/00130095.2018.1508994.

Murray, C.J.L., Aravkin, A.Y., Zheng, P., Abbafati, C., Abbas, K.M., Abbasi-Kangevari, M. *et al.* (2020). Global burden of 87 risk factors in 204 countries and territories, 1990–2019: A systematic analysis for the Global Burden of Disease Study 2019. *The Lancet* 396(10258), 1223–1249. https://doi.org/10.1016/S0140-6736(20)30752-2.

Myers, M.L. (2010). Review of occupational hazards associated with aquaculture. *Journal of Agromedicine* 15(4), 412–426. https://doi.org/10.1080/1

Mok, W.K., Tan, Y.X. and Chen W.N. (2020). Technology innovations for food security in Singapore: A case study of future food systems for an increasingly natural resource-scarce world. *Trends in Food Science and Technology* 102, 155-168. https://doi.org/10.1016%2Fj.tifs.2020.06.013.

Monteiro, C.A., Cannon, G., Levy, R.B., Moubarac, J.C., Louzada, M.L., Rauber, F. *et al.* (2019). Ultra-processed foods: what they are and how to identify them. Public Health Nutr 22, 936-941. https://doi.org/10.1017/s1368980018003762.

Morach, B., Witte, B., Walker, D., von Koeller, E., Grosse-Holz, F., Rogg, J. *et al.* (2021). *Food for Thought: The Protein Transformation*. Boston Consulting Group and Blue Horizon. https://www.bcg.com/publications/2021/ the-benefits-of-plant-based-meats.

059924x.2010.512854.

Nagamine, A. and Ezura, H. (2022). Genome editing for improving crop nutrition. *Frontiers in Genome Editing* 4, 850104. https://doi.org/10.3389/ fgeed.2022.850104.

Nakagawa, S. and Hart, C. (2019). Where's the beef? How masculinity exacerbates gender disparities in health behaviours. *Socius: Sociological Research for a Dynamic World* 5. https://doi. org/10.1177/237802311983180.

National Food Strategy: The Plan – July 2021. https://www.nationalfoodstrategy.org. Negulescu, P.G., Risner, D., Sumner, D., Block, D., Nandi, S., McDonal, K.A. (2022). Techno-economic modelling and assessment of cultivated meat: Impact of production bioreactor scale. https://doi.org/10.31224/2414.

Newton, P. and Blaustein-Rejto, D. (2021). Social and economic opportunities and challenges of plant-based and cultured meat for rural producers in the US. *Frontiers in Sustainable Food Systems* 5, 624270. https://doi. org/10.3389/fsufs.2021.624270.

Nicholas, I. and Silver, M. (2021) Tufts receives \$10 million grant to help develop cultivated meat. TuftsNow, 15 October. https://now.tufts. edu/2021/10/15/tufts-receives-10-million-grant-help-develop-cultivatedmeat Accessed 27 October 2023.

Nogueira, A. (2023). Patience – and innovation – needed as US plantbased meat market slows. *Just Food*, January 24. https://www.just-food. com/features/patience-and-innovation-needed-as-us-plant-based-meatmarket-slows/.

Nolden, A.A. and Forde, C.G. (2023). The nutritional quality of plant-based foods. *Sustainability* 15(4), 3324. https://doi.org/10.3390/su15043324.

Nordborg, M., Davis, J., Cederberg, C. and Woodhouse, A. (2017). Freshwater ecotoxicity impacts from pesticide use in animal and vegetable foods

Pender, J., Thomas, H., John, C. and Tracey, F. (2019). *Rural America at a Glance*. United States Department of Agriculture Economic Research Service. https://www.ers.usda.gov/publications/pub-details/?pubid=95340.

Perfect Day Inc., (2021). Comparative Life Cycle Assessment of Perfect Day Whey Protein Production to Dairy Protein. Portland. https://m4f6w9b2. rocketcdn.me/wp-content/uploads/2022/01/Comparative-Perfect-Day-Whey-LCA-report-prepared-by-WSP_20AUG2021_Non-Confidential-1.pdf.

People's Republic of China, Ministry of Agriculture and Rural Affairs (2021). 十四五' 全国农业农村科技发展. http://www.moa.gov.cn/govpublic/ KJJYS/202112/P020220106615353271383.pdf. Accessed 31 August 2022.

Pieper, M., Michalke, A. and Gaugler, T. (2020). Calculation of external climate costs for food highlights inadequate pricing of animal products. *Nature Communications* 11, 6117. https://doi.org/10.1038/ s41467-020-19474-6.

Piernas, C., Cook, B., Stevens, R., Stewart, C., Hollowell, J., Scarborough, P. et al. (2021). Estimating the effect of moving meat-free products to the meat aisle on sales of meat and meat-free products: A non-randomised controlled intervention study in a large UK supermarket chain. *PLoS Medicine* 18(7), e1003715. https://doi.org/10.1371/journal.pmed.1003715.

produced in Sweden. Science of the Total Environment 581-582, 448-459. https://doi.org/10.1016/j.scitotenv.2016.12.153.

Nutraceutical Business Review (2022). MycoTechnology and OIA establish alternative protein venture, 6 May 2022. https://nutraceuticalbusinessreview. com/news/article_page/MycoTechnology_and_OIA_establish_alternative_ protein_venture/200616. Accessed 3 July 2023.

O'Neill, E.N., Cosenza, Z.A., Baar, K., Block, D.E. (2020). Considerations for the development of cost-effective cell culture media for cultivated meat production. *Comprehensive Reviews in Food Science and Food Safety* 20(1), 686-709 https://doi.org/10.1111/1541-4337.12678.

Organization for Economic Cooperation and Development and Food and Agriculture Organization of the United Nations (2021). *OECD-FAO Agricultural Outlook 2021-2030*. Paris: OECD Publishing. https://doi. org/10.1787/19428846-en.

Organisation for Economic Cooperation and Development and Food and Agriculture Organization of the United Nations (2022). OECD-FAO Agricultural Outlook, OECD Agriculture statistics (database), http://dx.doi. org/10.1787/agr-outl-data-en.

Pagliai, G., Dinu, M., Madarena, M.P., Bonaccio, M., Iacoviello, L., Sofi, F. (2021). Consumption of ultra-processed foods and health status: A systematic review and meta-analysis. *British Journal of Nutrition* 125(3), 308-318. https://doi.org/10.1017/s0007114520002688.

Piochi, M., Micheloni, M. and Torri, L. (2022). Effect of informative claims on the attitude of Italian consumers towards cultured meat and relationship among variables used in an explicit approach. *Food Research International* 151, 110881. https://doi.org/10.1016/j.foodres.2021.110881.

Pfister, S., Koehler, A. and Hellweg, S. (2009). Assessing the environmental impacts of freshwater consumption in LCA. *Environmental Science* & *Technology*, 43(11), 4098-4104. https://doi.org/10.1021/es802423e.

Poletto, R. and Hötzel, M.J. (2012). The five freedoms in the global animal agriculture market: Challenges and achievements as opportunities. *Animal Frontiers* 2(3), 22–30. https://doi.org/10.2527/af.2012-0045.

Poore, J. and Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science* 360(6392), 987–992. https://doi.org/10.1126/science.aaq0216.

Post, M.J., Levenberg, S., Kaplan, D.L., Genovese, N., Fu, J., Bryant, C.J. *et al.* (2020). Scientific, sustainability and regulatory challenges of cultured meat. *Nature Food 1*, 403-415. https://doi.org/10.1038/s43016-020-0112-z.

Potter, H., Lundmark, L. and Röös, E. (2020). *Environmental Impact of Plant-Based Foods: Data Collection for the Development of a Consumer Guide for Plant-Based Foods*. Uppsala: Swedish University of Agricultural Sciences, NL Faculty/Department of Energy and Technology.

Powell, S., Lau, D., Grundy, K., Alezander, L., Deboo, M. (2019). *Thematic Research: The Great Protein Shake-up?* Jefferies. https://www.jefferies.com/ CMSFiles/Jefferies.com/Files/Insights/The_Great_Protein_Shakeup.pdf.

Pakseresht, A., Kaliji, S.A. and Canavari, M. (2022). Review of factors affecting consumer acceptance of cultured meat. *Appetite* 170, 105829. https:// doi.org/10.1016/j.appet.2021.105829.

Parry, J. and Szejda, K. (2019). How to Drive Plant-based Food Purchasing: *Key Findings from a Mindlab Study into Implicit Perceptions of the Plantbased Category*. Washington, D.C.: The Good Food Institute. https://gfi.org/ images/uploads/2019/10/GFI-Mindlab-Report-Implicit-Study_Strategic_ Recommendations.pdf. Proctor, H. (2012) Animal sentience: Where are we and where are we heading? *Animals* 2, 628-639. https://doi.org/10.3390/ani2040628.

Rao, N., Gazdar, H., Chanchani, D. and Ibrahim, M. (2019). Women's agricultural work and nutrition in South Asia: From pathways to a cross-disciplinary, grounded analytical framework. *Food Policy* 82, 50-62. https://doi.org/10.1016/j.foodpol.2018.10.014.

Raty, R. and Carlsson-Kanyama, A. (2010). Energy consumption by gender in some European countries. *Energy Policy* 38(1), 646-649. https://doi. org/10.1016/j.enpol.2009.08.010.

Ren, H., Eviner, V.T., Gui, W., Wilson, G.W., Cobb, A.B., Yang, G., Zhang, Y., Hu, S. and Bai, Y. (2018). Livestock grazing regulates ecosystem multifunctionality in semi-arid grassland. *Functional Ecology*, 32(12), 2790-2800. https:// doi.org/10.1111/1365-2435.13215.

Reardon, T., Tschirley, D., Liverpool-Tasie, L.S.O., Awokuse, T., Fanzo, J., Minten, B. *et al.* (2021). The processed food revolution in African food systems and the double burden of malnutrition. *Global Food Security* 28, 100466. https://doi.org/10.1016/j.gfs.2020.100466.

Richter, B.D., Bartak, D., Caldwell, P., Davis, K.F., Debaere, P., Hoekstra, A.Y. *et al*, 2020. Water scarcity and fish imperilment driven by beef production. *Nature Sustainability*, 3(4), 319-328. https://doi.org/10.1038/ s41893-020-0483-z

Reisinger, A., and Clark, H., 2018. How much do direct livestock emissions actually contribute to global warming? Global Change Biology 24, 1749–1761. https://doi.org/10.1111/gcb.13975.

Reiss, J., Robertson, S. and Suzuki, M. (2021). Cell sources for cultivated

Rojas-Downing, M.M., Nejadhashemi, A.P., Harrigan, T. and Woznicki, S.A. (2017). Climate change and livestock: Impacts, adaptation, and mitigation. *Climate Risk Management 16*, 145-163. https://doi.org/10.1016/j. crm.2017.02.001.

Rowntree, J.E., Stanley, P.L., Maciel, I.C.F., Thorbecke, M., Rosenzweig, S.T., Hancock, D.W. *et al.* (2020). Ecosystem impacts and productive capacity of a multi-species pastured livestock system. *Frontiers in Sustainable Food Systems* 4, 544984. https://doi.org/10.3389/fsufs.2020.544984.

Rubio, N.R., Xiang, N. and Kaplan, D.L. (2020). Plant-based and cell-based approaches to meat production. *Nature Communications* 11, 6276. https://doi.org/10.1038/s41467-020-20061-y.

Ruttan, V.W. (2005). Social science knowledge and induced institutional innovation. Staff Papers 13784. University of Minnesota Department of Applied Economics, College of Agricultural, Food and Environmental Sciences. https://doi.org/10.22004/ag.econ.13784.

Sabelli, C. (2023). Scientists protest Italy's ban on cultivated meat. *Nature Italy*, 5 April. https://doi.org/10.1038/d43978-023-00050-7.

Saget, S., Port Costa, M., Santos, C. S., Vasconcelos, M., Styles, D. and Williams M. (2021) Comparative life cycle assessment of plant and beed-

meat: Applications and considerations throughout the production workflow. *International Journal of Molecular Sciences* 22(14), 7513. https://doi. org/10.3390/ijms22147513.

Ren, H., Taube, F., Stein, C., Zhang, Y., Bai, Y. and Hu, S. (2017). Grazing weakens temporal stabilizing effects of diversity in the Eurasian steppe. *Ecology and Evolution* 8, 231-241. https://doi.org/10.1002/ece3.3669.

Rioja-Lang, F.C., Brown, J.A., Brockhoff, E.J. and Faucitano, L. (2019). A review of swine transportation research on priority welfare issues: a Canadian perspective. *Frontiers in Veterinary science*, 6, 36. https://doi. org/10.3389/fvets.2019.00036

Ripple, W.J., Wolf, C., Newsome, T.M., Betts, M.G., Ceballos, G., Courchamp,
F. et al. (2019). Are we eating the world's megafauna to extinction?
Conservation Letters 12(3), e12627. https://doi.org/10.1111/conl.12627.

Richter, B.D., Bartak, D., Caldwell, P., Davis, K.F., Debaere, P., Hoekstra, A.Y. (2020). Water scarcity and fish imperilment driven by beef production. *Nat Sustain* 3, 319–328 (2020). https://doi.org/10.1038/s41893-020-0483-z.

Ritchie, H., Rosado, P., Roser, M. (2019). Meat and dairy production. *Our World in Data*. https://ourworldindata.org/meat-production.

The Rockefeller Foundation (2021). *True Cost of Food: Measuring What Matters to Transform the U.S. Food System*. https://www. rockefellerfoundation.org/wp-content/uploads/2021/07/True-Cost-of-Food-Full-Report-Final.pdf. based patties, including carbon opportunity costs. *Sustainable Production and Consumption*, 28, 936-952. https://doi.org/10.1016/j.spc.2021.07.017.

Sakadevan, K. and Nguyen, M.L. (2017). Livestock production and its impact on nutrient pollution and greenhouse gas emissions. *Advances in Agronomy* 141, 147-184. https://doi.org/10.1016/bs.agron.2016.10.002.

Santo, R.E., Kim, B.F., Goldman, S.E., Dutkiewicz, J., Biehl, E.M.B., Bloem, M.W. et al. (2020). Considering plant-based meat substitutes and cellbased meats: A public health and food systems perspective. *Frontiers in Sustainable Food Systems* 4. https://www.frontiersin.org/article/10.3389/ fsufs.2020.00134https://doi.org/10.3389/fsufs.2020.00134.

Sara, R.B.M.E., Syse, K.L. and Mathiesen, S.D. (2022). Precious blood and nourishing offal: Past and present slaughtering perspectives in Sámi reindeer pastoralism. *Pastoralism* 12, 20. https://doi.org/10.1186/ s13570-021-00224-2.

Saunois, M., Stavert, A.R., Poulter, B., Bousquet, P., Canadell, J.G., Jackson, R.B. *et al.* (2020). The global methane budget 2000–2017. *Earth System Science Data* 12(3), 1561–1623. https://doi.org/10.5194/ essd-12-1561-2020.

Sawers, P. (2022). 'Hybrid meat'? Meatable wants to get lab-grown meat to market faster by combining with plant-based proteins. *TechCrunch*, 16 November. https://techcrunch.com/2022/11/16/hybrid-meat-meatablewants-to-get-lab-grown-meat-to-market-faster-by-combining-with-plantbased-proteins/. Accessed 25 August 2023.

Rosenfield, D.L. and Tomiyama, A.J. (2021). Gender differences in meat consumption and openness to vegetarianism. *Appetite* 166, 105475. https:// doi.org/10.1016/j.appet.2021.105475.

Rothwell, S.A., Doody, D.G., Johnston, C., Forber, K.J., Cencic, O., Rechberger, H. *et al.* (2020). Phosphorus stocks and flows in an intensive livestock dominated food system. *Resources, Conservation and Recycling* 163, 105065. https://doi.org/10.1016/j. resconrec.2020.105065. Scarborough, P., Adhikari, V., Harrington, R. A., Elhussein A., Briggs, A., Rayner, M. *et al.* (2020) Impact of the announcement and implementation of the UK Soft Drinks Industry Levy on sugar content, price, product size and number of available soft drinks in the UK, 2015-19: A controlled interrupted time series analysis. *PLOS Medicine*. https://doi.org/10.1371/ journal.pmed.1003025

Scherer, L., Boom, H.A., Barbarossa, V. and van Bodegom, P.M. (2023). Climate change threats to the global functional diversity of freshwater fish. *Global Change Biology* 29(13), 3781-3793. https://doi.org/10.1111/gcb.16723.

67

Scherer, L., Svenning, J.C., Huang, J., Seymour, C.L., Sandel, B., Mueller, N. et al. (2020). Global priorities of environmental issues to combat food insecurity and biodiversity loss. Science of The Total Environment 730, 139096. https://doi.org/10.1016/j.scitotenv.2020.139096.

Scherer, L., Tomasik, B., Rueda, O. and Pfister, S. (2018). Framework for integrating animal welfare into life cycle sustainability assessment. The International Journal of Life Cycle Assessment 23, 1476–1490. https://doi. org/10.1007/s11367-017-1420-x.

Schwingshackl, L., Schwedhelm, C., Hoffmann, G., Lampousi, A.M., Knüppel, S., Iqbal, K. et al. (2017). Food groups and risk of all-cause mortality: A systematic review and meta-analysis of prospective studies. The American Journal of Clinical Nutrition 105(6), 1462-1473. https://doi.org/10.3945/ ajcn.117.153148.

Scialabba, G. (2019). Nothing before us: Weighing the case for progress in an age of anxiety. The Baffler (43). https://thebaffler.com/salvos/nothing-before-us-. Accessed 25 August 2023.

Scoones, I. (2022). Livestock, methane and climate change: The politics of global assessments. WIREs Climate Change 14(1), e790. https://doi. org/10.1002%2Fwcc.790.

Singapore Food Agency (2023). Requirements for the safety assessment of novel foods and novel food ingredients. 20 July. https://www.sfa.gov.sg/ docs/default-source/food-information/requirements-for-the-safety-assessment-of-novel-foods-and-novel-food-ingredients.pdf.

Singh, S., Swan Yap, W., Yu Ge, X., Lee Xi Min, V., Choudhury, D. (2022). Cultured meat production fuelled by fermentation. Trends in Food Science and Technology 120, 48–58. https://doi.org/10.1016/j.tifs.2021.12.028.

Sinke, P., Swartz, E., Sanctorum, H., van der Giesen, C. and Odegard, I. (2023). Ex-ante life cycle assessment of commercial-scale cultivated meat production in 2030. The International Journal of Life Cycle Assessment 28, 234-254. https://doi.org/10.1007/s11367-022-02128-8.

Slade, J. and Alleyne, E. (2023). The psychological impact of slaughterhouse employment: A systematic literature review. *Trauma Violence Abuse* 24(2), 429-440. https://doi.org/10.1177/15248380211030243.

Smetana, S., Mathys, A., Knoch, A., Heinz, V. (2015). Meat alternatives: Life cycle assessment of most known meat substitutes. International Journal of Life Cycle Assessment 20, 1254–67. http://link.springer.com/10.1007/ s11367-015-0931-6.

Smetana, S., Ristic, D., Pleissner, D., Tuomisto, H.L., Parniakov, O., Heinz,

Sekaran, U., Lai, L., Ussiri, D.A.N., Kumar, S. and Clay, S. (2021). Role of integrated crop-livestock systems in improving agriculture production and addressing food security – A review. Journal of Agriculture and Food Research 5, 100190. https://doi.org/10.1016/j.jafr.2021.100190.

Sen, A. (1983). Poverty and Famines: An Essay on Entitlement and Deprivation. Oxford: Oxford University Press. https://doi.org/10.1093/0198 284632.001.0001.

Sexton, A. E., Garnett, T. and Lorimer, J. (2019). Framing the future of food: The contested promises of alternative proteins. Environment and Planning E: Nature and Space 2(1), 47–72. https://doi. org/10.1177/2514848619827009.

Shahbandeh, M. (2022). Meat industry value worldwide in 2021 and forecast for 2022 and 2027. Statista. https://www.statista.com/statistics/502286/ global-meat-and-seafood-market-value/#:~:text=The%20value%20of%20 the%20global,billion%20U.S.%20dollars%20by%202027.and text=The%20 production%20volume%20of%20meat,around%20328%20million%20 metric%20tons.

Shoup, M.E. (2022). SuperMeat to scale and optimize cultivated meat production processes. Food Navigator, 7 June. https://www.foodnavigator-usa. com/Article/2022/06/07/supermeat-to-scale-and-optimize-cultivated-meat-production-processes. Accessed 5 July 2023.

V. (2023). Meat substitutes: Resource demands and environmental footprints. Resources, Conservation and Recycling 190, 106831. https://doi. org/10.1016/j.resconrec.2022.106831.

Soice, E. and Johnston, J. (2021a) How cellular agriculture systems can promote food security. Frontiers in Sustainable Food Systems 5. https://doi. org/10.3389/fsufs.2021.753996.

Soice, E. and Johnston, J. (2021b). Immortalizing cells for human consumption. International Journal of Molecular Sciences 22(21), 11660. https://doi. org/10.3390/ijms222111660.

South Africa, Department of Agriculture, Land Reform and Rural Development (2022). Regulations regarding the classification, packing and marking of certain raw processed meat products intended for sale in the Republic of South Africa, 26 August. https://www.gov.za/sites/default/ files/gcis_document/202208/46789reg11478gon2410.pdf.

South Africa, Department of Health (2022). Food Control. https://www. health.gov.za/food-control/. Accessed 4 July 2023.

Southey, F. (2022) Plant-based peynir? Production and sale of 'vegan cheese' outlawed in Turkey. Food Navigator Europe, 12 July. https://www.foodnavigator.com/Article/2022/07/12/Turkey-bans-production-of-vegan-cheese.

Souza Filho, P.F., Andersson, D., Ferreira, J.A. and Taherzadeh, M.J. (2019). Mycoprotein: Environmental impact and health aspects. World Journal of Microbiology and Biotechnology 35, 147. https://doi.org/10.1007% 2Fs11274-019-2723-9.

Shurtleff, W. and Aoyagi, A. (2014). The History of Meat Alternatives (965 CE to 2013). Lafayette, CA: Soyinfo Center. https://www.soyinfocenter.com/ pdf/179/MAL.pdf.

Sillman, J., Uusitalo, V., Ruuskanen, V., Ojala, L., Kahiluoto, H., Soukka, R. et al. (2020). A life cycle environmental sustainability analysis of microbial protein production via power-to-food approaches. International Journal of Life Cycle Assessment. https://urn.fi/URN:NBN:fi-fe2020092976179.

Sim, S.Y.J., SRV, A., Chiang, J.H. and Henry, C.J. (2021). Plant proteins for future foods: A roadmap. Foods 10(8): 1967. https://doi.org/10.3390/ foods10081967.

Specht, L. (2020). An Analysis of Culture Medium Costs and Production Volumes for Cultivated Meat. The Good Food Institute. https://gfi.org/ wp-content/uploads/2021/01/clean-meat-production-volume-and-mediumcost.pdf.

Specht, L., Zaidel, M., Byrne, B., Crosser, N. and Weston, Z. (2020). Advancing Solutions for Alternative Proteins Project Report: A Systematic Analysis of the Plant Protein, Fermentation and Cultivated Meat Industries. The Good Food Institute. https://gfi.org/wp-content/uploads/2021/01/Report-Alternative-Protein-Solutions.pdf.

Springlea, C. (2022). Independent Office of Animal Production. *Effective Altruism Forum*. https://forum.effectivealtruism.org/ posts/2hKK7ZbG6funXLguc/independent-office-of-animal-protection.

Springmann, M. and Freund, F. (2022). Options for reforming agricultural subsidies from health, climate and economic perspectives. *Nature Communications* 13, 82. https://doi.org/10.1038/s41467-021-27645-2.

Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B.L., Lassaletta, L., *et al.* (2018). Options for keeping the food system within environmental limits. *Nature* 562, 519-525. https://doi.org/10.1038/ s41586-018-0594-0.

Springmann, M., Mason-D'Croz, D., Robinson, S., Wiebe, K., Godfray, H.C. J., Rayner, M. *et al.* (2017). Mitigation potential and global health impacts from emissions pricing of food commodities. *Nature Climate Change* 7, 69–74. https://doi.org/10.1038/nclimate3155.

Springmann, M., Mason-D'Croz, D., Robinson, S., Wiebe, K., Godfray, H.C.J., Rayner, M. *et al.* (2018). Health-motivated taxes on red and processed meat: A modelling study on optimal tax levels and associated health impacts. *PLoS ONE*, 13(11), e0204139. https://doi.org/10.1371/journal. pone.0204139. Sun, Z., Behrens, P., Tukker, A., Bruckner, M. and Scherer, L. (2022). Global human consumption threatens key biodiversity areas. *Environmental Science and Technology* 56, 9003-9014. https://doi.org/10.1021/acs. est.2c00506.

Swartz, E. (2021). Anticipatory Life Cycle Assessment and Techno-economic Assessment of Commercial Cultivated Meat Production. The Good Food Institute. https://gfi.org/wp-content/uploads/2021/03/cultured-meat-LCA-TEA-policy.pdf.

Swift, J. and Krätli, S. (2013). Ungovernable spaces? Rebuilding a resilient pastoral economy in northern Mali. *IIED Briefing Paper-International Institute for Environment and Development*, (17174). https://www.jstor.org/ stable/resrep01534.

Swinburn, B.A., Kraak, V.I., Allender, S., Atkins, V.J., Baker, P.I., Bogard, J.R. *et al.* (2019). The global syndemic of obesity, undernutrition and climate change: *The Lancet* Commission report. *The Lancet* Commissions 393(10173), 791–846. https://doi.org/10.1016/s0140-6736(18)32822-8.

Switzerland, Federal Food Safety and Veterinary Office (2022).

Authorization of novel foods. https://www.blv.admin.ch/blv/en/home/ lebensmittel-und-ernaehrung/rechts-und-vollzugsgrundlagen/bewilligung-und-meldung/bewilligung.html. Accessed 28 August 2023.

Sponenberg, D.P., Beranger, J., Martin, A.M. and Couch, C.R. (2018). Conservation of rare and local breeds of livestock.*World Organisation for Animal Health*, 37, 259-267. https://doi.org/10.20506/rst.37.1.2756.

Stakeholder Foods Ltd (2023). Stakeholder Foods 3D bio-prints first readyto-cook cultivated grouper fish. *PR Newswire*, 27 April. https://www. prnewswire.com/news-releases/steakholder-foods-3d-bio-prints-firstready-to-cook-cultivated-grouper-fish-301805262.html#:~:text=Unlike%20 fully%20cultivated%20meat%20products,the%20subject%20of%20a%20 provisional.

Steffen, W., Richardson, K., Rockström, J., Cornell, S.E., Fetzer, I., Bennet, E.M. et al. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science* 347, 6223. https://doi.org/10.1126/ science.1259855.

Steinfeld, H., Wassenaar, T. and Jutzi, S. (2006). Livestock production systems in developing countries: Status, drivers, trends. *Scientific & Technical Review* 25(2), 505–516. https://doi.org/10.20506/rst.25.2.1677.

Stephens, N., Di Silvio, L., Dunsford, I., Ellis, M., Glencross, A., and Sexton, A. (2018). Bringing cultured meat to market: Technical, socio-political, and regulatory challenges in cellular agriculture. *Trends in Food Science and Technology* 78, 155–166. https://doi.org/10.1016/j.tifs.2018.04.010.

Stewart, H., Kuchler, F., Cessna, J. and Hahn, W. (2020). Are plant-based analogues replacing cow's milk in the American diet? *Journal of Agricultural and Applied Economics*, 52(4), 562-579. https://doi.org/10.1017/ aae.2020.16. Systemiq (2023). The Breakthrough Effect: How to Trigger a Cascade of Tipping Points to Accelerate the Net Zero Transition. https://www.systemiq. earth/wp-content/uploads/2023/01/The-Breakthrough-Effect.pdf.

Szejda, K., Bryant, C., Urbanovich, T. (2021). US and UK consumer adoption of cultivated meat: A segmentation study. *Foods* 10(5), 1050; https://doi. org/10.3390/foods10051050.

Szenderak, J., Fróna, D., Rákos, M. (2022) Consumer acceptance of plantbased meat substitutes: A narrative Review. *Foods* 11(9), 1274. https://doi. org/10.3390/foods11091274.

Talebi Bezmin Abadi, A., Rizvanov, A.A., Haertlé, T. and Blatt, N.L. (2019) World Health Organization report: Current crisis of antibiotic resistance. *BioNanoScience* 9(4), 778–788. https://dx.doi.org/10.1007/ s12668-019-00658-4.

Taylor S. (2022). Meat wars: The unsettled intersection of federal and state food labeling regulations for plant-based meat alternatives. *University of Massachusetts Law Review*, 15(2), 269-291. https://scholarship.law. umassd.edu/umlr/vol15/iss2/3.

Taylor, P. and Reeder, R. (2020). Antibiotic use on crops in low and middle-income countries based on recommendations made by agricultural advisors. *CABI Agriculture and Bioscience* 1, 1. https://doi.org/10.1186/

Stoll-Kleemann, S. and O'Riordan, T. (2015). The sustainability challenges of our meat and dairy diets. *Environment: Science and Policy for Sustainable Dev elopment* 57(3), 34-48. https://doi.org/10.1080/00139157.2015.1025644.

Strom, S. (2016) Plant-based, the Beyond Burgers aims to stand sturdy among meat. *New York Times*, 22 May. https://www.nytimes. com/2016/05/23/business/plant-based-the-beyond-burger-aims-to-standsturdy-among-meat.html Accessed 2 November 2023.

s43170-020-00001-y.

Temple, D. and Manteca, X. (2020). Animal welfare in extensive production systems is still an area of concern. *Frontiers in Sustainable Food Systems*4. https://www.frontiersin.org/article/10.3389/fsufs.2020.545902.

Teng, P. (2020). Assuring food security in Singapore, a small island state facing COVID-19. *Food Security* 12, 801–804. https://doi.org/10.1007/s12571-020-01077-0.

Teng, T.S., Chin, Y.L., Chai, K.F. and Chen, W.N. (2021). Fermentation for future food systems: Precision fermentation can complement the scope and applications of traditional fermentation. *EMBO reports* 22(5), e52680. https://doi.org/10.15252/embr.202152680. Toribio-Mateas, M.A., Bester, A. and Klimenko, N. (2021). Impact of plantbased meat alternatives on the gut microbiota of consumers: A real-world study. Foods 10(9), 2040. https://doi.org/10.3390/foods10092040.

Treich, N. (2021). Cultured meat: Promises and challenges. *Environmental* and Resource Economics 79, 33–61. https://doi.org/10.1007/ s10640-021-00551-3.

Trisos, C.H., Merow, C. and Pigot, A.L. (2020). The projected timing of abrupt ecological disruption from climate change. *Nature* 580, 496–501. https:// doi.org/10.1038/s41586-020-2189-9.

Tso, R. and Forde, C.G. (2021). Unintended consequences: Nutritional impact and potential pitfalls of switching from animal- to plant-based foods. Nutrients 13(8), 2527. https://doi.org/10.3390/nu13082527.

TU Delft (2022). Dutch government confirms €60M investment into cellular agriculture. 24 October. https://www.tudelft.nl/en/2022/tnw/dutch-government-confirms-eur60m-investment-into-cellular-agriculture. Accessed 5 July 2023.

Tuomisto, H.L. and Teixeira de Mattos, M.J. (2011). Environmental impacts of cultured meat production. Environmental Science and Technology 45(14), 6117-6123. https://doi.org/10.1021/es200130u.

United Nations Environment Programme (2021c). The Role of Business Transforming Food Systems. UNEP, Nairobi. https://wedocs.unep.org/bitstream/handle/20.500.11822/36755/GEO4B3.pdf.

United Nations Environment Programme (2022). Plastics in Agriculture – An Environmental Challenge. Foresight Brief 029. Nairobi. https://wedocs.unep. org/bitstream/handle/20.500.11822/40403/Plastics_Agriculture.pdf.

United Nations Environment Programme (2023). Bracing for Superbugs: Strengthening environmental action in the One Health response to antimicrobial resistance. Geneva. https://www.unep.org/resources/superbugs/ environmental-action#:~:text=The%20report%20Bracing%20for%20 Superbugs,key%20part%20of%20the%20solution.

United Nations Environment Programme and Climate and Clean Air Coalition (2021). Global Methane Assessment: Benefits and Costs of Mitigating Methane Emissions. Nairobi. https://www.unep.org/resources/report/global-methane-assessment-benefits-and-costs-mitigating-methane-emissions.

United Nations Nutrition (2021). Livestock-derived foods and sustainable healthy diets. Rome. https://www.unnutrition.org/wp-content/uploads/ Livestock-Paper-EN_WEB.pdf.

United States Department of Agriculture (2011). Industry

Tuomisto, H.L., Allan, S.J., and Ellis, M.J. (2022). Prospective life cycle assessment of a bioprocess design for cultured meat production in hollow fiber bioreactors. The Science of The Total Environment 851(1), 158051. https://doi.org/10.1016/j.scitotenv.2022.158051.

Tuomisto, H.L., Ellis, M.J., and Haastrup, P. (2014). Environmental impacts of cultured meat: Alternative production scenario. Proceedings of the 9th International conference on life cycle assessment in the agri-food sector. San Francisco, CA, 8-10 October 2014. https://core.ac.uk/download/ pdf/38629617.pdf.

United Kingdom Research and Innovation (2022). New report reveals next steps for the alternative protein industry. United Kingdom Research and Innovation, 14 June. https://www.ukri.org/news/new-report-reveals-nextsteps-for-the-alternative-protein-industry/ Accessed 5 July 2023.

United Kingdom, Food Standards Agency (2023). Novel foods authorisation guidance. 3 May 2023. https://www.food.gov.uk/business-guidance/ regulated-products/novel-foods-guidance.

United Nations Department of Economic and Social Affairs, Population Division (2022). World Population Prospects 2022: Summary of Results. UN DESA/ POP/2022/TR/NO. 3. https://www.un.org/development/desa/pd/sites/www. un.org.development.desa.pd/files/wpp2022_summary_of_results.pdf.

United Nations, Department of Economic and Social Affairs, Population Division (2023). UN Population Division Data Portal. https://population. un.org/dataportal/. Accessed 12 January 2023.

Insight: Checkoff Programs Empower Business, 21 September. https://www.usda.gov/media/blog/2011/09/21/ industry-insight-checkoff-programs-empower-business.

United States Department of Agriculture Economic Research Service (2022). Quantifying Consumer Welfare Impacts of Higher Meat Prices During the COVID-19 Pandemic. https://www.ers.usda.gov/webdocs/publications/103813/err-306.pdf?v=7765/. Accessed 4 July 2023.

United States Food and Drug Administration (2022a). FDA Completes First Pre-Market Consultation for Human Food Made Using Animal Cell Culture Technology,16 November. https://www.fda.gov/food/ cfsan-constituent-updates/fda-completes-first-pre-market-consultationhuman-food-made-using-animal-cell-culture-technology. Accessed 25 July 2023.

United States Food and Drug Administration (2022b) Human Food Made with Cultured Animal Cells. Available at: https://www.fda.gov/food/food-ingredients-packaging/human-food-made-cultured-animal-cells. Accessed 4 July 2023.

United States Food and Drug Administration (2023). Human Food Made with Cultured Animal Cells Inventory. https://www.cfsanappsexternal.fda.gov/ scripts/fdcc/?set=AnimalCellCultureFoods.

United States, The White House (2023). Bold Goals for United States

United Nations Environment Programme (2019). Global Environment Outlook – GEO-6: Healthy Planet, Healthy People. Nairobi. https://doi. org/10.1017/9781108627146.

United Nations Environment Programme (2021a). Food Waste Index Report 2021. Nairobi. http://www.unep.org/resources/report/ unep-food-waste-index-report-2021.

United Nations Environment Programme (2021b). Making Peace with Nature: A Scientific Blueprint to Tackle the Climate Change, Biodiversity and Pollution Emergencies. Nairobi. https://www.unep.org/resources/ making-peace-nature.

Biotechnology and Biomanufacturing. https://www.whitehouse.gov/ wp-content/uploads/2023/03/Bold-Goals-for-U.S.-Biotechnology-and-Biomanufacturing-Harnessing-Research-and-Development-To-Further-Societal-Goals-FINAL.pdf.

United States, The White House (2022). FACT SHEET: The Biden-Alarris action plan for a fairer, more competitive, and more resilient meat and poultry supply chain, 3 January. https://www.whitehouse.gov/briefing-room/ statements-releases/2022/01/03/fact-sheet-the-biden-harris-action-plan-fora-fairer-more-competitive-and-more-resilient-meat-and-poultry-supply-chain/.

Van Boeckel, T.P., Brower, C., Gilbert, M., Grenfell, B.T., Levin, S.A., Robinson, T.P. et al. (2015). Global trends in antimicrobial use in food animals. Proceedings of the National Academy of Sciences 112(18), 5649–5654. https://doi.org/10.1073/pnas.1503141112.

Van Boeckel, T.P., Pires, J., Silvester, R., Zhao, C., Song, J., Criscuolo, N.G. et al. (2019) Global trends in antimicrobial resistance in animals in lowand middle-income countries. *Science* 365(6459), eaaw1944. https://doi. org/10.1126/science.aaw1944.

Vallone, S. and Lambin, E. F. (2023). Public policies and vested interests preserve the animal farming status quo at the expense of animal product analogs. *One Earth* 6, 1213-1226. https://doi.org/10.1016/j. oneear.2023.07.013.

Van der Weele, C. and Driessen, C. (2013). Emerging profiles for cultured meat; ethics through and as design. *Animals* 3(3), 647–662. https://doi. org/10.3390/ani3030647.

Van der Weele, C., and Tramper, J. (2014). Cultured meat: every village its own factory? *Trends Biotechnol*. 32, 294–296. https://doi.org/10.1016/j. tibtech.2014.04.009.

van Vliet, S., Bain, J.R., Muehlbauer, M.J., Provenza, F.D., Kronberg, S.L., Pieper, C.F. *et al.* (2021). A metabolomics comparison of plant-based meat and grass-fed meat indicates large nutritional differences despite comparable Nutrition Facts panels. *Scientific Reports* 11, 13828. https://doi. org/10.1038/s41598-021-93100-3. Wan, A.C.A., Tai, B.C.U. and Du, C. (2021). Food security and nutrition – A systematic approach. *Trends in Food Science and Technology* 109, 738-745. https://doi.org/10.1016/j.tifs.2020.12.024.

Wang, M., Kroeze, C., Strokal, M. and Ma, L. (2017). Reactive nitrogen losses from China's food system for the shared socioeconomic pathways (SSPs). Science of The Total Environment 605–606, 884–893. https://doi. org/10.1016/j.scitotenv.2017.06.235.

Watson, E. (2020). Mission Barns tests cell-cultured pork fat with leading pork producer; gears up for hybrid plant/animal bacon tasting. *Food Navigator USA*, 3 August. https://www.foodnavigator-usa.com/
Article/2020/08/03/Mission-Barns-tests-cell-cultured-pork-fat-with-leadingpork-producer-gears-up-for-hybrid-plant-animal-bacon-tasting#. Accessed
21 February 2023.

Watson, E. (2021). 'A transformational event in our history...' Roquette unveils 'world's largest pea protein plant'. *Food Navigator Europe*, 3 December. https://www.foodnavigator.com/Article/2021/12/03/A-transformationalevent-in-our-history-Roquette-unveils-world-s-largest-pea-proteinplant#:~:text=Roquette%20%2D%20a%20key%20supplier%20of,both%20 sides%20of%20the%20Atlantic. Accessed 3 July 2023.

Webb, P., Flynn, D.J., Kelly, N.M. and Thomas, S.M. (2023). The Transition Steps Needed to Transform Our Food Systems. In Science and Innovations for Food Systems Transformation. von Braun, J., Afsana, K., Fresco, L.O., Hassan, M.H.A. (eds). Cham: Springer. 893-907. https://link.springer.com/ chapter/10.1007/978-3-031-15703-5_48.

N

van Zanten, H.H.E., Muller, A. and Frehner, A. (2022). 5 – Land-use modeling: From farm to food systems. In *Food Systems Modelling*. Peters, C. and Thilmany, D. (eds). London: Academic Press. 89–105. https://doi. org/10.1016/B978-0-12-822112-9.00011-4.

Varijakshapanicker, P., Mckune, S., Miller, L., Henrickx, S., Balehegn, M., Dahl, G.E. *et al.* (2019). Sustainable livestock systems to improve human health, nutrition and economic status. *Animal Frontiers* 9(4), 39–50. https://doi. org/10.1093/af/vfz041.

Vergeer, R., Odegard, I. and Sinke, P. (2021). TEA of Cultivated Meat. Future Projections for Different Scenarios. Delft: CE Delft. https://cedelft.eu/ publications/tea-of-cultivated-meat/.

Vergeer, R., Rozema, J., Odegard, I. and Sinke, P. (2020). A Sustainability Charge on Meat. Delft: CE Delft. https://cedelft.eu/ publications/a-sustainability-charge-on-meat/.

Verkuijl, C., Strambo, C., Hocquet, R., Butterfield, R., Achakulwisut, P., Boyland, M. et al. (2022). A Just Transition in the Meat Sector: Why, Who and How? Stockholm: Stockholm Environment Institute. http://doi. org/10.51414/sei2022.046.

Vermeulen, S.J., Park, T., Khoury, C.K. and Béné, C. (2020). Changing diets and the transformation of the global food system. *Annals of the New York Academy of Sciences* 1478(1), 3–17. https://doi.

Weber, T., Wirths, F., Brakebusch, N. and van der Valk, J. (2021). Reply to comment "Animal welfare and ethics in the collection of fetal blood for the production of fetal bovine serum". *ALTEX – Alternatives to animal experimentation* 38(2), 324–326. https://doi.org/10.14573/altex.2103191.

Wegner, G.I., Murray, K.A., Springmann, M., Muller, A., Sokolow, S.H., Saylors, K. *et al.* (2022). Averting wildlife-borne infectious disease epidemics requires a focus on socio-ecological drivers and a redesign of the global food system. *eClinicalMedicine* 47, 101386. https://doi.org/10.1016/j. eclinm.2022.101386.

Welsh, C, Majkut, J. and Swanson, Z. (2023). The Future Appetite for Alternative Proteins. Washington, D.C.: Center for Strategic and International Studies. https://features.csis.org/ the-future-appetite-for-alternative-proteins/.

Wickramasinghe, K., Breda, J., Berdzuli, N., Rippin, H., Farrand, C. and Halloran, A. (2021). The shift to plant-based diets: Are we missing the point? *Global Food Security* 29, 100530. https://doi.org/10.1016/j. gfs.2021.100530.

org/10.1111%2Fnyas.14446.

Vishwanathan, K.H., Singh, V., Subramanian, R. (2011). Influence of particle size on protein extractability from soybean and okara. *Journal* of Food Engineering 102(3), 240-246. https://doi.org/10.1016/j. jfoodeng.2010.08.026.

Vogelsang-O'Dwyer, M., Petersen, I.L., Joehnke, M.S., Sørensen, J.C., Bez, J., Detzel, A. *et al.* (2020). Comparison of faba bean protein ingredients produced using dry fractionation and isoelectric precipitation: Techno-functional, nutritional and environmental performance. *Foods* 9(3), 322. https://doi.org/10.3390/foods9030322. Wiebers D.O. and Feigin V.L. (2020). What the COVID-19 crisis is telling humanity. *Neuroepidemiology* 54(4), 283-286. https://doi. org/10.1159/000508654.

Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T.,
Vermeulen, S. *et al.* (2019). Food in the Anthropocene: The EAT–Lancet
Commission on healthy diets from sustainable food systems. *The Lancet Commissions* 393(10170), 447-492. https://doi.org/10.1016/S0140-6736(18)31788-4.

Witte, B., Obloj, P., Koktenturk, S., Morach, B., Brigl, M., Rogg, J. *et al.* (2021). *Food for Thought: The Protein Transformation*. Boston Consulting Group and Blue Horizon Corporation. https://web-assets.bcg.com/a0/28/4295860343c6a2a5b9f4e3436114/bcg-food-for-thought-the-protein-transformation-mar-2021.pdf.

Wood, B., Williams, O., Nagarajan, V. and Sacks, G. (2021). Market strategies used by processed food manufacturers to increase and consolidate their power: A systematic review and document analysis. *Globalization and Health* 17, 17. https://doi.org/10.1186/s12992-021-00667-7.

World Bank (2022). The World Bank Annual Report 2022: Helping Countries Adapt to a Changing World. Washington, D.C. http://hdl.handle. net/10986/37972.

World Health Organisation (2017). Stop using antibiotics in healthy animals to prevent the spread of antibiotic resistance. https://www.who.int/news/item/07-11-2017-stop-using-antibiotics-in-healthy-animals-to-prevent-the-spread-of-antibiotic-resistance.

World Health Organisation (2020). Zoonoses. https://www.who.int/newsroom/fact-sheets/detail/zoonoses. Accessed 5 June 2023.

World Health Organisation (2021a). *Malnutrition*. https://www.who.int/news-

Zhong, V.W., Van Horn, L., Cornelis, M.C., Wilkins, J.T., Ning, H., Carnethon, M.R. *et al.* (2019). Associations of dietary cholesterol or egg consumption with incident cardiovascular disease and mortality. *JAMA* 321(11), 1081– 1095. https://doi.org/10.1001/jama.2019.1572.

Zhong, V.W., Van Horn, L., Greenland, P., Carnethon, M.R., Ning, H., Wilkins, J.T. *et al.* (2020). Associations of processed meat, unprocessed red meat, poultry or fish intake with incident cardiovascular disease and all-cause mortality. *JAMA Internal Medicine* 180(4), 503–512. https://doi.org/10.1001/jamainternmed.2019.6969.

Zhu,Y. and Begho, T. (2022). Towards responsible production, consumption and food security in China: A review of the role of novel alternatives to meat protein. *Future Foods* 6, 100186. https://doi.org/10.1016/j. fufo.2022.100186.

Zimmermann, M.B. and Hurrell, R.F. (2007). Nutritional iron deficiency. *The Lancet* 370(9586), 511–520. https://doi.org/10.1016/ s0140-6736(07)61235-5.

room/fact-sheets/detail/malnutrition. Accessed 1 June 2022.

World Health Organisation (2021b). *Obesity and overweight*. https://www.who. int/news-room/fact-sheets/detail/obesity-and-overweight. Accessed 1 June.

World Health Organisation (2023a). *Macronutrients*. https://www.emro.who. int/health-topics/macronutrients/introduction.html.

World Health Organisation (2023b). *Micronutrients*. https://www.who.int/ health-topics/micronutrients#tab=tab_1.

World Health Organization (2023c). Noncommunicable diseases. Fact sheet, 16 September. https://www.who.int/news-room/fact-sheets/detail/non-communicable-diseases. Accessed 21 November 2023.

Xiang, H., Sun-Waterhouse, D., Waterhouse, G.I.N., Cui, C. and Ruan, Z. (2019). Fermentation-enabled wellness foods: A fresh perspective. *Food Science and Human Wellness* 8(3), 203–243. https://doi.org/10.1016/j. fshw.2019.08.003.

Xu, X., Sharma, P., Shu, S., Lin, T., Ciais, P., Tubiello, F.N., Smith, P. et al. (2021). Global greenhouse gas emissions from animal-based foods are twice those of plant-based foods. Nature Food 2, 724–732. https://doi.org/10.1038/s43016-021-00358-x.

Zabel, F., Delzeit, R., Schneider, J.M., Seppelt, R., Mauser, W. and Václavík, T. (2019). Global impacts of future cropland expansion and intensification

on agricultural markets and biodiversity. *Nature Communications* 10, 2844. https://doi.org/10.1038/s41467-019-10775-z.

Zhong, V.W., Allen, N.B., Greenland, P., Carnethon, M.R., Ning, H., Wilkins, J.T. et al. (2021). Protein foods from animal sources, incident cardiovascular disease and all-cause mortality: A substitution analysis. *International Journal of Epidemiology* 50(1), 223–233. https://doi.org/10.1093/ije/ dyaa205.



UN Control Con

United Nations Avenue, Gigiri P.O. Box 30552, 00100 Nairobi, Kenya Tel. +254 20 762 1234 unep-publications@un.org www.unep.org

Frontiers 2023

SPECIAL ISSUE